SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of Earliest Event Reported): October 14, 1995

THE TJX COMPANIES, INC.

(Exact name of registrant as specified in charter)

DELAWARE

1-4908

04-2207613

(State or other jurisdiction (Commission File Number)

(I.R.S. Employer

of incorporation)

Identification No.)

770 Cochituate Road, Framingham, MA (Address of principal executive offices)

01701 (Zip Code)

Registrant's telephone number, including area code: (508) 390-2662

This is page 1 of ___ pages. Exhibit Index appears on page ___.

On October 14, 1995, The TJX Companies, Inc., a Delaware corporation (the "Registrant") and Melville Corporation, a New York corporation ("Melville"), entered into a Stock Purchase Agreement (the "Stock Purchase Agreement") pursuant to which the Registrant will acquire from Melville (the "Acquisition") at the Closing (as defined in the Stock Purchase Agreement) all of the capital stock of Marshalls of Roseville, Minn., Inc. (the "Shares"). purchase price for the Shares is \$375,000,000 in cash plus junior convertible preferred stock (the "Preferred Stock") that has an aggregate liquidation preference equal to \$175,000,000. The preferred stock will be issued in two series - \$25,000,000 of Series D Cumulative Convertible Preferred Stock (the "Series D Preferred Stock"), which is automatically convertible into shares of the Registrant's common stock ("Common Stock") on the first anniversary of its issuance if not earlier redeemed for cash or converted into such Common Stock, and \$150,000,000 of Series E Cumulative Convertible Preferred Stock (the "Series E Preferred Stock"), which is automatically convertible into Common Stock on the third anniversary of its issuance if not earlier converted into such Common Stock. The Preferred Stock will be convertible, in the aggregate, into between 9.4 million and 13.5 million shares of Common Stock, depending on the market price of such Common Stock during the five trading days before the Closing and further depending on the market price of such Common Stock at the time of conversion. The cash portion of the purchase price is subject to adjustment following the Closing in accordance with the Stock Purchase Agreement.

At the Closing, the Registrant and Melville will enter into a Standstill and Registration Rights Agreement (the "Standstill and Registration Rights Agreement") pursuant to which Melville will agree (i) not to acquire any voting securities of the Registrant until such voting securities held by Melville represent less than 3 percent of the total combined voting power of all of the Registrant's outstanding voting securities and (ii) to vote all of the Registrant's voting securities held by it in the manner recommended by the Registrant's Board of Directors or, if the agreement to so vote shall be prohibited or invalid, then to vote such voting securities in the same proportion as the votes cast by or on behalf of the other holders of the Registrant's voting securities.

The Standstill and Registration Rights Agreement also provides that the Registrant will register, under the Securities Act of 1933, the Series E Preferred Stock held by Melville, or the shares of Common Stock received by Melville upon conversion of Series D Preferred Stock or Series E Preferred Stock, on not more than two separate occasions on demand and on not more than three separate occasions in connection with a registration of Common Stock by the Registrant.

Consummation of the Acquisition is subject to the satisfaction of certain conditions, including absence of regulatory prohibition, and the Stock Purchase Agreement is subject to termination in certain circumstances, including termination by either party if the Closing does not occur by January 31, 1996.

The Registrant has obtained financing commitments, subject to certain conditions, from a group of banks to fund the cash portion of the purchase price and the working capital needs of the Registrant and the acquired business. These financing commitments will terminate automatically unless definitive loan documents are executed on or before January 31, 1996.

The foregoing description is qualified in its entirety by reference to the Stock Purchase Agreement, a copy of which is attached hereto as Exhibit 2.2, the form of Preferred Stock Subscription Agreement (which includes forms of the Certificates of Designations, Preferences and Rights for the Series D Preferred Stock and the Series E Preferred Stock, respectively), a copy of which is attached hereto as Exhibit 10.1, and the form of Standstill and Registration Rights Agreement, a copy of which is attached hereto as Exhibit 10.2.

Item 7. Financial Statements and Exhibits.

- (c) Exhibits
- 2.2 Stock Purchase Agreement dated as of October 14, 1995 between the Registrant and Melville Corporation.
- 10.1 Form of Preferred Stock Subscription Agreement between the Registrant and Melville Corporation, including form of Certificate of Designations, Preferences and Rights for Series D Cumulative Convertible Preferred Stock and form of Certificate of Designations, Preferences and Rights for Series E Cumulative Convertible Preferred Stock
- 10.2 Form of Standstill and Registration Rights Agreement between the Registrant and Melville Corporation.
- 10.3 Commitment Letter dated October 14, 1995 among The First National Bank of Chicago, Bank of America Illinois, The Bank of New York, Pearl Street L.P. and the Registrant.
- 99.1 Press Release issued by the Registrant on October 16, 1995.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

THE TJX COMPANIES, INC.

By:/s/ Donald G. Campbell

Name: Donald G. Campbell Title: Senior Vice President/Finance

Date: November 7, 1995

EXHIBIT INDEX

Description of Exhibits

Page

Exhibit No.

2.2	Stock Purchase Agreement dated as of October 14, 1995 between the Registrant and Melville Corporation.	
10.1	Form of Preferred Stock Subscription Agreement between the Registrant and Melville Corporation, including form of Certificate of Designations, Preferences and Rights for Series D Cumulative Convertible Preferred Stock and form of Certificate of Designations, Preferences and Rights for Series E Cumulative Convertible Preferred Stock.	
10.2	Form of Standstill and Registration Rights Agreement between the Registrant and Melville Corporation.	
10.3	Commitment Letter dated October 14, 1995 among The First National Bank of Chicago, Bank of America Illinois, The Bank of New York, Pearl Street L.P. and the Registrant.	
99.1	Press Release issued by the Registrant on October 16, 1995.	

MARSHALLS OF ROSEVILLE, MINN., INC.

STOCK PURCHASE AGREEMENT

between

The TJX Companies, Inc.

as Buyer

and

Melville Corporation,

as Seller

Dated as of October 14, 1995

TABLE OF CONTENTS

1.	DEFINI	TIONS	1
	1.1. 1.2. 1.3.	Certain Matters of Construction	1 2 4
2.	ACQUIS	ITION; EXCLUDED LIABILITIES	12
3.	PAYMEN	T AND CLOSING	12
	3.1. 3.2. 3.3. 3.4.	Transaction Price. Time and Place of Closing. Delivery. Cash Price Adjustment.	13 13
4.	REPRESE	ENTATIONS AND WARRANTIES OF SELLER	16
	4.1. 4.2. 4.3. 4.4. 4.5. 4.6. 4.7. 4.8. 4.9. 4.11. 4.12. 4.13. 4.14. 4.15. 4.16. 4.17. 4.18.	Corporate Matters, etc. Financial Statements, etc. Change in Condition. Liabilities. Assets. Intellectual Property Rights. Accounts; Funds, etc. Certain Contractual Obligations Insurance, etc. Transactions with Affiliates. Compliance with Laws, etc. Tax Matters. [Intentionally Omitted.] Employee Benefit Plans. Environmental Matters, etc Employees and Labor Relations Litigation, etc. [Intentionally Omitted.] Brokers, etc.	18 19 22 24 25 27 28 28 30 31 32 32 33
5.	REPRESE	ENTATIONS AND WARRANTIES OF BUYER	33
	5.1. 5.2. 5.3. 5.4.	Corporate Matters	33 33 34

	5.6. 5.7.	FinancingBrokers, etc	
6.	CERTAIN 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 6.9. 6.10. 6.11. 6.12. 6.13. 6.14. 6.15. 6.16. 6.17. 6.18. 6.19. 6.20.	N AGREEMENTS OF THE PARTIES. No Solicitation of Other Offers. Access to Premises and Information. Confidentiality Covenant of Buyer. Operation of Business in the Ordinary Course. Certain Notices. Preparation for Closing. Tax Matters. Expenses of Transaction. Confidentiality Covenants. Books and Records; Personnel. Use of Certain Names and Marks. Further Assurances. Reimbursement by the Parties. Open and Closed Stores. Financial Statement Deliveries. Insurance Policies. Divestiture. Failed Consents. No Solicitation or Employment. Manchester, CT and Staten Island, NY Stores; Etc.	35 36 36 33 38 33 38 33 44 45 46 47 47 47 48 48 55 55
7.	CONDITI 7.1. 7.2. 7.3. 7.4. 7.5. 7.6. 7.7.	CONS TO THE OBLIGATION TO CLOSE OF BUYER. Representations, Warranties and Covenants. Closing Agreements. Legality; Governmental Authorization; Litigation. Affiliate Debt. Financing. Opinion of Counsel. General.	53 54 54 54 54 54
8.	CONDITI 8.1. 8.2. 8.3. 8.4.	CONS TO THE OBLIGATION TO CLOSE OF SELLER	55 55 56
9.		MENT AND EMPLOYEE BENEFITS ARRANGEMENTS	

	9.2. 9.3. 9.4. 9.5. 9.6. 9.7.	Disposition of Company Plans and Benefit Arrangements Employment	58 58 60 60 61
10.	INDEMN: 10.1. 10.2. 10.3. 10.4. 10.5. 10.6.	IFICATION Indemnification. Time Limitation on Indemnification. Monetary Limitations on Seller's Indemnification. Certain Matters of Construction. Third Party Claims. No Circular Recovery. Nature of Indemnification Payments.	63 64 64 64
11.	CONSENT 11.1. 11.2.	TO JURISDICTION; JURY TRIAL WAIVER	66
12.	TERMINA 12.1. 12.2.	TION Termination of Agreement Effect of Termination	67
13.	MISCELLA 13.1. 13.2. 13.3. 13.4. 13.5. 13.6. 13.7. 13.8. 13.9. 13.10. 13.11. 13.12.	ANEOUS. Entire Agreement; Waivers Amendment or Modification. Survival, etc Independence of Representations and Warranties. Severability. Knowledge Successors and Assigns. Notices. Public Announcements. Headings, etc Third Party Beneficiaries. Counterparts. Governing Law.	69 69 69 70 71 71 71

EXHIBITS

Exhibit A - Preferred Stock Subscription Agreement

Exhibit B - Standstill and Registration Rights Agreement

Exhibit C - Forms of Legal Opinions of Seller's Counsel

Exhibit D - Forms of Legal Opinions of Buyer's Counsel

SCHEDULES

Form of Company Net Assets Statement

Schedule 3.4A

Schedule 3.4B	Estimated Statement of Company Net Assets as of October 30, 1995
Schedule 4.1.4	Required Approvals, Consents, Filings, Etc.
Schedule 4.1.7	List of Company Subsidiaries
Schedule 4.3	Changes in Condition
Schedule 4.3(b)(i)	Budget
Schedule 4.3(b)(iv)(A)	Cash Management Program
Schedule 4.3(b)(iv)(B)	Distributions Made by the Company
Schedule 4.5.1	Real Property; Liens on Assets
Schedule 4.5.2(a)	Real Property Leases-In
Schedule 4.5.2(b)	Real Property Leases, Etc. Under Which Company or a Subsidiary is Lessor
Schedule 4.6	Intangibles
Schedule 4.7	Bank Accounts
Schedule 4.8	Contractual Obligations
Schedule 4.9(a)	Insurance Policies and Binders
Schedule 4.10	Affiliate Relationships
Schedule 4.11	Violations and Defaults; Permits and Applications Therefor
Schedule 4.12	Tax Exceptions
Schedule 4.14	Company Plans and Benefit Arrangements

- V -

Schedule 4.16	Disputes with Employees
Schedule 4.17	Litigation Against Company or Subsidiaries
Schedule 6.4(d)	Certain Debt Incurrences
Schedule 6.20	Principal Terms of Manchester, CT and Staten Island, NY Leases
Schedule 7.2(i)	Terms of Transitional Services
Section 9.5	Severance Benefits
Section 9.5A	Limit on Buyer Responsibility for Severance Benefits
	-vi-

Schedule 4.15 Environmental Matters

STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT (this "Agreement") is made as of the 14th day of October, 1995, between The TJX Companies, Inc., a Delaware corporation (the "Buyer"), and Melville Corporation, a New York corporation (the "Seller").

Recitals

- 1. Seller owns all of the issued and outstanding shares of capital stock (the "Shares") of Marshalls of Roseville, Minn., Inc., a Minnesota corporation (the "Company").
- 2. Seller desires to sell and transfer the Shares to Buyer, and Buyer desires to purchase (the "Purchase") the Shares from Seller, all upon the terms

and subject to the conditions set forth in this Agreement.

Agreement

Therefore, in consideration of the foregoing and the mutual agreements and covenants set forth below, the parties hereto hereby agree as follows:

- DEFINITIONS. For purposes of this Agreement:
- 1.1. Certain Matters of Construction. In addition to the definitions referred to as set forth below in this Section 1:
 - (a) The words "hereof", "herein", "hereunder" and words of similar import shall refer to this Agreement as a whole and not to any particular Section or provision of this Agreement, and reference to a particular Section of this Agreement shall include all subsections thereof.
 - (b) The words "party" and "parties" shall refer to Seller and Buyer.
 - (c) Definitions shall be equally applicable to both the singular and plural forms of the terms defined, and references to the masculine, feminine or neuter gender shall include each other gender.
 - (d) Accounting terms used herein and not otherwise defined herein are used herein as defined by generally accepted accounting principles in effect as of the date hereof.
 - (e) All references in this Agreement to any Exhibit or Schedule shall, unless the context otherwise requires, be deemed to be a reference to an Exhibit or Schedule, as the case may be, to this Agreement, all of which are made a part of this Agreement.

Term 	Definition
"Accountants" "Accounting Policy Changes" "Affected Store" "Affiliate Relationships" "Affiliated Group" "Agreement" "Annual Balance Sheet" "Annual Financials" "Assets" "Assets" "Assemed Store Contribution" "Benefit Arrangement" "Benefit Transition Period" "Books and Records" "Budget" "Buyer's Closed Store Damages" "Buyer's Deemed Sales Price Notice" "Cash Management Program" "Cash Price Adjustment" "Cash Price Adjustment" "Closing" "Closing Agreements" "Closing Balance Sheet" "Closing Date" "Commitment Letter" "Company Net Assets Statement" "Company Net Assets Statement" "Company Net Assets Statement" "Confidentiality Agreement" "Consent" "Conpers Report" "Delivered Lease" "Dol" "Employee"	Section 6.7(g) Section 3.4(a) Section 6.17 Section 4.10 Section 4.12 Preamble Section 4.2.1(a) Section 4.5.1 Section 9.1(d) Section 9.1(d) Section 9.2 Section 6.17.1 Section 6.17.2 Section 6.17.2 Section 6.7(g) Section 6.7(g) Section 3.4(b) Section 3.2 Section 3.2 Section 3.2 Section 3.2 Section 5.6 Recitals Section 3.4(a) Section 9.1(c) Section 4.8 Section 4.8 Section 4.8 Section 3.4(a) Section 4.8 Section 3.4(a) Section 4.8 Section 3.4(a) Section 3.4(a) Section 4.8 Section 3.4(a) Section 9.1(c)
"Environmental Condition"	Section 10.8

"Estimated Cash Purchase Price" Section 3.1 "Estimated Purchase Price" Section 3.1 "Eviction" Section 6.18 "Financial Statements" Sections 4.2.1(a) & 6.2 "Former Employee" Section 9.1(b) "General Survival Period"
"Goldman" Section 10.2 Section 5.7 "HSR Act" Section 4.1.4 "HSR Act"
"Indemnifying Party"
"Indemnitee"
"Insurance Policies"
"Intangibles"
"Interim Balance Sheet"
"Interim Financials"
"Inventory Firm"
"Inventory Statement" Section 10.1 Section 10.1 Section 4.9 Section 4.6 Section 4.2.1(b) Sections 4.2.1(b) & 6.2 Section 3.4(a) Section 3.4(a) "IRS" Section 4.14(a) "Landlord" Section 6.18 "Lead Lenders" Section 5.6 "Leases" Section 4.5.2 "Leases-In" Section 4.5.2 "Licenses" Section 4.6 "Miscellaneous Taxes" Section 6.7(c) "Monthly Financials" Sections 4.2.1(c) & 6.2 "Morgan Stanley"
"New Buyer Store" Section 4.19 Section 6.17.1 "New Seller Store"
"Non-Active Employees" Section 6.17.1 Section 9.3 "Non-Active Limple, SSI
"Permits"
"Post-Closing Claims"
"Post-Closing Tax Period"
"Pre-Closing Tax Period"
"Purchase" Section 4.11 Section 6.16 Section 6.7(c) Section 6.7(c) Recitals Section 3.1 Section 4.5.1 "Purchase Price"
"Real Property"
"Rent Increase" Section 6.18 "Reserved Claims" Section 10.2 "Salomon" Section 5.7 "Section 338(h)(10) Election"
"Securities Act" Section 6.7(a) Section 5.4(b) "Seller" Preamble "Seller's Employee Stock Ownership Plan" Section 9.1(e)

Section 4.8(f)

"Equipment"

Section 9.1(f) "Seller's 401(k) Profit Sharing Plan" "Shares" Recitals "Similar Sales Volume Stores" Section 6.17.1 "Store Contribution" Section 6.17.1 "Store Schedule" Section 6.17 "Target Net Asset Amount" Section 3.4(a) "Tax Loss" Section 6.7(c)(i)"Transfer Taxes" Section 6.7(e) "Transitional Services Agreement" Section 7.2 "Undelivered Leases" Section 4.5.2

- 1.3. Certain Definitions. The following terms shall have the following meanings:
 - 1.3.1. Action. The term "Action" shall mean any claim, action, cause of action or suit (in contract or tort or otherwise), arbitration, proceeding or investigation by or before any Governmental Authority (and whether brought by any Governmental Authority or any other Person).
 - 1.3.2. Affiliate. The term "Affiliate" shall mean, as to the Company (or other specified Person), each Person directly or indirectly controlling, controlled by or under direct or indirect common control with the Company (or such specified Person); provided that neither the Company

nor any Subsidiary shall be considered an Affiliate of Seller.

- 1.3.3. Affiliate Debt. "Affiliate Debt" shall mean all Debt between the Company or any Subsidiary, on the one hand, and Seller or any Affiliate of Seller, on the other hand, and all intercompany advances of funds between the Seller or any of its Affiliates, on the one hand, and the Company or any Subsidiary, on the other hand.
- 1.3.4. Alternative Accountants. The term "Alternative Accountants" shall mean an accounting firm of recognized national standing (other than Coopers and KPMG Peat Marwick) mutually acceptable to Seller and Buyer or if Seller and Buyer do not designate such a mutually acceptable firm within three Business Days of the date any dispute under this Agreement is required to be submitted to such a firm then an accounting firm of nationally recognized standing (other than Coopers and KPMG Peat Marwick) chosen by lot.
- 1.3.5. Business. The term "Business" shall mean the business of the Company and its Subsidiaries as such business is currently conducted.
- 1.3.6. Business Day. The term "Business Day" shall mean any day on which banking institutions in New York, New York and Boston, Massachusetts are customarily open for the purpose of transacting business.

- 1.3.7. Buyer Stock. The term "Buyer Stock" shall mean the shares of Buyer's Series D Cumulative Convertible Preferred Stock, par value \$1 per share, and Series E Cumulative Convertible Preferred Stock, par value \$1 per share, to be issued by Buyer to Seller at Closing pursuant and subject to the terms of the Preferred Stock Subscription Agreement.
- 1.3.8. By-laws. The term "By-laws" shall mean all written rules, regulations and by-laws, and all other documents (other than the Charter), relating to the management, governance or internal regulation of a Person (other than an individual) or interpretative of the Charter of such Person, each as from time to time in effect.
- 1.3.9. Charter. The term "Charter" shall mean the certificate or articles of incorporation or organization, statute, constitution, joint venture or partnership agreement or articles or other charter documents of any Person (other than an individual), each as from time to time in effect.
- 1.3.10. Company Marks. The term "Company Marks" shall mean the name "Marshalls" and those trademarks required to be listed on Schedule 4.6, and all variations of the foregoing.
- 1.3.11. Code. The term "Code" shall mean the Internal Revenue Code of 1986, as amended.
- 1.3.12. Compensation. The term "Compensation", as applied to any Person, shall mean all salaries, compensation, remuneration or bonuses of any character, and medical, surgical, dental, hospital, disability, unemployment, retirement, pension, vacation, insurance or fringe benefits of any kind, or other payments or benefits of any kind whatsoever made or provided directly or indirectly by or on behalf of the Company to such Person or members of the immediate family of such Person.
- 1.3.13. Contractual Obligation. The term "Contractual Obligation" shall mean, with respect to any Person, any contract, agreement, deed, mortgage, lease, sublease, license, indenture, Guarantee, commitment, undertaking or arrangement, written or oral, or other consensual document or instrument, including, without limitation, any document or instrument evidencing or otherwise relating to any indebtedness but excluding the Charter and By-laws of such Person, to which or by which such Person is a party or otherwise subject or bound or to which or by which any property or right of such Person is subject or bound.
- 1.3.14. Controlled Group. The term "Controlled Group", with respect to any Person, shall mean any Person which is a member of the same "controlled group", or under "common control", within the meaning of Section 414(b) or (c) of the Code or Section 4001(b) of ERISA, with such Person.

- 1.3.15. Debt. "Debt" of any Person shall mean all obligations of such Person (i) in respect of indebtedness for borrowed money, (ii) evidenced by notes, bonds, debentures or similar instruments, (iii) for the deferred purchase price of goods or services (other than trade payables or accruals incurred in the Ordinary Course of Business), (iv) under capital leases and (v) in the nature of Guarantees of the obligations described in clauses (i) through (iv) above of any other Person.
- 1.3.16. Distribution. The term "Distribution" shall mean, with respect to the capital stock of or other equity interests in any Person, (i) the declaration or payment of any dividend on or in respect of any shares of any class of such capital stock or in respect of any such equity interest; (ii) the purchase, redemption or other retirement of any shares of any class of such capital stock or of any such equity interest, directly, or indirectly through a Subsidiary or otherwise; and (iii) any other distribution on or in respect of any shares of any class of such capital stock or on or in respect of any such equity interest.
- 1.3.17. Enforceable. The term "Enforceable" shall mean, with respect to any Contractual Obligation, that such Contractual Obligation is the legal, valid and binding obligation of the Person in question, enforceable against such Person in accordance with its terms.
- 1.3.18. Environmental Laws. The term "Environmental Laws" shall mean any Federal, state or local law relating to (i) releases or threatened releases of Hazardous Substances; (ii) the manufacture, handling, transport, use, treatment, storage or disposal of Hazardous Substances or materials containing Hazardous Substances; or (iii) otherwise relating to pollution of the environment or the protection of human health.
- 1.3.19. Equity Securities. The term "Equity Securities" shall mean, with respect to any Person that is not a natural person, all shares of capital stock or other equity or beneficial interests issued by or created in or by such Person, all stock appreciation or similar rights or grants of, or other Contractual Obligation for, any right to share in the equity, income, revenues or cash flow of such Person, and all securities or other rights, warrants or other Contractual Obligations to acquire any of the foregoing, whether by conversion, exchange, exercise or otherwise.
- 1.3.20. Excluded Liabilities. The term "Excluded Liabilities" shall mean all Liabilities (other than Liabilities for Taxes, which are referred to in Section 6.7) whenever arising with respect to any of the following (whether or not such matters are the subject of any representation or warranty made by or on behalf of Seller herein or are disclosed in any Schedule hereto or otherwise disclosed to Buyer):

- (i) operations and assets discontinued at any time on or prior to the Closing Date or leased or subleased or otherwise transferred to Persons other than the Company and its Subsidiaries at any time on or prior to the Closing Date or stores closed at any time on or prior to the Closing Date (including, without limitation, Liabilities for severance and other employee costs, lease and other occupancy costs and contingent lease obligations);
- (ii) product liability claims whenever arising from products sold on or prior to the Closing Date;
- (iii) general, automobile or public liability claims (whether or not insured, including as a result of deductibles, retentions or limits in any insurance policies), other than product liability claims, arising from occurrences on or prior to the Closing Date;
- (iv) any Action pending or, to the knowledge of Seller, threatened in writing against the Company or any Subsidiary on or prior to the Closing Date (including without limitation the matter disclosed on Schedule 4.17 related to the Americans with Disabilities Act), except insofar as any such Action is brought as a result of the transactions contemplated hereby;
- (v) any Liability for which Seller is responsible under Section 9;
- (vi) corporate owned life insurance directly or indirectly benefitting Seller or any of its Affiliates and any related employee or death benefits;
- (vii) the environmental contamination on or adjacent to or emanating from the portion of the property leased by Gasolinas de Puerto Rico at the Bayamon, Puerto Rico Store No. 626 of the Company or any Subsidiary and on or adjacent to or emanating from the portion of the property leased by Gasolinas de Puerto Rico or Isla Petroleum Corporation at the Carolina, Puerto Rico owned Real Property, in each case as reported in certain environmental site assessment reports previously provided by Seller to Buyer (and including any contamination arising from the spread, migration or leaching of such reported contamination);
- (viii) environmental contamination emanating from (y) a 5,000 gallon underground storage tank removed in 1993 and located at the Company's warehouse/distribution center in Woburn, Massachusetts (but only to the extent the Liabilities resulting from the matters described in this clause (y) exceed \$50,000), and (z) two underground storage tanks removed in 1983 and a sandy area possibly used for the disposal of waste located at the Company's warehouse/distribution center in Chelmsford, Massachusetts (but only to the

extent the Liabilities resulting from the matters described in this clause (z) exceed \$50,000).

- (ix) any Liability of the Company or any Subsidiary to Seller or any of its Affiliates in respect of Affiliate Debt or in respect of other obligations, states of facts or conditions in existence at or prior to the Closing Date, except for (A) Liabilities incurred after the Closing Date under agreements or leases to be entered into between the Company or its Subsidiaries, on the one hand, and Seller and its Affiliates, on the other hand, pursuant to Sections 6.20 and 7.2(i) and (B) any Guarantee by Seller or its Affiliates of any Lease or the Guarantee of Seller disclosed in Schedule 4.8(e);
- (x) except for capital lease obligations set forth in the Interim Balance Sheet and incurred since such date in the Ordinary Course of Business and in amounts consistent with the Budget, any Liability of the Company or any Subsidiary in respect of Debt incurred prior to the Closing (other than Debt owed solely to the Company or any wholly owned Subsidiary) or any other Liability of the Company or any Subsidiary incurred prior to, or arising out of the conduct of the Business prior to, the Closing of a category properly classified as long-term in accordance with generally accepted accounting principles consistently applied by the Company (and including the current portion of any such long-term Liability and the remaining portion of any Liability that when incurred or arising should have been properly classified as a long-term Liability in accordance with generally accepted accounting principles and as to which only the current portion remains); and
- (xi) all Liabilities of a category set forth under the caption "Items to be Assumed/Retained by Seller" in Schedule 3.4B, including Appendix A thereto (whether or not any such Liability would be required to be set forth on a balance sheet prepared as of the Closing Date in accordance with generally accepted accounting principles).
- 1.3.21. ERISA. The term "ERISA" shall mean the federal Employee Retirement Income Security Act of 1974 or any successor statute, and the rules and regulations thereunder, and in the case of any referenced section of any such statute, rule or regulation, any successor section thereto, collectively and as from time to time amended and in effect.
- 1.3.22. Generally Accepted Accounting Principles. The term "generally accepted accounting principles" shall mean generally accepted accounting principles, as in effect on the date hereof.

- 1.3.23. Governmental Authority. The term "Governmental Authority" shall mean any U.S. federal, state or local or any foreign government, governmental authority, regulatory or administrative agency, governmental commission, court or tribunal (or any department, bureau or division thereof) or any arbitral body.
- 1.3.24. Governmental Order. The term "Governmental Order" shall mean any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.
- 1.3.25. Guarantee. The term "Guarantee", with respect to any Person, shall mean (i) any guarantee of the payment or performance of, or any contingent obligation in respect of, any Debt or other obligation of any other Person, (ii) any other arrangement whereby credit is extended to any other Person on the basis of any promise or undertaking of such Person (A) to pay the Debt of such other Person, (B) to purchase any obligation owed by such other Person, (C) to purchase or lease assets (other than inventory in the ordinary course of business) under circumstances that would enable such other Person to discharge one or more of its obligations, or (D) to maintain the capital, working capital, solvency or general financial condition of such other Person, and (iii) any liability of such Person as a general partner of a partnership or as a venturer in a joint venture in respect of Debt or other obligations of such partnership or venture.
- 1.3.26. Hazardous Substances. The term "Hazardous Substances" shall mean (i) substances which contain substances defined in or regulated under the following Federal statutes, as amended, and their state counterparts, as well as these statutes' implementing regulations as amended from time to time and as interpreted by administering Governmental Authorities: the Hazardous Materials Transportation Act, the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation and Liability Act, the Clean Water Act, the Safe Drinking Water Act, the Asbestos Hazard Emergency Response Act, the Atomic Energy Act, the Toxic Substances Control Act, the Federal Insecticide, Fungicide, and Rodenticide Act, and the Clean Air Act; (ii) petroleum and petroleum products, including crude oil and any fractions thereof; (iii) natural gas, synthetic gas and any mixtures thereof; (iv) radon; (v) PCBs; (vi) asbestos; (vii) any substance with respect to which a Governmental Authority requires environmental investigation, monitoring, reporting or remediation; and (viii) any other hazardous, noxious, radioactive or toxic materials or substances.
- 1.3.27. Income Tax. The term "Income Tax" means any Tax which is, in whole or in part, based on or measured by income or gains.

order, rule, regulation, resolution or promulgation, or any Governmental Order, or any license, franchise, consent, approval, permit or similar right granted under any of the foregoing, or any similar provision having the force and effect of law.

- 1.3.29. Liabilities. The term "Liabilities" shall mean any and all liabilities and obligations, whether accrued, fixed, absolute or contingent, matured or unmatured or determined or determinable, or otherwise.
- 1.3.30. Lien. The term "Lien" shall mean any mortgage, pledge, lien, security interest, charge, attachment, equity or other encumbrance, or restriction on the creation of any of the foregoing, whether relating to any property or right or the income or profits therefrom; provided, however, that the term "Lien" shall not include (i) statutory liens for Taxes to the extent that the payment thereof is not in arrears or otherwise due, (ii) encumbrances in the nature of zoning restrictions, easements, rights or restrictions of record on the use of real property if the same do not detract from the value of the property encumbered thereby or impair the use of such property in the Business as currently conducted or proposed to be conducted, (iii) statutory or common law liens to secure landlords, lessors or renters under leases or rental agreements confined to the premises rented to the extent that no payment or performance under any such lease or rental agreement is in arrears or is otherwise due, (iv) deposits or pledges made in connection with, or to secure payment of, worker's compensation, unemployment insurance, old age pension programs mandated under applicable Legal Requirements or other social security, (v) statutory or common law liens in favor of carriers, warehousemen, mechanics and materialmen, statutory or common law liens to secure claims for labor, materials or supplies and other like liens, which secure obligations to the extent that (A) payment of such obligations is not in arrears or otherwise due and (B) such liens do not and will not, individually or in the aggregate, have a Material Adverse Effect or materially affect the use of any Real Property, and (vi) restrictions on transfer of securities imposed by applicable state and federal securities laws.
- 1.3.31. Losses. The term "Losses" shall mean any and all losses, damages, obligations, Liabilities, claims, awards (including, without limitation, awards of punitive or treble damages or interest), assessments, amounts paid in settlement, judgments, orders, decrees, fines and penalties, costs and expenses (including, without limitation, reasonable legal costs and expenses and costs and expenses of collection).
- 1.3.32. Material Adverse Effect. The term "Material Adverse Effect" shall mean any adverse change in or effect on the business, condition (financial or otherwise), operations, performance, properties or prospects of the Company or any of its Subsidiaries (or of another specified Person) that, when considered either singly or together with all other adverse changes and effects with respect to which such phrase is used in this Agreement, is material to the Company and its Subsidiaries (or to such

other specified Person and its Subsidiaries), taken as a whole; provided,

however, that when such term is used in reference to Buyer, such term shall

not include any change or effect attributable to the Company or any Subsidiary.

- 1.3.33. Ordinary Course of Business. The term "Ordinary Course of Business" shall mean the ordinary course of the Business consistent with past custom and practice.
- 1.3.34. Preferred Stock Subscription Agreement. The term "Preferred Stock Subscription Agreement" shall mean the Preferred Stock Subscription $\,$ Agreement to be entered into by Buyer and Seller at or prior to the Closing in the form of Exhibit A hereto.
- 1.3.35. Person. The term "Person" shall mean any individual, partnership, corporation, association, trust, joint venture, unincorporated organization or other entity, and any Governmental Authority.
- 1.3.36. Standstill and Registration Rights Agreement. The term "Standstill and Registration Rights Agreement" shall mean the Standstill and Registration Rights Agreement to be entered into by Buyer and Seller at or prior to the Closing in the form of Exhibit B hereto.
- 1.3.37. Subsidiary. The term "Subsidiary" shall mean any Person of which the Company (or other specified Person) shall own directly or indirectly at least a majority of the outstanding capital stock (or other shares of equity interest) entitled to vote generally in the election of directors or in which the Company (or other specified Person) is a general partner or joint venturer without limited liability.
- 1.3.38. Taxes. The term "Tax" shall mean any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental taxes under Code Section 59A, customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax, fee, levy, duty, impost or charge of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or
- 1.3.39. Tax Return. The term "Tax Return" shall mean any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

2. ACQUISITION; EXCLUDED LIABILITIES. Upon the terms, subject to the conditions, and in reliance on the representations, warranties and covenants set forth herein, Seller agrees to sell and transfer to Buyer, and Buyer agrees to purchase from Seller, on the Closing Date, all of the Shares. At the Closing, Seller shall retain and assume and remain or become primarily obligated with respect to all Excluded Liabilities.

PAYMENT AND CLOSING.

- 3.1. Transaction Price. (a) In consideration of the sale and transfer of the Shares by Seller to Buyer and of the agreement by Seller to perform each of the other obligations and covenants to be fulfilled or complied with by them hereunder, including, without limitation, the agreement of Seller to perform its obligations and covenants set forth in Section 6.9 hereof, Buyer shall:
 - (i) pay to Seller at the Closing (by wire transfer of immediately available federal funds to an account designated by Seller by notice to Buyer not fewer than three Business Days prior to the Closing Date) an aggregate amount (the "Estimated Cash Purchase Price") equal to \$375,000,000; and
 - (ii) issue and deliver to Seller the Buyer Stock.

At the option of Buyer, Buyer may increase the amount of the Estimated Cash Purchase Price by all or any portion of the proceeds received by Buyer after the date hereof from the sale of any operating Subsidiary or division of Buyer or the sale by Buyer of shares of its common stock or preferred stock mandatorily convertible into common stock. If Buyer elects such option, the amount of Series D Preferred Stock to be issued and delivered at Closing as part of the Buyer Stock shall be reduced (and the Estimated Cash Purchase Price shall be appropriately increased) so that the aggregate liquidation preferences of the outstanding Series D Preferred Stock immediately following issuance of the Series D Preferred Stock actually issued equals the amount such preferences would have been if Buyer had not elected such option minus the amount by which

Buyer has elected to increase the Estimated Cash Purchase Price. For all purposes of this Agreement, the terms Estimated Cash Purchase Price and Buyer Stock shall mean the amounts thereof after giving effect to the provisions of this paragraph.

The Buyer Stock and the Estimated Cash Purchase Price are referred to herein collectively as the "Estimated Purchase Price". The Estimated Cash Purchase Price and the Estimated Purchase Price (but not the Buyer Stock) shall be subject to adjustment following the Closing as set forth in Section 3.4 below. The Estimated Cash Purchase Price and the Estimated Purchase Price as so adjusted are referred to herein as the "Cash Purchase Price" and the "Purchase Price", respectively. The parties agree that the value of the Buyer Stock equals \$175,000,000.

- 3.2. Time and Place of Closing. The closing of the purchase and sale of the Shares and the other transactions contemplated by this Agreement (the "Closing") shall take place at the Conference Center of Ropes & Gray in Boston, Massachusetts at 10:00 a.m. (local time) on November 16, 1995, or at such other time or place upon which the parties may agree, provided that all conditions to Closing have been satisfied or waived as provided in Sections 7 and 8 (the day on which the Closing takes place being referred to herein as the "Closing Date")
- 3.3. Delivery. At the Closing, Seller will convey, transfer and assign the Shares to Buyer free and clear of any Liens (including without limitation restrictions on transfer or voting), and will deliver to Buyer a certificate or certificates evidencing all of the Shares duly endorsed or accompanied by separate stock power(s) duly endorsed, in each case with signature(s) guaranteed, with all required stock transfer Tax stamps affixed and in form proper for transfer, against delivery by Buyer of the Estimated Purchase Price as set forth in Section 3.1 above.

3.4. Cash Price Adjustment.

(a) As promptly as possible following the Closing Date, the Company shall prepare a consolidated balance sheet of the Company and its Subsidiaries as of a time immediately prior to the Closing (the "Closing Balance Sheet") in accordance with generally accepted accounting principles applied consistently with the Company's past practices used in the preparation of the Annual Financials, except that inventory will be determined using the first-in first-out inventory cost method and without regard to the Company's adoption of Financial Accounting Standard No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of," and the change in the Company's accounting policy with respect to the capitalization of internally developed software "Accounting Policy Changes"). During the fourteen (14) days preceding the Closing Date, RGIS (the "Inventory Firm") shall conduct a wall-to-wall physical count of all the owned inventory located at the stores, distribution centers and warehouses of the Company and its Subsidiaries. Buyer, Seller and their respective accountants shall have the opportunity to observe the physical count of the inventory. As promptly as practicable and no later than fourteen (14) days following the Closing Date, the Inventory Firm shall deliver to Seller and Buyer a written statement setting forth the counted value of the inventory (the "Inventory Statement"). The inventory reflected on the Closing Balance Sheet shall be calculated in accordance with generally accepted accounting principles consistently applied by the Company based on the Inventory Statement, utilizing the retail method for the store inventory and the cost method for the warehouse and distribution centers as consistently applied by the Company in preparation of the Annual Financials, except inventory cost will be determined using the first-in first-out inventory cost method. Amounts reflected on the Closing Balance Sheet for those elements, accounts or items to be included in the calculation of Company Net Assets shall include all known and estimated assets and

liabilities as of the Closing Date consistent with the Company's fiscal year-end cut-off procedures.

As promptly as possible following the receipt of the unaudited Closing Balance Sheet, Coopers & Lybrand L.L.P. ("Coopers") shall perform procedures agreed upon by the parties and Coopers (as set forth in Appendix B to Schedule 3.4B) in connection with the elements, accounts or items of the Closing Balance Sheet that are to be included in the calculation of Company Net Assets for the purposes of issuing a report (the "Coopers Report") thereon detailing the results of such procedures as applied by Coopers in accordance with standards established by the American Institute of Certified Public Accountants (and prior to the issuance by Coopers of such report, KPMG Peat Marwick and representatives of Seller and the Company reasonably designated by Seller shall have the opportunity to review Coopers' work papers and to be present during the performance of all such procedures). Adjustments proposed by Coopers to the elements, accounts or items of the Closing Balance Sheet to be included in the calculation of Company Net Assets will be aggregated and to the extent the total of such adjustments exceeds \$750,000 on a net basis, such excess adjustments shall be reflected in the Closing Balance Sheet for the purpose of calculating Company Net Assets. Coopers shall furnish the Coopers Report to Seller and Buyer within 75 days following the Closing or as soon thereafter as practicable.

Within 10 days following delivery of the Coopers Report, Coopers shall prepare and deliver a Company Net Assets Statement in substantially the form of Schedule 3.4A, which will include a calculation of the Cash Purchase Price in the form of Appendix A thereto. Assets and liabilities on the Company Net Assets Statement will be equal to such items in the Closing Balance Sheet except as otherwise specified in Schedule 3.4A and will exclude the impact of the Accounting Policy Changes and will reflect property on a gross cost basis. The Company Net Assets Statement will exclude those assets and liabilities detailed in Schedule 3.4B, including Appendix A thereto, under the columns titled "Items to be Assumed/Retained by Seller" and "Other Adjustments". "Company Net Assets" shall mean the net asset figure appearing on the Company Net Assets Statement.

Schedule 3.4B sets forth Company Net Assets on an estimated basis as of October 30, 1995. The "Target Net Asset Amount" shall mean \$968,372,000 (which amount equals the net asset figure shown under the column styled "Estimated Statement of Net Assets" on said Schedule 3.4B).

(b) If the amount of Company Net Assets as set forth on the Company Net Assets Statement is: (i) less than the Target Net Asset Amount, then the amount by which Company Net Assets are less than the Target Net Asset Amount shall be paid to Buyer by Seller; or (ii) greater than the Target Net Asset Amount, then the amount by which Company Net Assets are greater than the Target Net Asset Amount shall be paid

to Seller by Buyer but in no event shall such payment to Seller be greater than \$50 million. Any amount due as a result of the operation of this Section 3.4(b) shall be referred to herein as the "Cash Price Adjustment" and shall be treated for all purposes as an adjustment to the Purchase Price. The Cash Purchase Price shall be the Estimated Cash Purchase Price decreased or increased, as the case may be, by the Cash Price Adjustment.

- (c) If Seller disagrees with the Company Net Assets Statement or the Cash Purchase Price, Seller shall, within twenty (20) Business Days after receipt of the Company Net Assets Statement furnish to Buyer a written statement of such disagreement, together with an explanation of the reasons therefor. If Seller does not furnish such a statement to Buyer within such period, the amount of Company Net Assets set forth on the Company Net Assets Statement and the amount of the Cash Purchase Price derivable therefrom shall be binding and conclusive on all parties hereto. If Seller does furnish such a statement to Buyer within such period, the parties hereto shall first use commercially reasonable efforts to resolve such disagreement among themselves. If the parties are unable to resolve the dispute within 20 calendar days after delivery of such notification, the dispute shall be submitted to the Alternative Accountants for resolution. The parties shall request the Alternative Accountants to resolve the dispute within 30 calendar days after submission. The determination of the Alternative Accountants as to the resolution of any dispute shall be binding and conclusive upon all parties hereto. All determinations pursuant to this Section 3.4(c) shall be in writing and shall be delivered to the parties hereto. Any award made pursuant to this Section 3.4(c) may be entered in and enforced by any court referred to in Section 11.1 hereof and the parties hereby consent and commit themselves to the jurisdiction of any such court for purposes of the enforcement of any such award.
- (d) The fees and expenses of Coopers related to the matters set forth in this Section 3.4 shall be borne by Buyer and the fees and expenses of KPMG Peat Marwick related to the matters set forth in this Section 3.4 shall be borne by Seller. Each party will bear all expenses for any special work performed at its request in connection with the matters that are the subject of this Section 3.4. The fees and costs of the Inventory Firm associated with the taking of the physical inventory shall be shared equally by Buyer and Seller. The fees and expenses of the Alternative Accountants in connection with the resolution of disputes pursuant to paragraph (d) above shall be shared equally by Buyer and Seller.
- (e) The amount of the Cash Price Adjustment shall bear interest at an annual rate equal to the reference rate from time to time of Morgan Guaranty Trust Company of New York plus 1% from and including the Closing Date to, but excluding, the date of payment. The Cash Price Adjustment, together with accrued interest thereon, shall be paid within three Business Days after the later of (i) five Business Days after delivery of the Company Net Assets Statement, including the calculation of the Cash

Purchase Price appended thereto, and (ii) the earlier of the resolution of any dispute by Buyer and Seller following notification of Seller's disagreement to Buyer pursuant to Section 3.4(c) above or a determination by the Alternative Accountants pursuant to Section 3.4(c) above. Any such amount shall be paid by wire transfer of immediately available funds to an account designated by the party to receive such payment.

- (f) The provisions of this Section 3.4 and the Schedules referred to in this Section 3.4 shall be used solely for the purpose of calculating the amount of the Cash Price Adjustment, if any, and shall not affect the ownership of any assets or rights or the responsibility for any liabilities, except to the extent set forth in the definition of Excluded Liabilities.
- 4. REPRESENTATIONS AND WARRANTIES OF SELLER. In order to induce Buyer to enter into and perform this Agreement and to consummate the transactions contemplated hereby, Seller represents and warrants to Buyer as follows:

4.1. Corporate Matters, etc.

- 4.1.1. Incorporation and Authority of Seller. Seller is a corporation duly organized, validly existing and in good standing under the laws of the State of New York and has all requisite power and authority, corporate and otherwise, to enter into this Agreement and each of the Closing Agreements, to carry out and perform its obligations hereunder and to consummate the transactions contemplated hereby.
- 4.1.2. Organization, Power and Standing. The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Minnesota, and each of the Subsidiaries is a corporation duly organized, validly existing and in good standing under the jurisdiction of its incorporation or organization. Each of the Company and each of its Subsidiaries has all requisite power and authority, corporate and otherwise, to carry on the Business as currently conducted, and to consummate the transactions contemplated hereby. Each of the Company and its Subsidiaries is duly qualified or licensed to do business as a foreign corporation or otherwise, and is in good standing as such, in each jurisdiction where the nature of the Company's or such Subsidiary's activities or its ownership or leasing of property require such qualification, except such failures to be so qualified as have not had and will not have a Material Adverse Effect.
- 4.1.3. Authorization and Enforceability. This Agreement has been duly authorized, executed and delivered by Seller and is Enforceable against Seller. Each of the Closing Agreements to which Seller or any of its Affiliates is intended to be party has been duly authorized, and, on or before the Closing Date, will be duly executed and delivered by Seller or its applicable Affiliate and be Enforceable against Seller or such Affiliate.

- 4.1.4. Non-Contravention, etc. No approval, consent, waiver, authorization or other order of, and no filing, registration, qualification or recording with, any Governmental Authority or any other Person (other than any party to any Lease-In other than the Company or any Subsidiary) is required to be obtained or made by or on behalf of Seller or the Company or any of its Subsidiaries in connection with the execution, delivery or performance of this Agreement and the consummation of the transactions contemplated hereby, except for (i) satisfaction of the requirements of the Hart-Scott-Rodino Antitrust Improvements of 1976, as amended (the "HSR Act"), (ii) items listed on Schedule 4.1.4 which shall have been obtained or made and shall be in full force and effect at the Closing (subject to the materiality exception set forth at the end of the next sentence) and (iii) any other of the foregoing items required to be obtained from or made with any Person other than any Governmental Authority which the failure to obtain or make, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect nor a material adverse effect on the ability of Seller to consummate the transactions contemplated hereby. Except as set forth on Schedule 4.1.4, neither the execution, delivery and performance of this Agreement nor the consummation of any of the transactions contemplated hereby (including, without limitation, the execution, delivery and performance of the Closing Agreements) does or will constitute, result in or give rise to (i) a breach or violation or default under any Legal Requirement applicable to Seller, the Company or any of its Subsidiaries, (ii) a breach of or a default under any Charter or By-Laws provision of Seller, the Company or any of its Subsidiaries, (iii) the acceleration of the time for performance of any obligation under any Contractual Obligation (other than any Lease-In) of Seller, the Company or any of its Subsidiaries, (iv) the imposition of any Lien upon or the forfeiture of any Asset, other than any Asset held under any Lease-In, (v) a breach of or a default under any Contractual Obligation (other than any Lease-In) of Seller, the Company or any of its Subsidiaries, or (vi) right to any severance payments other than by operation of law (including without limitation if such payments become due only if employment is terminated following the Closing), termination, right of termination, modification of terms or change in benefits or burdens under any Contractual Obligation (other than any Lease-In), other than in the case of clauses (i), (iii), (iv), (v) and (vi) such as, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect nor a material adverse effect on the ability of Seller to consummate the transactions contemplated hereby.
- 4.1.5. Title, etc. Seller has good and marketable title to all Shares free and clear of any Liens (including without limitation restrictions on transfer or voting). Except for this Agreement, there is no Contractual Obligation pursuant to which Seller has, directly or indirectly, granted any option, warrant or other right to any Person to acquire the Shares or any other securities of the Company or any of its Subsidiaries. Upon delivery of certificates representing the Shares, and delivery of the consideration therefor as herein contemplated, Buyer will receive good and marketable title to the

Shares, free and clear of any Liens (including without limitation restrictions on transfer or voting) and subject to no rescission or similar rights or equities of any kind.

- 4.1.6. Capitalization. The only issued and outstanding shares of capital stock of the Company are the Shares, all of which are duly authorized, validly issued, fully paid and non-assessable, and all of which are beneficially owned and held of record by Seller. There is no Contractual Obligation or Charter or By-law provision that obligates the Company or any Subsidiary to issue, purchase or redeem, or make any payment in respect of, any Equity Security.
- Subsidiaries. Except as indicated therein, the Company has only the Subsidiaries listed in Schedule 4.1.7. Not later than the tenth day after the date hereof, Seller will deliver to Buyer a complete substitute Schedule 4.1.7 that lists no Persons other than those that are Subsidiaries as of the date hereof and sets forth the name and jurisdiction of incorporation or organization, the date of incorporation or organization, the issued and outstanding shares of capital stock, and the federal or foreign taxpayer identification number of each such Subsidiary. The Company is the direct or indirect record and beneficial owner of all of the issued and outstanding shares of capital stock of each of its Subsidiaries, such shares have been duly authorized and validly issued and are fully paid and nonassessable, and the Company or a wholly owned direct or indirect Subsidiary has good and marketable title to such shares free and clear of any Liens (including without limitation restrictions on transfer or voting). There is no outstanding Equity Security of any Subsidiary other than its issued and outstanding shares of capital stock. The Company has no investment in any Person other than (a) its Subsidiaries, (b) demand deposit or money market accounts and (c) advances made to suppliers in the Ordinary Course of Business. No Subsidiary has any investment in any Person other than (a) demand deposit or money market accounts and (b) advances made to suppliers in the Ordinary Course of Business.
- 4.1.8. Charter and By-laws. Seller has heretofore delivered to Buyer a true and complete copy of the Charter and By-laws of the Company and each of its material Subsidiaries, in each case in the form currently in effect and as will be in effect immediately prior to the Closing, and at least ten Business Days prior to the Closing, Seller will have delivered to Buyer a true and complete copy of the Charter and By-laws of each of the other Subsidiaries, in each case in the form then in effect and as will be in effect immediately prior to the Closing.

4.2. Financial Statements, etc.

4.2.1. Financial Information. Buyer has been furnished with true and complete copies of each of the following:

- (a) The unaudited consolidated balance sheet of the Company and its Subsidiaries as of December 31, 1994 (the "Annual Balance Sheet") and the related statement of earnings for the fiscal year then ended, and similar financial statements as at the end of and for each of the preceding two fiscal years (the "Annual Financials" and, together with the Interim Financials and the Monthly Financials, the "Financial Statements").
- (b) The unaudited consolidated balance sheets of the Company and its Subsidiaries as of July 1, 1995 (the "Interim Balance Sheet") and as of July 1, 1994 and related unaudited consolidated statements of earnings of the Company and its Subsidiaries for each of the sixmonth periods then ended (the "Interim Financials").
- (c) Monthly unaudited consolidated statements of earnings of the Company and its Subsidiaries for the months of July and August, 1995, such monthly income statements having been prepared in the form customarily prepared by management for internal use (the "Monthly Financials").
- (d) The adjusted unaudited consolidated balance sheet of the Company and its Subsidiaries as of December 31, 1994 and the related statement of earnings for the fiscal year then ended, and similar financial statements as at the end of and for each of the preceding two fiscal years.
- 4.2.2. Character of Financial Information. The Annual Financials and the Interim Financials were prepared in accordance with generally accepted accounting principles consistently applied by the Company throughout the periods specified therein and present fairly, in all material respects, the consolidated financial position and results of operations of the Company and its Subsidiaries for the periods specified therein, subject to an absence of footnotes and in the case of the Interim Financials to normal year-end audit adjustments which will not in the aggregate be material. Such financial information reflects the Company's accounting policies currently in effect and does not reflect the impact of the Accounting Policy Changes.
- 4.3. Change in Condition. Except for the matters set forth in Schedule 4.3, since the date of the Interim Balance Sheet:
 - (a) The Business has been conducted only in the Ordinary Course of Business (except as otherwise required by the terms of this Agreement), and without limiting the generality of the foregoing, the Company and its Subsidiaries have made capital expenditures in the Ordinary Course of Business and in an aggregate amount consistent with the Budget;

- (b) Neither the Company nor any of its Subsidiaries has:
- (i) made any capital expenditures except routine expenditures for repairs and maintenance and except in an aggregate amount consistent with the budget set forth on Schedule 4.3(b)(i) hereto (the "Budget");
- (ii) incurred or otherwise become liable in respect of any Debt (other than Affiliate Debt, none of which will be outstanding as of the Closing) or become liable in respect of any Guarantee, other than Debt, intercompany advances or any Guarantee between the Company and its wholly owned Subsidiaries or between wholly owned Subsidiaries;
- (iii) mortgaged or pledged an Asset or subjected any Asset to any Lien except Liens disclosed on Schedule 4.5.1;
- (iv) declared or made any Distribution (other than (A) distributions of cash to Seller in the Ordinary Course of Business pursuant to the cash management program of Seller and its Subsidiaries, a true and correct description of which is included on Schedule 4.3(b)(iv)(A) (the "Cash Management Program"); (B) the distributions set forth on Schedule 4.3(b)(iv)(B); (C) distributions of cash or of any receivable constituting Affiliate Debt in connection with the repayment or cancellation of Affiliate Debt; and (D) distributions or contributions in connection with an increase in or the repayment or cancellation (in whole or in part) of Debt or intercompany advances between the Company and its wholly owned Subsidiaries or between wholly owned Subsidiaries);
- (v) sold, leased to others or otherwise disposed of any of its Assets (except for sales of inventory in the Ordinary Course of Business and except as contemplated by clause (iv) of this Section 4.3(b));
- (vi) purchased any Equity Security of any Person other than of a direct or indirect wholly owned Subsidiary of the Company, or any assets (other than inventory) material in amount or constituting a business, or been party to any merger, consolidation or other business combination or entered into any Contractual Obligation relating to any such purchase, merger, consolidation or business combination;
- (vii) made any loan, advance or capital contribution to or investment in any Person other than loans, advances or capital contributions to or investments in or to the Company or its wholly owned Subsidiaries and other than loans or advances to suppliers in the Ordinary Course of Business;

- (viii) canceled or compromised any Debt or claim other than in the Ordinary Course of Business and other than any Affiliate Debt or any Debt, intercompany advances or claim between the Company and a wholly owned Subsidiary or between wholly owned Subsidiaries;
- (ix) sold, transferred, licensed or otherwise disposed of any Intangibles other than in the Ordinary Course of Business;
- (x) made or agreed to make any material change in its customary methods of accounting or accounting practices except for the Accounting Policy Changes;
- (xi) engaged in or become obligated in respect of any transaction with Seller or any Affiliate of Seller, except in accordance with the Cash Management Program or as described on Schedule 4.3(b)(iv)(B) or Schedule 4.10.
- (xii) waived or released or permitted to lapse any right of material value except in the Ordinary Course of Business; or
- $\mbox{(xiii)}\mbox{\ instituted, settled or agreed to settle any material Action;}$
- (c) Neither the Company nor any of its Subsidiaries has (i) had any material change in its relationships with its employees, agents, customers or suppliers, or (ii) made any changes in the rate of Compensation payable (or paid or agreed or orally promised to pay, conditionally or otherwise, any extra Compensation) to any director, officer, manager, employee, consultant or agent of the Company (other than increases granted in the Ordinary Course of Business and consistent with past practices, which increases will not have a Material Adverse Effect);
- (d) There has been no amendment of any material provision of any Equity Security of the Company or any Subsidiary;
- (e) Neither Seller nor any of its Affiliates nor the Company nor any of its Subsidiaries has entered into any Contractual Obligation (and Seller and its Affiliates have not entered into any Contractual Obligation obligating the Company or any of its Subsidiaries) to do any of the things referred to in clauses (a) through (d) above with respect to the Company, any Subsidiary or the Business; and
 - (f) No Material Adverse Effect has occurred.

- 4.4. Liabilities. Neither the Company nor any of its Subsidiaries has any Liabilities, other than, to the extent the existence thereof is consistent with all other representations and warranties of Seller:
 - (a) set forth on the Interim Balance Sheet;
 - (b) incurred in usual amounts since the date of the Interim Balance Sheet in the Ordinary Course of Business;
 - (c) in respect of the Leases and Contracts; or
 - (d) in respect of purchase orders or similar Contractual Obligations for the purchase of inventory in the Ordinary Course of Business.
 - (e) between the Company and a wholly owned Subsidiary or between wholly owned Subsidiaries;
 - (f) any Excluded Liabilities; or
 - (g) other undisclosed Liabilities which, individually or in the aggregate, are not material to the Company or its Subsidiaries, taken as a whole.

4.5. Assets.

4.5.1. Title to Assets; Owned Real Estate. The Company and its Subsidiaries have good and marketable title to, or, in the case of property held under lease or other Contractual Obligation, a valid and enforceable right to use under an Enforceable Lease or License, all of their properties, rights and assets, whether real, personal or intellectual and whether tangible or intangible (collectively, the "Assets"), including, without limitation, all properties, rights and assets reflected in the Interim Balance Sheet or acquired after the date of the Interim Balance Sheet (except as sold or otherwise disposed of since the date of the Interim Balance Sheet in the Ordinary Course of Business or as otherwise permitted by this Agreement to be disposed of since the date of the Interim Balance Sheet). Schedule 4.5.1 contains a true, correct and complete list of all real property and improvements (collectively, "Real Property") which are owned by the Company or any Subsidiary. Seller has furnished to Buyer true and correct copies of all title reports and title insurance policies with respect to the Real Property listed in Schedule 4.5.1. The Assets are not subject to any Lien except as described in said Schedule. The Assets (including, without limitation, Real Property, the Intangibles, the Leases and the Contracts), together with the services to be provided to the Company after the Closing pursuant to the Transitional Services Agreement, constitute all properties, rights and Assets held for or used in or necessary for the conduct of the Business as currently conducted.

4.5.2. Real Property Leases.

Schedule 4.5.2(a) sets forth a true, correct and complete list of each Real Property location which is leased or subleased, or which has been agreed to be leased or subleased, as lessee or sublessee by the Company or any Subsidiary (all of the leases, subleases or other Contractual Obligations pursuant to which such Real Property locations are held or are to be held being referred to herein collectively as the "Leases-In"). Section A of said Schedule lists for each Lease-In pertaining to the retail department stores the store ID number, the name of the lessee, and the city and state of its location. Section B of said Schedule lists Leases-In pertaining to the home office facilities. Section C of said Schedule lists Leases-In pertaining to regional and local offices, distribution facilities and local warehouses. Section D of such Schedule lists those other properties in which the Company or a Subsidiary retains an interest. Sections B, C and D include information comparable to that required in Section A. Schedule 4.5.2(b) sets forth a true, correct and complete list of each lease, sublease or other Contractual Obligation under which the Company or a Subsidiary is a lessor or sublessor of any Real Property (together with the Leases-In, the "Leases") and includes information comparable to that required in Schedule 4.5.2(a). True, correct and complete copies of the Leases, and all material amendments, modifications and supplemental agreements thereto, have been previously delivered to Buyer. Each Lease to which Seller or an Affiliate of Seller is a party is noted on Schedule 4.5.2(a) and 4.5.2(b). All of the retail department stores operated in the Business are owned or leased as lessee or subleased as sublessee by the Company or a Subsidiary.

Except as set forth on Schedules 4.5.2 to the best of Seller's knowledge:

- (i) each Lease is an Enforceable agreement of the Company or the Subsidiary party thereto, and Seller does not have any knowledge that any Lease is not an Enforceable agreement of the other parties thereto, other than as a result of or arising out of the transactions contemplated hereby;
- (ii) the Company or the Subsidiary party thereto has fulfilled all material obligations required pursuant to the Leases to have been performed by the Company or the Subsidiary party thereto on its part, other than as a result of or arising out of the transactions contemplated hereby;
- (iii) neither the Company nor the Subsidiary party thereto is in material breach of or default under any Lease, and no event has occurred which with the passage of time or giving of notice or both would constitute such a default, result in a loss of rights or result in the creation of any Lien thereunder or pursuant thereto (other than as a result of or arising out of the transactions contemplated hereby):

- (iv)(A) there is no existing material breach or default by any other party to any Lease, and (B) no event has occurred which with the passage of time or giving of notice or both would constitute a material default by such other party, result in a loss of rights or result in the creation of any Lien thereunder or pursuant thereto;
- (v) neither the Company nor any Subsidiary is obligated to pay any material leasing or brokerage commission as a result of the transaction contemplated hereby; and
- (vi) there is no pending or threatened eminent domain taking affecting any of the properties which are the subject of the Leases.
- 4.6. Intellectual Property Rights. Schedule 4.6 lists and identifies all trade and product names; patents, patent applications, trademarks, service marks, logos and copyrights (including registrations and applications); trade secrets, know-how; computer software; and all other intellectual or intangible property and rights; in each case that are directly or indirectly owned, licensed or otherwise used by and are material to the Company or any Subsidiary (the "Intangibles", which term shall also include any other proprietary or confidential information that is directly or indirectly owned, licensed or otherwise used by the Company or any Subsidiary) and identifies the owner and any licensee or other user thereof; provided, however, that there have been only generally described on such Schedule such of the Intangibles (including trade secrets, know-how and computer software) as are not readily susceptible to Schedule 4.6 also lists and identifies the term of each license or other Contractual Obligation (including all amendments) under which any material Intangible is held or used by the Company or any Subsidiary in the conduct of the Business or otherwise (the "Licenses") and identifies the owner and any licensee or other user thereof. Except as disclosed on Schedule 4.6, all Intangibles are owned solely by the Company or a Subsidiary or licensed to the Company or its Subsidiaries under an exclusive perpetual License not requiring payment of any royalty or fee (other than in the case of licenses of software in the Ordinary Course of Business). Except as set forth on Schedule 4.6, there is no material license or other Contractual Obligation under which the Company or any Subsidiary is liable as licensor with respect to any Intangibles and neither the Company nor any Subsidiary has granted any material license to any third party with respect to any Intangible. Except as set forth on Schedule 4.6 and to the Seller's knowledge, the use or sale by the Company and its Subsidiaries of any products or services in the Business and use by the Company and its Subsidiaries of the Intangibles does not infringe and has not infringed any rights of any third party, and no activity of any third party infringes upon the rights of the Company or any Subsidiary with respect to any of the Intangibles. Except as set forth on Schedule 4.6, no Action alleging or relating to any such infringement against the rights of the Company or any Subsidiary or any third parties is currently pending or, to the knowledge of Seller, threatened.

- 4.7. Accounts; Funds, etc. Subject to the provisions of Section 6.13, after the Closing Date, all monies and bank or other depository accounts arising out of, relating to or established for the Business or the Company or any Subsidiary shall be held by, and accessible only to, the Company or the relevant Subsidiary; provided, however, that all such monies or accounts in respect of change funds shall be held by, and accessible only to, the Company or the relevant Subsidiary on or after the Closing Date. Schedule 4.7 identifies each bank account or similar account for the deposit of cash or securities maintained by or on behalf of the Company or any Subsidiary (indicating the name and address of the bank or other financial institution, the account name and number and the individuals with signing authority with respect to such account).
- 4.8. Certain Contractual Obligations. Set forth on Schedule 4.8 is a true and complete list of all of the following Contractual Obligations of the Company or any Subsidiary:
 - (a) All collective bargaining agreements and other labor agreements; all employment or material consulting agreements; and all other plans, agreements, arrangements or practices which constitute Compensation or benefits to any of the directors, officers or employees of the Company or any Subsidiary, except to the extent any of the foregoing constitute a Company Plan or Benefit Arrangement;
 - (b) All Contractual Obligations under which the Company or any Subsidiary is or may become obligated to pay any legal, accounting, brokerage, finder's or similar fees or expenses in connection with, or incur any severance pay or special Compensation obligations which would become payable by reason of, this Agreement or the consummation of the transactions contemplated hereby;
 - (c) All Contractual Obligations under which the Company or any Subsidiary is or will after the Closing be restricted from carrying on any business or other activities anywhere in the world;
 - (d) All Contractual Obligations (including, without limitation, options) to: (i) sell or otherwise dispose of any Assets except in the Ordinary Course of Business or (ii) purchase or otherwise acquire any material property or properties or other assets except pursuant to purchase orders for inventory and other arrangements with suppliers in the Ordinary Course of Business;
 - (e) All Contractual Obligations under which the Company or any Subsidiary has any liability for Debt or obligation for Debt or constituting or giving rise to a Guarantee of any liability or obligation of any Person (other than any Lease, any Debt or intercompany advances between the Company and any wholly owned Subsidiary or between wholly owned Subsidiaries, or any Affiliate Debt), or under which any Person has any liability or obligation constituting or giving rise to a Guarantee of any liability

or obligation of the Company or any Subsidiary (including, without limitation, partnership and joint venture agreements) other than any Guarantee by Seller or any of its Affiliates of any Lease or the Guarantee of Seller disclosed in Schedule 4.8(e), or under which any default could arise or penalty or payment could be required in the event of any action or inaction of Seller or any of its Affiliates other than any Guarantee by Seller or its Affiliates of any Lease or the Guarantee of Seller disclosed in Schedule 4.8(e);

- (f) Any lease or other Contractual Obligation under which any tangible personal property other than inventory (the "Equipment") having a cost or capital lease obligation in excess of \$250,000 is held or used by the Company or any Subsidiary;
- (g) Any Contractual Obligation under which the Company or any Subsidiary may become obligated to pay any amount in excess of \$500,000 in respect of indemnification obligations or purchase price adjustment provisions in connection with any (i) acquisition or disposition of assets or securities, real property or of property constituting a product line, (ii) other acquisition or disposition of assets other than sales of inventory in the Ordinary Course of Business, (iii) assumption of liabilities or warranty, (iv) settlement of claims, (v) merger, consolidation or other business combination, or (vi) series or group of related transactions or events of a type specified in subclauses (i) through (v); and if with respect to any such Contractual Obligation there exists any pending or, to the knowledge of Seller, threatened Action that could reasonably be expected to result in the Company and its Subsidiaries being liable to pay an amount in excess of \$100,000 or there currently exist circumstances that would reasonably be expected to give rise to such an Action, such Action or circumstances are described on Schedule 4.17; and
- (h) Any other Contractual Obligation of a type not specifically covered in clauses (a) through (g) above entered into other than in the Ordinary Course of Business or which in the case of such other Contractual Obligations individually is likely to involve payments by or on behalf of, or to, the Company or any of its Subsidiaries in excess of \$250,000 during the calendar year ended December 31, 1995 or \$500,000 over the remaining term of such Contractual Obligation or the termination of which may reasonably be expected to require payments by the Company or any of its Subsidiaries exceeding \$250,000 (other than (i) purchase orders for inventory and other arrangements with suppliers entered into in the Ordinary Course of Business, (ii) agreements pertaining to common area maintenance entered into in the Ordinary Course of Business, (iii) concessionaire and other store related contracts applicable to fewer than 10 stores entered into in the Ordinary Course of Business, and (iv) purchase and lease commitments, construction contracts and other capital expenditure and maintenance and repair commitments reflected in the Company's capital expenditure and maintenance and repair budgets and entered into in the Ordinary Course of Business).

Seller has heretofore delivered to Buyer a true and complete copy (or, in the case of oral contracts or arrangements, a full and accurate written summary) of each of the Contractual Obligations listed on Schedule 4.8, each as in effect on the date hereof and (except as otherwise required by this Agreement) as it will be in effect at the Closing, including, without limitation, all amendments (such Contractual Obligations required to be listed on Schedule 4.8, together with the Licenses, and Insurance Policies, but excluding the Company Plans and Benefit Arrangements, being referred to herein collectively as the "Contracts"). Contract is Enforceable by the Company or the Subsidiary party thereto, against each Person (other than the Company or such Subsidiary) party thereto, except: (i) as otherwise required by the terms of this Agreement and (ii) as such enforceability may be limited by or as a result of (A) bankruptcy, insolvency, reorganization or other similar laws affecting creditors' rights generally, (B) general principles of equity (whether considered in a proceeding at law or in equity) and (C) execution, delivery and performance of this Agreement and the Closing Agreements. No material breach or default by the Company or any Subsidiary under any of the Contracts has occurred and is continuing, and no event has occurred or circumstance exists which with notice or lapse of time would constitute such a breach or default or permit termination, modification or acceleration by any other Person under any of the Contracts or would result in a loss of rights or creation of any lien, charge or encumbrance thereunder or pursuant thereto except as would arise from execution, delivery and performance of this Agreement and the Closing Agreements. To the knowledge of Seller, no material breach or default by any other Person under any of the Contracts has occurred and is continuing, and no event has occurred or circumstance exists that with notice or lapse of time would constitute such a breach or default or permit termination, modification or acceleration by the Company or any Subsidiary under any of the Contracts or would result in a loss of rights or creation of any lien, charge or encumbrance thereunder or pursuant thereto except as would arise from execution, delivery and performance of this Agreement and the Closing Agreements.

4.9. Insurance, etc. All material properties of the Company and its Subsidiaries are covered by valid and currently effective insurance policies $\frac{1}{2}$ issued in favor of the Company and its Subsidiaries or are self-insured in amounts that are customary in the place in which such properties are located for companies operating similar businesses and operations. All insurance coverages required under the Leases are in force under valid and currently effective insurance policies. Set forth on Schedule 4.9(a) is a summary of all policies or binders of insurance by which the Company or any Subsidiary (or any risk of the Business) is insured (the "Insurance Policies") together with (i) the name and telephone number of the agent or broker, (ii) the name of the insurer and the names of the principal insured and each named insured, (iii) the policy number and period of coverage, the type, scope (including an indication of whether the coverage is on a claims made, occurrence or other basis) and amounts (including a description of how deductibles, retentions, aggregates and retroactive premium adjustments or other loss-sharing arrangements are calculated and operate) of coverage. Seller has provided Buyer with true and correct copies of all Insurance Policies in the forms that are available to Seller. All premiums due under such policies have been paid,

and none of Seller, the Company, any Subsidiary or any of Seller's other Affiliates is in default with respect to its respective obligations under any of such policies.

- 4.10. Transactions with Affiliates. Except for the matters specified in Schedule 4.10 (the "Affiliate Relationships"), none of Seller or any of its Affiliates is an officer, director, employee, consultant, distributor, supplier or vendor of, or is party to any Contractual Obligation with, the Company or any Subsidiary, and after the Closing neither the Company nor any Subsidiary will have any liability or obligation to or for the benefit of Seller or any of its Affiliates except pursuant to the Transitional Services Agreement. Except for the matters specified in Schedule 4.10, there are no Assets (including without limitation Intangibles, franchises and know-how) that Seller or any of its Affiliates owns or is licensed or otherwise has the right to use which are used in or necessary to the conduct of the Business.
- 4.11. Compliance with Laws, etc. The operations of the Business as heretofore or currently conducted were not since June 30, 1991 and are not in violation of, nor is the Company or any Subsidiary in default under, any Legal Requirement, except for such violations or defaults listed on Schedule 4.11 or as have not had and will not have individually or in the aggregate a Material Adverse Effect. The Company and its Subsidiaries have been duly granted and continue to hold, and at the Closing will hold, all licenses, permits, consents, approvals, franchises and other authorizations under any Legal Requirement or trade practice necessary for the conduct of the Business as currently conducted (the "Permits"), except such as have not had and will not have individually or in the aggregate a Material Adverse Effect. All of the Permits are now and after giving effect to the Closing will be in full force and effect, except such as will not have a Material Adverse Effect. Schedule 4.11 includes a list of all Permits and applications therefor that are material to the Business. Neither the Company nor any Subsidiary nor Seller has received any notice that any Governmental Authority or other licensing authority or association will revoke, cancel, rescind, materially modify or refuse to renew in the ordinary course any of the Permits, which individually or in the aggregate could reasonably be expected to have a Material Adverse Effect.
- 4.12. Tax Matters. Except as set forth in Schedule 4.12 or the Interim Balance Sheet:
 - (i) To Seller's knowledge, (a) all material Tax Returns required to be filed on or before the date hereof with respect to any Pre-Closing Tax Period (as defined in Section 6.7(c)) by, or with respect to the Company or any Subsidiary have been duly and timely filed (taking into account extensions); (b) no position is reflected in a Tax Return referred to in (a) for which the applicable limitation period has not expired (and for which a closing agreement has not been entered into) which (x) was not, at the time such Tax Return was filed, supported by substantial authority (as determined for purposes of Section 6662 of the Code, or any predecessor provision, and any comparable provisions of applicable federal, state, or local tax statutes, rules or regulations) and (y) would have a Material Adverse Effect if decided against the taxpayer; (c) the Company and its Subsidiaries have timely paid, withheld or made

provision for all Taxes shown as due and payable on any Tax Return and have timely paid, withheld, or made provision for all material Taxes, whether or not shown on any Tax Return; (d) no Liens for Taxes upon the assets of the Company or any Subsidiary exist; (e) neither the Company nor any of its Subsidiaries currently is the beneficiary of any extension of time within which to file any Tax Return; and (f) no claim has ever been made by an authority in a jurisdiction where any of the Company and its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.

- (ii) The Company and each of its Subsidiaries other than New York Department Stores de Puerto Rico, Inc. is a member of the affiliated group, within the meaning of Section 1504(a) of the Code, of which Seller is the common parent (the "Affiliated Group"), and such Affiliated Group files a consolidated federal Income Tax Return. Since the acquisition of Marshalls, Inc. by Seller and Brandon's Plantation, Inc., Brandon's Kendall, Inc., Brandon's Town Center, Inc., Brandon's N. Miami, Inc. and Brandon's S. Miami, Inc., neither the Company nor any of its Subsidiaries has at any time been a member of an affiliated group filing a consolidated federal Income Tax Return other than a group, the common parent of which is Seller. To Seller's knowledge, all Income Taxes shown on any Tax Return of the Affiliated Group have been paid for each taxable period during which any of the Company and its Subsidiaries was a member of the Affiliated Group.
- (iii) To Seller's knowledge, each of the Company and its Subsidiaries has withheld and paid all material Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder, foreign person, or other third party.
- (iv) There is no dispute or claim concerning any material Tax liability of any of the Company and its Subsidiaries either (A) claimed or raised by any taxing authority in writing or (B) as to which Seller has knowledge based upon personal contact with any agent of such taxing authority. Schedule 4.12 lists all federal, state, local, and foreign Income Tax Returns filed with respect to the Company or any of its Subsidiaries for taxable periods ended on or after December 31, 1992, indicates those Tax Returns that have been audited, and indicates those Tax Returns that currently are the subject of audit. Seller has delivered to Buyer correct and complete copies of all portions of federal Income Tax Returns and examination reports which pertain to the Company and its Subsidiaries, and statements of deficiencies assessed against or agreed to by any of the Company and its Subsidiaries since December 30, 1992.
- (v) To Seller's knowledge, neither the Company nor any of its Subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.

(vi) Neither the Company nor any of its Subsidiaries has made any payments, is obligated to make any payments, or is a party to any agreement that under certain circumstances could obligate it to make any payments that will not be deductible under Code Section 280G.

4.13. [Intentionally Omitted.]

4.14. Employee Benefit Plans.

- (a) Schedule 4.14 lists all Company Plans and Benefit Arrangements. The following have previously been delivered to Buyer: (i) true and complete copies of all Company Plans and Benefit Arrangements that have been reduced to writing, together with all amendments; (ii) written summaries of the material terms of all other Company Plans and Benefit Arrangements and related amendments; (iii) in the case of each Company Plan $\,$ and Benefit Arrangement, copies of each of the following to the extent applicable: summary plan descriptions and similar employee summaries (including employee handbooks), trust agreements, and insurance contracts; (iv) in the case of each Company Plan or Benefit Arrangement for which a Form 5500 Series annual report is required to be filed, a copy of the most recent such annual report, together with all schedules, attachments, and related opinions; (v) in the case of each Company Plan or Benefit Arrangements intended to be qualified under Section 401(a) of the Code, a copy of the most recent Internal Revenue Service ("IRS") determination letter (plus, if a request for determination is currently pending, a copy of the request); and (vi) in the case of each Company Plan or Benefit Arrangement, copies of any correspondence from or to the IRS, the Department of Labor ("DOL"), or other U.S. government department or agency relating to an audit or penalty assessment with respect to such Plan or Arrangement or relating to requested relief from any liability or penalty (including, but not limited to, any correspondence relating to the IRS's Voluntary Compliance Resolution Program or Closing Agreement Program, or the DOL's amnesty programs for late filers and non-filers). Except as indicated on Schedule 4.14, all Company Plans and Benefit Arrangements are maintained by Melville Corporation.
- (b) Section 401(a)(11) of the Code does not apply to any participant in Seller's 401(k) Profit Sharing Plan (as defined in Section 9.1(f)).
- (c) There have been no prohibited transactions (within the meaning of Section 406 of ERISA or Section 4975 of the Code) that have resulted or could result in the Company, any of its Subsidiaries, or any of their employees being held liable for a civil penalty under Section 502 or ERISA or an excise tax under Section 4975 of the Code.

- (d) Each of the Company Plans and the Benefit Arrangements has been maintained and administered in all material respects in accordance with its terms and applicable law (including the provisions of the Code). No event has occurred that has resulted or could result in the Company or any of its Subsidiaries being liable with respect to a Tax under Chapter 43 of the Code
- (e) Neither the Company nor any of its Subsidiaries has any obligation or liability with respect to any plan (including, without limitation, any multiemployer plan) subject to Part 3 of Title I of ERISA, Title IV of ERISA, or Section 412 of the Code.
- (f) All required contributions to, premium payments or assessments on account or, and benefit payments under each Company Plan and each Benefit Arrangement that are due or accruable through the Closing Date, for which the Company or any Subsidiary is or may be liable, have been paid or properly accrued.
- (g) Except as set forth in Schedule 4.14, there are not pending or, to the knowledge of Seller, threatened actions or other controversies relating to any Company Plan or Benefit Arrangement, other than claims for benefits in the normal course.
- (h) Other than as required under Section 601 et seq. of ERISA or as specifically set forth in Schedule 4.14, no Company Plan or Benefit Arrangement provides benefits or coverage in the nature of severance or health or life insurance following retirement or other termination of employment. No event has occurred that has resulted or could result in a loss of any deduction to the Company or any of its Subsidiaries under Section 162(n) of the Code. Except for individual employment agreements, neither the Company nor any Subsidiary is obligated under any Company Plan or Benefit Arrangement that cannot be amended, terminated or modified prospectively without the consent of any Employee or beneficiary.

4.15. Environmental Matters, etc.

(a) Except as set forth in Schedule 4.15, or except as has not had and will not have a Material Adverse Effect, the Company and each Subsidiary is and has at all times been in compliance in all respects with all Environmental Laws and any other applicable Legal Requirements relating to environmental, natural resource, health or safety matters. Except as set forth in Schedule 4.15, the Company has not received notice of any Action pending against the Company or any Subsidiary nor, to the knowledge of Seller, is there any basis for any Action or is any Action threatened, in each case in respect of (i) noncompliance by the Company or any Subsidiary with any Environmental Laws or any such Legal Requirement, or (ii) the presence or release or threatened release into the environment of any Hazardous Substance whether or not generated by the Company or any Subsidiary or located at or about or emanating from

or to a site included in the Real Property currently or heretofore owned, leased or otherwise used by the Company or any Subsidiary or any predecessor entity.

- (b) Except as set forth in Schedule 4.15 and for any other matters that would not result in any material liability to the Company and its Subsidiaries taken as a whole, no event has occurred or condition exists or operating practice is being employed that could give rise to any Liability or Losses on the part of the Company or any Subsidiary (or, after the Closing, Buyer) either at the present or at any future time (including, without limitation, any obligation to conduct any remedial or monitoring work) under any Environmental Laws or otherwise resulting from or relating to the handling, storage, use, transportation or disposal of any Hazardous Substance by or on behalf of the Company or any Subsidiary or any of their respective predecessors or otherwise.
- (c) To the knowledge of the Seller, the Seller has previously provided to Buyer true and correct copies of all written reports or other documents arising out of environmental inspections, investigations, studies, audits, tests, reviews or other analyses conducted with respect to any Real Property listed on Schedule 4.5.1. or any Real Property leased by the Company or any of its Subsidiaries as warehouse space.
- 4.16. Employees and Labor Relations. None of the employees of the Company or any Subsidiary is represented by a labor union, and no petition has been filed or proceedings instituted by any employee or group of employees with any labor relations board seeking recognition of a bargaining representative. To the knowledge of Seller, there is no organizational effort currently being made or threatened by or on behalf of any labor union to organize any employees of the Company or any Subsidiary. Except as set forth on Schedule 4.16, there are no controversies or disputes pending between the Company or any Subsidiary on the one hand and any of their respective employees on the other hand, except for controversies and disputes with individual employees arising in the Ordinary Course of Business that have not had and will not have a Material Adverse Effect. Other than G. Politzer and certain internal auditors and tax, bad check processing or workmen's compensation related personnel, no individual who is an employee of Seller or any of its Affiliates devotes his or her business time primarily to the affairs of the Company or any Subsidiary.
- 4.17. Litigation, etc. There is no Action against the Company or any Subsidiary, pending or, to the knowledge of Seller, threatened, which could reasonably be expected to have a Material Adverse Effect, except for such of the foregoing as are described in Schedule 4.17. Except as set forth on Schedule 4.17, there is no Action pending or, to the knowledge of Seller, threatened with respect to which Seller or any of its Affiliates, on the one hand, and the Company or any of its Subsidiaries, on the other hand, are or would be parties. There is no Action pending or, to the knowledge of Seller, threatened, that seeks rescission of, seeks to enjoin the consummation of, or otherwise relates to, this Agreement or any of the transactions contemplated hereby and that could reasonably be expected to have a Material

Adverse Effect or a material adverse effect on Seller's ability to consummate the transactions contemplated hereby. No Governmental Order specifically directed at the Company or any of its Subsidiaries has been issued which has had or could reasonably be expected to have a Material Adverse Effect.

4.18. [Intentionally Omitted]

- 4.19. Brokers, etc. Except for Morgan Stanley & Co. Incorporated ("Morgan Stanley") and Financo, Inc., no broker, finder, investment bank or similar agent is entitled to any brokerage, finder's or other fee, Compensation or reimbursement of expenses in connection with the transactions contemplated by this Agreement based upon agreements or arrangements made by or on behalf of (or the conduct of) Seller, the Company, any Subsidiary or any of their respective Affiliates. Seller shall be solely responsible for the payment of the fees and expenses of Morgan Stanley and Financo, Inc.
- 5. REPRESENTATIONS AND WARRANTIES OF BUYER. In order to induce Seller to enter into and perform this Agreement and to consummate the transactions contemplated hereby, Buyer represents and warrants to Seller as follows:
- 5.1. Corporate Matters. Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority, corporate and otherwise, to enter into this Agreement, to carry out and perform its obligations hereunder and to consummate the transactions contemplated hereby.
- 5.2. Authorization and Enforceability. This Agreement has been duly authorized, executed and delivered by Buyer, and is Enforceable against Buyer.
- 5.3. Non-Contravention, etc. No approval, consent, waiver, authorization or other order of, and no filing, registration, qualification or recording with, any Governmental Authority or any other Person is required to be obtained or made by or on behalf of Buyer or any of its Subsidiaries in connection with the execution, delivery or performance of this Agreement and the consummation of the transactions contemplated hereby, except for (i) satisfaction of the requirements of the HSR Act and (ii) any item required to be obtained from or made with any Person other than a Governmental Authority the failure to obtain or make which, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect on Buyer nor a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby. Neither the execution, delivery and performance of this Agreement nor the consummation of any of the transactions contemplated hereby (including, without limitation, the execution, delivery and performance of the Closing Agreements and performance by Buyer of its obligations in respect of the Buyer Stock in accordance with its terms) does or will constitute, result in or give rise to (i) a breach or violation or default under any Legal Requirement applicable to Buyer or any of its Subsidiaries (assuming the accuracy of the representations and warranties of Seller in Section

4.4 of the Preferred Stock Subscription Agreement), (ii) a breach of or a default under any Charter or By-Laws provision of Buyer or any of its Subsidiaries, (iii) the acceleration of the time for performance of any obligation under any Contractual Obligation of Buyer or any of its Subsidiaries, (iv) the imposition of any Lien upon or the forfeiture of any asset of Buyer or any of its Subsidiaries, (v) a breach of or a default under any Contractual Obligation of Buyer or any of its Subsidiaries, or (vi) termination, right of termination, modification of terms or change in benefits or burdens under any Contractual Obligation of Buyer or any of its Subsidiaries, other than in the case of clauses (i), (iii), (iv), (v) and (vi) such as, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect on Buyer nor a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby (including, without limitation, Buyer's ability to consummate its obligations under the other Closing Agreements and its obligations in respect of the Buyer Stock in accordance with its terms).

5.4. Investment Intent.

- (a) Buyer is acquiring the Shares hereunder for its own account, for investment, and not with a view to, or for sale in connection with, any distribution thereof within the meaning of the Securities Act.
- (b) Buyer understands and agrees that the Shares will not be registered or qualified under the Securities Act of 1933, as amended (the "Securities Act"), or state "blue-sky" or other securities laws and therefore cannot be resold unless they are registered under the Securities Act and applicable state laws or unless an exemption from such registration requirement is available.
- (c) Buyer is able to bear the economic risk of holding the Shares for an indefinite period of time and is experienced and has such knowledge and experience in financial and business matters that it is capable of evaluating the risks and merits of acquiring the Shares. Buyer acknowledges that the Shares will bear a legend to the effect that transfers are restricted unless (i) the transfer is exempt from the registration requirements under the Securities Act and the Company receives an opinion of counsel reasonably satisfactory to the Company to that effect or (ii) the transfer is made pursuant to an effective registration statement under the Securities Act.
- (d) Buyer understands that the Company is under no obligation to effect a registration of the Shares under the Securities Act .
- (e) Buyer is an Accredited Investor within the definition set forth in Rule 501(a) of the Securities Act.
- (f) Nothing in this Section 5.4 shall limit or qualify the representations, warranties, covenants or agreements made by Seller herein or in any Closing $\,$

Agreement or in any certificate or document delivered pursuant hereto or thereto. Seller acknowledges and agrees that the purpose of this Section 5.4 is solely to ensure that the sale of the Shares pursuant hereto complies with the transfer restrictions under the Securities Act and the transfer restrictions under other applicable securities laws.

- 5.5. Litigation. There is no Action against Buyer or any of its Subsidiaries, pending or, to the knowledge of Buyer, threatened, which could reasonably be expected to have a Material Adverse Effect on Buyer. There is no Action pending or, to the knowledge of Buyer, threatened, that seeks rescission of, seeks to enjoin the consummation of, or otherwise relates to, this Agreement or any of the transactions contemplated hereby and that could reasonably be expected to have a Material Adverse Effect on Buyer or a material adverse effect on Buyer's ability to consummate the transactions contemplated hereby. No Governmental Order specifically directed at Buyer or any of its Subsidiaries has been issued which has had or could reasonably be expected to have a Material Adverse Effect on Buyer.
- 5.6. Financing. Buyer has received and delivered to Seller a letter from The First National Bank of Chicago, Bank of America Illinois, The Bank of New York and Pearl Street L.P. ("Lead Lenders") dated as of the date hereof (the "Commitment Letter"), with respect to debt financing in an amount sufficient to enable Buyer to pay the Cash Purchase Price. Buyer represents and warrants that the terms of such letter have not been altered or amended by Buyer or Lead Lenders in a manner that would have a material adverse effect upon Buyer's ability to perform its obligations under this Agreement and that such letter remains in full force and effect (unless superseded by definitive credit documentation that would not have a material adverse effect upon Buyer's ability to perform its obligations under this Agreement).
- 5.7. Brokers, etc. Except for Goldman Sachs & Co. ("Goldman") and Salomon Brothers Inc ("Salomon"), no broker, finder, investment bank or similar agent is entitled to any brokerage, finder's or other fee, Compensation or reimbursement of expenses in connection with the transactions contemplated by this Agreement based upon agreements or arrangements made by or on behalf of (or the conduct of) Buyer or its Affiliates. Buyer shall be solely responsible for the payment of the fees and expenses of Goldman and Salomon.

CERTAIN AGREEMENTS OF THE PARTIES.

6.1. No Solicitation of Other Offers. Seller will not, and will cause all of its Affiliates and all of its and their respective employees, representatives and agents not to, directly or indirectly, solicit or initiate or enter into discussions or transactions or Contractual Obligations with or encourage or provide any information to any Person (other than Buyer and its designees) concerning any sale of stock of, or any merger or share exchange or sale or other disposition of securities or substantial assets or any recapitalization or any similar transaction involving, the Company or any of its Subsidiaries. Seller will notify Buyer immediately upon becoming aware that any Person has made any proposal, offer, inquiry, or

contact with respect to any such transaction, which notice shall include the identity of all relevant parties and the content of such communication.

- 6.2. Access to Premises and Information. Prior to the Closing, Seller will cause the Company and its Subsidiaries to permit Buyer and its prospective lenders, and their respective representatives, to have full access to their premises and documents, books and records and to make copies during normal business hours of such financial and operating data and other information with respect to the Company as Buyer, such lenders, or any of their representatives shall reasonably request; provided, however, that Seller shall not be required to provide access to, or copies of, the portions of any documents, books and records or other data that contain information relating solely to entities other than the Company and its Subsidiaries. In addition, Seller shall cause the Company's and its Subsidiaries' management to be available to Buyer and its prospective lenders at such times, and from time to time, as Buyer and its prospective lenders may reasonably request in connection with the transactions contemplated hereby and their review of the Business. Seller will cause to be delivered such additional information and copies of documents, books and records relating to the Company, its Subsidiaries or the Business as may be reasonably requested by Buyer, such lenders, or any of their representatives, including, without limitation, all quarterly and monthly financial statements that become available prior to the Closing (which financial statements shall be included within the meaning of the term Financial Statements for all purposes of this Agreement from and after their respective dates).
- 6.3. Confidentiality Covenant of Buyer. The provisions of that certain letter agreement between Buyer and Seller dated June 26, 1995 (the "Confidentiality Agreement") are hereby confirmed and remain in effect; provided, however, that the Confidentiality Agreement is hereby amended so that (i) it shall terminate not later than the consummation of the Closing, (ii) it shall not prohibit any retention of records or disclosure made in connection with the enforcement of any right or remedy relating to this Agreement or the transactions contemplated hereby and (iii) in the event the Closing does not occur and this Agreement is terminated, the term of such Confidentiality Agreement shall be extended until the fifth anniversary of the date hereof.
- 6.4. Operation of Business in the Ordinary Course. On and prior to the Closing Date, except as otherwise required by this Agreement (including without limitation Section 6.14), Seller will cause the Company to conduct the Business only in the Ordinary Course of Business and use its reasonable best efforts to maintain the value of the Business as a going concern and the relationships of the Company with customers, suppliers, vendors, employees, agents and Governmental Authorities. Seller agrees to cause the Company and its Subsidiaries to make capital expenditures in the Ordinary Course of Business up to the Closing Date and prior to the Closing Date to consult with Buyer before making significant decisions regarding merchandise markdowns, inventory purchases and promotional activity. Without limiting the generality of the foregoing, on and prior to the Closing Date Seller will cause the Company not to, without the prior written consent of Buyer:

- (a) Enter into any transactions with Seller or any of its Affiliates except in the Ordinary Course of Business or with respect to Affiliate Debt or as set forth on Schedules 4.10 or 7.2(i);
- (b) Pay or accrue any Compensation other than in the Ordinary Course of Business or increase any Compensation of any officer or employee other than such increases in Compensation for individual employees as may be made in the Ordinary Course of Business;
- (c) Make any Distribution other than (i) distributions of cash to Seller in the Ordinary Course of Business pursuant to the Cash Management Program, Seller's Employee Stock Ownership Plan as currently in effect or Seller's VEBA as currently in effect; (ii) the distributions set forth on Schedule 4.3(b)(iv)(B); (iii) distributions of cash or of any receivable constituting Affiliate Debt in connection with the repayment or cancellation of Affiliate Debt; and (iv) distributions or contributions in connection with an increase in or the repayment or cancellation (in whole or in part) of Debt or intercompany advances between the Company and its wholly owned Subsidiaries or between wholly owned Subsidiaries:
- (d) Except as set forth on Schedule 6.4(d), incur any Debt except capitalized leases entered into in the Ordinary Course of Business and in amounts consistent with the Budget or Debt or intercompany advances between the Company and any wholly owned Subsidiary or between wholly owned Subsidiaries, or incur any Lien except in the Ordinary Course of Business;
 - (e) Amend the Charter or Bylaws of the Company or any Subsidiary;
- (f) Allow any material Permit or License to lapse or terminate or fail to renew any Permit or License in accordance with prudent business practice:
- (g) Fail to operate the Business and maintain the Company's and its Subsidiaries' books, accounts and records in the Ordinary Course of Business and maintain in good repair the Company's and its Subsidiaries' business premises, fixtures, machinery, furniture and equipment in a manner consistent with past practice;
- (h) Engage any new employee of the Company or its Subsidiaries for a salary in excess of \$100,000 per annum;
- (i) Enter into, amend in any material respect, extend, terminate or permit any renewal notice period or option to lapse with respect to any Lease or any other Contractual Obligation that contains either consideration to be given or performed by the Company or any of its Subsidiaries of a value exceeding \$250,000 or a term exceeding one year (except for purchases of inventory in the Ordinary Course of

Business and the making of capital expenditures in an aggregate amount consistent with the Budget and except to the extent otherwise required by this Agreement or to the extent the obligation is between the Company and a wholly owned Subsidiary or between wholly owned Subsidiaries);

- (j) Purchase or otherwise acquire any Real Property;
- (k) Take any of the actions specified in any of subsections (a) through (d) of Section 4.3; or
 - (1) Consent or agree to do any of the foregoing.
- 6.5. Certain Notices. Prior to the Closing, Seller will promptly upon becoming aware thereof give Buyer written notice of any material development affecting the Business, or the financial condition of the Company and any material breach of or inaccuracy in any representation or warranty of Seller contained in this Agreement; provided, however, that no such disclosure shall be deemed to amend any Schedule hereto, or prevent or cure any breach of or inaccuracy in, or disclose any exception to, any of the representations and warranties set forth herein.
- 6.6. Preparation for Closing. Each party will use its reasonable best efforts to bring about the timely fulfillment of each of the conditions precedent to the obligations of the other parties hereto set forth in this Agreement. Without limiting the generality of the foregoing, the parties shall take the actions set forth below in this Section 6.6.
 - 6.6.1. HSR Filing. Promptly upon execution and delivery of this Agreement, each of Seller and Buyer will prepare and file, or cause to be prepared and filed, with the appropriate Governmental Authorities, a notification with respect to the transactions contemplated by this Agreement pursuant to the HSR Act. Each of Seller and Buyer will promptly provide all additional information requested, and take all other actions necessary or appropriate, to comply with notification requirements under the HSR Act and to cause the expiration of all waiting periods under the HSR Act.
 - 6.6.2. Closing Agreements. Seller will enter into each of the Closing Agreements to which it is intended to be a party, and Seller will cause each of its Affiliates, the Company and any Subsidiary which is intended to be a party to any Closing Agreement to enter into each Closing Agreement to which such Person is intended to be a party. Buyer will enter into each of the Closing Agreements to which it is intended to be a party, and Buyer will cause each of its Affiliates which is intended to be party to any Closing Agreement to enter into each Closing Agreement to which such Affiliate is intended to be a party.

6.7. Tax Matters.

- (a) Section 338(h)(10) Election and Corporate Structure. Buyer and Seller agree to join in making a timely, effective and irrevocable election under Section 338(h)(10) of the Code (and any corresponding elections under state, local, or foreign tax law, other than Puerto Rican law) (collectively, the "Section 338(h)(10) Election") with respect to the Company and each of its Subsidiaries, and to file such election in accordance with applicable regulations. Seller and Buyer agree to cooperate (and cause their respective subsidiaries to cooperate) in all respects for the purpose of effectuating a timely and effective Section 338(h)(10) Election, including without limitation, the execution and filing of any forms or returns.
- (b) Buyer's Covenants. Buyer covenants that it will not cause or permit the Company, any Subsidiary or any Affiliate of Buyer to take any action on the Closing Date other than in the ordinary course of business, including but not limited to the distribution of any dividend or the effectuation of any redemption that could give rise to any tax liability to the Seller Affiliated Group or the reduction of any loss of the Seller or the Seller Affiliated Group. Except for the transactions contemplated by this Agreement, Buyer agrees that Seller is to have no liability for any Tax resulting from any action of the Company, Buyer or any Affiliate of Buyer on the Closing Date, which action is not within the ordinary course of business of such persons, and agrees to indemnify and hold harmless Seller and its Affiliates against any such tax. Seller agrees to give prompt notice to Buyer of the assertion of any claim, or the commencement of any action or proceeding, in respect of which indemnity may be sought under this Section 6.7(b). Buyer may participate in and assume the defense of any such suit, action or proceeding at its own expense. If Buyer assumes such defense, Seller shall have the right (but not the duty) to participate in the defense thereof and to employ counsel, at its own expense, separate from the counsel employed by Buyer. Whether or not Seller chooses to defend or prosecute any claim, the parties hereto shall cooperate in the defense or prosecution thereof.

(c) Tax Indemnification.

(i) Seller shall be liable for and shall pay (and shall indemnify and hold Buyer harmless from) (x) all Income Taxes with respect to the Company and its Subsidiaries for any Pre-Closing Tax Period (including without limitation any Income Taxes attributable to the Section 338(h)(10) Elections), (y) all Taxes other than Income Taxes and Transfer Taxes ("Miscellaneous Taxes") with respect to any Pre-Closing Tax Period, but only to the extent the amount payable exceeds the amount reflected on the Company Net Assets Statement for Miscellaneous Taxes and (z) any and all federal Income Taxes of the Affiliated Group of which Seller is a member imposed on the Company or any of its Subsidiaries pursuant to Section 1.1502-6 of the Treasury Regulations, in each

case incurred or suffered by Buyer, any of its Affiliates or, effective upon the Closing, the Company or any Subsidiary (the sum of (x), (y) and (z) being referred to as a "Tax Loss"); provided,

however, that Seller shall have no liability for the payment of any \hdots

Tax Loss attributable to or resulting from any action described in Section 6.7(b) hereof. For purposes of this Section 6.7, (A) the term "Pre-Closing Tax Period" shall mean all taxable periods ending on or before the close of the Closing Date and the portion ending at the close of the Closing Date of any taxable period that includes (but does not end on) the Closing Date, and (B) the term "Post-Closing Tax Period" shall mean all taxable periods that begin on or after the day following the Closing Date and the portion ending after the Closing Date of any taxable period that includes (but does not end on) the Closing Date. In the case of a taxable period that includes (but does not end on) the Closing Date, the Tax attributable to the Pre-Closing Tax Period shall be the responsibility of Seller (and not the Company and its Subsidiaries) and the Taxes attributable to the Post-Closing Tax Period shall be the responsibility of Buyer and the Company and its Subsidiaries.

- (ii) For purposes of this Section 6.7(c), in the case of any Taxes that are imposed on a periodic basis and are payable for a Tax period that includes (but does not end on) the Closing Date, the portion of such Tax related to the portion of such Tax period ending on the Closing Date shall (x) in the case of any Taxes other than Taxes based upon or related to income, sales, gross receipts, wages, capital expenditures or expenses, be deemed to be the amount of such Tax for the entire Tax period multiplied by a fraction the numerator of which is the number of days in the Tax period ending on the Closing Date and the denominator of which is the number of days in the entire Tax period, and (y) in the case of any Tax based upon or related to income, sales, gross receipts, wages, capital expenditures or expenses, be deemed equal to the amount which would be payable if the relevant Tax period ended on the Closing Date. In the case of Income Taxes described in the preceding sentence, if either Buyer or Seller is adversely affected (as a consequence of an increase in liability or a reduction of refund or other tax attribute) that would have been available to it if the relevant tax period had ended on the Closing Date, and the other party is benefitted from such circumstance, the party benefitted shall reimburse the party adversely affected to the extent of the benefit realized.
- (iii) If as a result of an adjustment Seller makes a payment to any taxing authority in respect of a Miscellaneous Tax of the Company with respect to any Pre-Closing Tax Period, then Buyer shall promptly pay to Seller an amount equal to such payment made by Seller, provided, however, that the aggregate of such payments by

Buyer shall not exceed the amount reflected on the Company Net Assets Statement for Miscellaneous Taxes.

- (iv) Any payment by Seller pursuant to this Section 6.7(c) shall be made (x) if reflected on a Tax Return, contemporaneously with the filing of such Return and (y) in all other cases, not later than 30 days after receipt by Seller of written notice from Buyer stating that any Tax Loss has been paid by Buyer, any of its Affiliates or, effective upon the Closing, the Company or any Subsidiary and the amount thereof and of the indemnity payment
- (v) If any claim or demand for Taxes in respect of which indemnity may be sought pursuant to this Section 6.7(c) is asserted in writing against Buyer, any of its Affiliates or, effective upon the Closing, the Company or any Subsidiary, Buyer shall promptly notify Seller of such claim or demand within sufficient time that would allow Seller to timely respond to such claim or demand, and shall give Seller such information with respect thereto as Seller may reasonably request. Seller may discharge, at any time, its indemnification obligation under this Section 6.7(c) by paying to Buyer the amount of the applicable Tax Loss, calculated on the date of such payment. Seller may, at its own expense, participate in and, upon notice to Buyer, assume the defense of any such claim, suit, action, litigation or proceeding (including any Tax audit). If Seller assumes such defense and if the relevant claim, suit, action, litigation or proceeding relates to a taxable period that includes (but does not end on) the Closing Date, Buyer shall have the right (but not the duty) to participate in the defense thereof and to employ counsel, at its own expense, separate from the counsel employed by Seller. Whether or not Seller chooses to defend or prosecute any claim, all of the parties hereto shall cooperate in the defense or prosecution thereof. Seller shall not be liable under this Section 6.7(c), for (x) any Tax claimed or demanded by any taxing authority, the payment of which was made without Seller's prior written consent unless Seller refused to participate in the proceedings and assume the defense or (y) any settlements effected without the consent of Seller, or resulting from any claim, suit, action, litigation or proceeding in which Seller was not permitted an opportunity to participate.
- (vi) Except with respect to Taxes described in clause (z) of Section 6.7(c)(i) or any tax detriment attributable to any agreement entered into prior to the Closing Date obligating the Company or any Subsidiary to make payments that are not deductible under Code Section 280G, for which Seller shall indemnify Buyer and which shall be treated as a Tax Loss, Buyer shall be liable for any Taxes pertaining to any Post-Closing Tax Period, and any transactions described in Section 6.7(b).
- (vii) No action of Buyer, the Company or any Subsidiary with respect to the ownership or corporate existence of the Company or any Subsidiary undertaken subsequent to the Closing Date shall in any way expand or limit the

amount or scope of any obligation of either Buyer or Seller under this Section 6.7(c).

- (d) Tax Sharing Agreements. All Tax sharing agreements or similar agreements with respect to or involving the Company and its Subsidiaries shall be terminated as of the Closing Date and, after the Closing Date, the Company and its Subsidiaries shall not be bound thereby or have any liability thereunder.
- (e) Transfer Taxes. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees (including any penalties and interest) incurred in connection with this Agreement ("Transfer Taxes") shall be borne by the party on whom such Tax is imposed under the relevant law, except that the aggregate liability for any New York State Real Property Gains Tax and New York City Real Estate Transfer Tax shall be shared equally by Buyer and Seller.
 - (f) Return Filings, Refunds and Credits.
 - (i) Seller shall include the income of the Company and its Subsidiaries in Seller's federal consolidated Income Tax Returns, and shall file all state, foreign and local Income Tax Returns of the Company and its Subsidiaries, for all Tax periods ending on or before the Closing Date and shall be responsible for remitting all Taxes reflected on such Income Tax Returns. Seller shall also prepare or cause to be prepared and file or cause to be filed all Miscellaneous Tax Returns due on or before the Closing Date (taking into account extensions) and shall be responsible for remitting all Miscellaneous Taxes reflected on such Miscellaneous Tax Returns. Copies of all such Tax Returns (or the relevant portion thereof relating to the Company and its Subsidiaries) shall be furnished to Buyer.
 - (ii) Buyer shall prepare or cause to be prepared and file or cause to be filed on a timely basis (with the assistance of Seller to the extent provided in any separate agreement for continuing services) all Income Tax Returns with respect to the Company and its Subsidiaries for taxable periods including (but not ending on) the Closing Date and all Miscellaneous Tax Returns due after the Closing Date (taking into account extensions) and shall be responsible for remitting all Taxes reflected on such Tax Returns. Any such Tax Return shall be prepared in a manner consistent with past practice and without a change of any election or any accounting method and shall be submitted by Buyer to Seller in sufficient time to permit a reasonable review prior to the due date (including extensions) of such Tax Return. Seller shall have the right to review all work papers and procedures used to prepare any such Tax Return. If Seller, within 10 business days after delivery of any such Tax Return, notifies Buyer in writing that it objects to any items in such Tax Return, the parties shall proceed

in good faith to resolve the disputed items and, if they are unable to do so within 10 business days, the disputed items shall be resolved (within a reasonable time, taking into account the deadline for filing such Tax Return) by the Accountants (as defined in Section 6.7(g), provided, however, that if the Accountants shall determine that two or more alternative positions with respect to the matter in question are equally supported by applicable law, the party that is liable for such item under Section 6.7(c) shall determine which position shall be taken on the relevant Tax Return. Upon resolution of all disputed items, the relevant Tax Return shall be adjusted to reflect such resolution and shall be binding upon the parties without further adjustment. The costs, fees and expense of such accounting firm shall be borne equally by Buyer and Seller.

- (iii) Seller and Buyer shall reasonably cooperate, and shall cause their respective Affiliates, agents, auditors, representatives, officers and employees reasonably to cooperate, in preparing and filing all Tax Returns (including amended returns and claims for refund), including maintaining and making available to each other all records necessary in connection with Taxes and in resolving all disputes and audits with respect to all taxable periods relating to Taxes. Buyer and Seller agree to retain or cause to be retained all books and records pertinent to the Company and the Subsidiaries until the applicable period for assessment under applicable law (giving effect to any and all extensions or waivers) has expired, and to abide by or cause the abidance with all record retention agreements entered into with any taxing authority. The Company and its Subsidiaries agree to give Seller reasonable notice prior to transferring, discarding or destroying any such books relating to Tax matters and, if Seller so requests, the Company or any Subsidiary shall allow Seller to take possession of such books and records. Buyer and Seller shall cooperate with each other in the conduct of any audit or other proceedings involving the Company or any Subsidiary for any Tax purposes and each shall execute and deliver such powers of attorney and other documents as are necessary to carry out the intent of this subsection. For any Income Tax Return of the Company or its Subsidiaries that Seller is responsible for filing and that requires the signature of an officer of the Company or one of its Subsidiaries, Seller shall present a completed Income Tax Return for the signature of an appropriate officer designated by the Company or the applicable Subsidiary. Seller shall give such officer any support for the Tax Return reasonably requested by such officer. The officer shall sign the return and deliver it to Seller as soon as reasonably practicable.
- (iv) Except in the case of refunds attributable to carrybacks from Post-Closing Tax Periods, any refunds with respect to Income Tax Returns paid to Buyer for any period ending on or before the Closing Date which were not reflected on the Company Net Assets Statement shall be paid to Seller by Buyer

within three (3) business days after receipt in cash or as a credit to Buyer's Tax liability.

(g) Allocation of Purchase Price. In connection with the Section 338(h)(10) Election, Buyer and Seller shall cooperate as provided herein in determining the modified ADSP (as such term is defined in Treasury Regulations Section 1.338(h)(10)-1) of the assets of the Company and the allocation of the modified ADSP for purposes of Section 338(a)(1) of the Code in accordance with all applicable Treasury Regulations promulgated under Section 338 of the Code. Buyer initially shall determine such modified ADSP and allocation of the modified ADSP and shall notify Seller in writing of the price and allocation so determined ("Buyer's Deemed Sales Price Notice") within 120 days after the Closing Date. Seller shall be deemed to have accepted such determination unless, within 30 days after receipt of Buyer's Deemed Sales Price Notice, Seller notifies Buyer in writing of (i) the amount that Seller proposes as the deemed sales price (if it differs from that proposed by Buyer), (ii) the allocation of the deemed sales prices proposed by Seller and (iii) the reasons for Seller's allocations. If Seller provides such notice to Buyer, the parties shall proceed in good faith to determine mutually the matters in dispute and, if they are unable to do so within 30 days, the matter shall be referred to a nationally recognized independent accounting firm chosen and mutually acceptable to both Buyer and Seller (the "Accountants") who shall within 30 days decide the matter. The decision of the Accountants shall be final and bind both parties. The Accountants' fees shall be shared equally by Buyer and Seller. Neither Buyer nor Seller shall take, nor shall they permit any affiliated corporation (including, without limitation, the Company and its Subsidiaries) to take, any position for Tax purposes that is inconsistent with the deemed sales price and allocation as finally determined hereunder; provided, however, that the deemed purchase price of the assets shall differ from the deemed sales price to the extent necessary to reflect the inclusion in the total deemed purchase price of items (for example, Buyer's capitalized acquisition costs in addition to the Purchase Price) not included in the total deemed sales price.

6.8. Expenses of Transaction.

6.8.1. Transaction Costs of Seller. Except to the extent specifically otherwise provided by Section 3.4(d), Seller shall pay all financial advisory, legal, accounting and other fees and expenses incurred by Seller, the Company or any of their Affiliates in connection with the transactions contemplated by this Agreement (including, without limitation, any sale bonus or retention or similar forms of Compensation payable by the Company or any of its Subsidiaries to any directors, officers or employees of any Person in connection with or as a result of the sale of the Shares or otherwise in preparation for or connection with the transactions contemplated hereby); provided that with respect to the Company and its Subsidiaries the Seller shall pay only fees and expenses incurred prior to Closing.

- 6.8.2. Transaction Costs of Buyer and Affiliates. Except to the extent specifically otherwise provided by Section 3.4(d), Buyer shall bear all financial advisory, legal, accounting and other fees and expenses incurred by Buyer in connection with the transactions contemplated by this Agreement
- 6.9. Confidentiality Covenants. Seller acknowledges that the success of the Company after the Closing and its ability to generate earnings sufficient to service the financing of the Purchase Price depends upon the continued preservation of the confidentiality of certain information possessed by Seller and its Affiliates, that the preservation of the confidentiality of such information and its Affiliates is an essential premise of the bargain between the parties, and that Buyer would be unwilling to enter into this Agreement in the absence of this Section 6.9. Accordingly, Seller hereby agree with Buyer as follows:
 - 6.9.1. Confidentiality Covenant. For a period beginning on the date hereof and ending five years after the Closing, Seller will not, and will cause its Affiliates (and the Company and each Subsidiary of the Company until the Closing Date) not to, directly or indirectly, without the prior written consent of Buyer, disclose any confidential or proprietary information involving or relating to Buyer, the Company or their respective Subsidiaries or, insofar as such information relates exclusively to Buyer and its Subsidiaries (other than the Company and its Subsidiaries), use such information in any manner adverse to Buyer and its Subsidiaries; provided, however, that the information subject to the foregoing provisions of this sentence shall be deemed not to include any information known generally in the industry or otherwise in the public domain (other than as a result of disclosure in violation hereof by Seller or any such Affiliate thereof); and provided, further, that the provisions of this Section 6.9.1 shall not prohibit any retention of records or disclosure required by law or made in connection with the enforcement of any right or remedy relating to this Agreement or the transactions contemplated hereby.
 - 6.9.2. Enforcement. Seller acknowledges and agrees that (i) it regards the restrictions contained in this Section 6.9 as reasonable and designed to provide Buyer with limited, legitimate and reasonable protection against subsequent diminution of the value of the Shares attributable to any actions of Seller or any of its Affiliates contrary to such covenants and (ii) because the legal remedies of Buyer may be inadequate in the event of a breach of, or other failure to perform, any of the covenants and obligations set forth in this Section 6.9, Buyer may, in addition to obtaining any other remedy or relief available to it (including, without limitation, damages at law), obtain specific enforcement of this Section 6.9 and other equitable remedies. Seller also acknowledges and agrees that no breach by Buyer of, or other failure by Buyer to perform, any of the covenants or obligations of Buyer under this Agreement or otherwise shall relieve Seller of any of its obligations under this Section 6.9.

6.10. Books and Records; Personnel. (a) Seller acknowledges and agrees that from and after the Closing the Company will be entitled to own and possess, subject to the next succeeding sentence, all documents, books, records, agreements and financial data of any sort relating to the Company, its Subsidiaries or the Business. Seller agrees to deliver and cause its Affiliates to deliver, prior to the Closing, all such books and records in their possession to the Company or, to the extent such books and records are not readily separable from the books and records of Seller or any of its Affiliates relating to their businesses other than the Business, true and complete copies of such books and records.

(b) From and after the Closing Date:

- (i) Buyer shall not, and shall cause the Company not to, dispose of or destroy any of the books and records of the Company or its Subsidiaries relating to periods prior to the Closing ("Books and Records") in a manner or at a time inconsistent with Seller's ordinary policies and procedures with respect to document retention as in effect prior to the Closing without first offering to turn over possession thereof to Seller by written notice to Seller at least 30 days prior to the proposed date of such disposition or destruction.
- (ii) Buyer shall, and shall cause the Company to, allow Seller and its agents reasonable access to all Books and Records during normal working hours at Buyer's principal place of business or at any location where the Books and Records are stored, and Seller shall have the right, at its own expense, to make copies of any Books and Records; provided, however, that any such access or copying shall be had or done (A) in such a manner so as not to interfere with the normal conduct of Buyer's or the Company's business and (B) for a legitimate business purpose (such as tax preparation) that does not involve direct or indirect competition with the Business.
- (iii) Buyer shall, and shall cause the Company to, make available to Seller upon written request (A) copies of any Books and Records, (B) Buyer's and the Company's personnel to assist Seller in locating and obtaining any Books and Records, and (C) any of Buyer's and the Company's personnel whose assistance or participation is reasonably required by Seller or any of its Affiliates in anticipation of, or preparation for, existing or future Actions, Tax returns or other matters in which Seller or any of its Affiliates is involved.
- (iv) Seller shall reimburse Buyer, the Company or its Subsidiaries for the reasonable out-of-pocket expenses incurred by any of them in performing the covenants contained in this Section 6.10 and shall protect and accord confidential treatment to the information disclosed pursuant to this Section 6.10 as provided in Section 6.9.1.
- 6.11. Use of Certain Names and Marks. Seller acknowledges and confirms that: (i) from and after the Closing, neither Seller nor any of its Affiliates has or shall have any

rights in the Company Marks, and (ii) neither Seller nor any of its Affiliates will contest the ownership or validity of any rights of Buyer or the Company in or to any of the Company Marks, or registrations (or applications for registration) thereof. Promptly following the Closing, Seller will deliver to the Company or, at the option of Buyer destroy, all letterhead, invoices and other documents bearing any of the Company Marks and related symbols. Neither Seller nor any of its Affiliates shall have any right, after the Closing, to use or exploit any of the Company Marks, or any name confusingly similar thereto.

- 6.12. Further Assurances. Each party, upon the request from time to time of any other party hereto after the Closing, and without further consideration, will do each and every act and thing as may be necessary or reasonably requested to consummate the transactions contemplated hereby in an orderly fashion.
- 6.13. Reimbursement by the Parties. To the extent that Seller, on the one hand, or the Company or Buyer, on the other hand, receives any payment after the Closing which belongs to the other party, it shall promptly pay over such payment to the other party.
- 6.14. Open and Closed Stores. Seller shall cause the Company and its Subsidiaries to operate as of the Closing Date all of the stores listed under the caption "Open Stores" in the document entitled "Open Stores, Store Commitments and Store Closings" initialed by the parties (the "Section 6.14 Letter"). Until the Closing, Seller shall cause the Company and its Subsidiaries not to make any commitment or expenditure with respect to any store location not open on the date hereof without the written consent of Buyer (which consent may be withheld in Buyer's sole discretion), except for store locations specified in the Section 6.14 Letter under the caption "Permitted Future Store Locations". Seller shall cause the Company and its Subsidiaries to close the stores listed under the caption "Stores to be Closed" in the Section 6.14 Letter on or prior to the applicable date specified therein if such date occurs prior to the Closing Date (and if not so closed by such applicable date prior to the Closing Date, such stores shall be deemed closed prior to such date for all purposes of this Agreement, including, without limitation, the definition of Excluded Liabilities).
- 6.15. Financial Statement Deliveries. As soon as reasonably practicable following the date hereof and in any event not later than 50 days after the date hereof (or in the case of clause (c) below, such later date as shall be specified by Buyer), Seller shall cause to be delivered to Buyer, at the cost and expense of Seller, each of the following:
 - (a) The consolidated balance sheet of the Company and its Subsidiaries as of December 31, 1994 and the related statements of earnings, stockholders' equity and cash flows for the fiscal year then ended, accompanied by the notes thereto and the unqualified report thereon of KPMG Peat Marwick and similar financial statements for each of the two preceding fiscal years accompanied by the notes thereto and the unqualified report thereon of KPMG Peat Marwick.

- (b) The unaudited consolidated balance sheets of the Company and its Subsidiaries as of March 31, 1995, July 1, 1995 and October 1, 1995 and related unaudited consolidated statements of earnings and stockholders equity and cash flows for the Company and its Subsidiaries for the threemonth and year-to-date periods then ended, together with comparable information for the same periods in 1994 and a review report thereon of KPMG Peat Marwick (the expense of which report up to \$45,000 shall be borne by Buyer).
- (c) Such additional financial statements of the Company and its Subsidiaries for such periods and dates as Buyer shall request for the purpose of permitting Buyer to make required filings pursuant to the Securities Exchange Act of 1934, as amended (the "Exchange Act").

Seller shall cause all financial statements (including the notes thereto) referred to in this Section 6.15 to be prepared in accordance with generally accepted accounting principles consistently applied throughout the periods specified therein (other than as a result of the Accounting Policy Changes), and to present fairly in all material respects, the consolidated financial position and results of operations of the Company and its Subsidiaries for the periods specified therein, subject in the case of financial statements for interim periods to an absence of footnotes and to normal year-end audit adjustments which will not in the aggregate be material. In the case of any financial statements referred to in this Section 6.15 that reflect the Accounting Policy Changes, Seller shall deliver to Buyer with such financial statements, a written reconciliation showing the effects of the Accounting Policy Changes on such financial statements. Seller shall cause KPMG Peat Marwick to consent in writing (and to deliver such consent to Buyer) to the inclusion of its report on the financial statements referred to in clause (a) of this Section 6.15 in any filings made by Buyer under the Securities Act or the Exchange Act.

- 6.16. Insurance Policies. Seller will cause insurance coverage to remain in effect following the Closing for claims made under the Insurance Policies following the Closing in respect of events occurring prior to the Closing ("Post-Closing Claims"), and will not take or fail to take any action that would impair the ability of the Company and its Subsidiaries to make claims thereunder or to obtain the benefits afforded them thereby in accordance with the terms of the Insurance Policies as in effect on the date hereof; provided, however, that following the Closing, the Company and any Subsidiary shall not be entitled to make claims under the Insurance Policies in respect of Post-Closing Claims to the extent that the Company or the applicable Subsidiary has been fully compensated for its Losses in respect of such Post-Closing Claims by reason of the operation of Section 10 hereof.
- 6.17. Divestiture. If at any time prior to or subsequent to the Closing, Buyer shall cease to operate or shall vacate or transfer to any Person, or shall agree to cease to operate or to vacate or transfer to any Person, any store designated as a "Seller Store" or "Buyer Store" listed on the document captioned "Divestiture and Failed Consent Store Schedule" initialed by

the parties (the "Store Schedule") (any store so listed on such schedule being referred to in this Section 6.17 and in Section 6.18 as an "Affected Store") as the result of any agreement with or order of any Governmental Authority (including, without limitation, an order consented to by Buyer) related to any written inquiry, investigation or complaint by any Governmental Authority received by Buyer prior to or within one year after the Closing, the parties agree as follows:

 $6.17.1.\,$ Open Stores. If the Affected Store is listed on the Store Schedule under the caption "Open Store Schedule," Seller shall, upon written notice thereof furnished to Seller by Buyer at any time within 30 days following the cessation of operations, vacation or transfer of such Affected Store pursuant to such agreement or order, promptly pay to Buyer an amount in cash equal to 3 multiplied by, in the case of a Seller Store, the store contribution for such store as listed on the Open Store Schedule or, in the case of a Buyer Store, the store contribution for such store for the fiscal year of Buyer ended January 28, 1995 as certified to Seller by Buyer (and calculated in accordance with Buyer's customary methods of calculation thereof) (any such store contribution of either a Seller Store or a Buyer Store being referred to herein as "Store Contribution"), but in no event shall the amount of any such payment be less than \$250,000; provided that in respect of any Seller Store that was in operation for less than all of the Company's fiscal year ended December 31, 1994 or opened thereafter (a "New Seller Store") or any Buyer Store that was in operation for less than all of Buyer's fiscal year ended January 28, 1995 or opened thereafter (a "New Buyer Store"), the Store Contribution shall be equal to the Assumed Store Contribution. The term "Assumed Store Contribution" shall mean 66% of the weighted average Store Contribution (such weighting to be based on sales volumes) for the applicable fiscal year for Similar Sales Volume Stores of Seller in the case of a New Seller Store or of Buyer in the case of a New Buyer Store. The term "Similar Sales Volume Stores" shall mean (i) in the case of a New Seller Store, Seller Stores (other than New Seller Stores) that produced sales volume for the fiscal year of Seller ended December 31, 1994 equal to at least 90% and not more than 110% of the sales volume of the applicable New Seller Store for the most recent full 12 calendar months for which such New Seller Store has been in operation prior to the Closing Date (or if such New Seller Store has been in operation for less than 12 full calendar months prior to the Closing Date, then the annualized sales volume based upon the number of full calendar months for which such New Seller Store has been in operation prior to the Closing Date) and (ii) in the case of a New Buyer Store, Buyer Stores (other than New Buyer Stores) that produced sales volume for the fiscal year of Buyer ended January 28, 1995 equal to at least 90% and not more than 110% of sales volume of the applicable New Buyer Store for the most recent full 12 calendar months for which such New Buyer Store has been in operation prior to the Closing Date (or if such New Buyer Store has been in operation for less than 12 full calendar months prior to the Closing Date, then the annualized sales volume based upon the number of full calendar months for which such New Buyer Store has been in operation prior to the Closing

6.17.2. Closed Stores. If the Affected Store is listed on the Store Schedule under the caption "Closed Stores Schedule," Seller shall, subject to the provisions of this Section 6.17.2 regarding dispute resolution, upon written notice thereof furnished to Seller by Buyer at any time or from time to time within two years following any such cessation of operations, vacation or transfer of such Affected Store pursuant to such agreement or order, promptly pay to Buyer an amount (in aggregate) that constitutes 50% of Buyer's Closed Store Damages with respect to such cessation, vacation or transfer, but in no event more than \$250,000. "Buyer's Closed Store Damages" shall mean the present value (applying a discount rate of 9%) of all actual Losses that are incurred by Buyer as a result of such cessation, vacation or transfer, after netting out from such amount the present value (applying a discount rate of 9%) of all actual Losses that Buyer reasonably would have been expected to incur if such store had been voluntarily closed by Buyer in the ordinary course of business. If the parties fail to agree upon the amount payable from Seller to Buyer in the event of an occurrence subject to this Section 6.17.2, the parties shall first use commercially reasonable efforts to resolve such disagreement among themselves. If such disagreement shall not have been resolved within 30 days following delivery of a notice by Buyer to Seller referred to in the first sentence of this Section 6.17.2, the dispute shall be submitted to Alternative Accountants for resolution within 30 calendar days after submission. The determination of the Alternative Accountants as to the resolution of any dispute shall be binding and conclusive upon all parties hereto. All determinations pursuant to this Section 6.17 shall be in writing and shall be delivered to the parties hereto. Any award made pursuant to this Section 6.17 may be entered in and enforced by any court referred to in Section 11.1 hereof and the parties hereby consent and submit themselves to the jurisdiction of any such court for purposes of the enforcement of any such award.

Buyer agrees to keep Seller generally apprised of the status of any written inquiry, investigation or complaint by any Governmental Authority that may result in an event to which this Section 6.17 would apply. Buyer and Seller agree that the provisions of this Section 6.17 constitute a material inducement upon which each is relying and will rely in entering into this Agreement and any other agreements relating hereto or contemplated hereby, and that the Purchase Price has been determined in light of the provisions of this Section 6.17. Buyer and Seller acknowledge and agree that the amount of Losses that will be incurred by Buyer will be difficult or impossible to determine if an event occurs which is the subject matter of this Section 6.17 and that the payments required under this Section 6.17 are intended to compensate Buyer for such Losses and to be the sole remedy for such Losses. Buyer shall have no duty to take any action in respect of any such event for the purpose of mitigating such Losses.

6.18. Failed Consents.

If prior to the first anniversary of the Closing Date, neither Seller nor Buyer has received any written objection from a lessor or sublessor (a "Landlord") under a lease or other Contractual Obligation for any Seller Store indicating that a consent or other approval (a "Consent") of such Landlord is required under such lease or other Contractual Obligation in connection with the transactions contemplated hereby, a Consent shall be deemed to have been delivered with respect to such lease or other Contractual Obligation (a "Delivered Lease"). If prior thereto either Seller or Buyer has received any such written objection from a Landlord, the related lease or other Contractual Obligation shall not be considered a Delivered Lease until (a) two (2) years have elapsed from the Closing Date without the commencement of legal proceedings by such Landlord against Seller, Buyer, the Company or any of its Subsidiaries or (b) such Landlord irrevocably withdraws its objections in writing or legal proceedings having been commenced within two years after the Closing Date are discontinued with prejudice or otherwise finally resolved against such Landlord without further right of appeal on the part of such Landlord. If legal proceedings are commenced against Seller, Buyer, the Company or any Subsidiary regarding any such Consent more than two (2) years after the Closing Date, Buyer shall be responsible for defending such action at Buyer's sole cost and expense.

If at any time prior to or subsequent to the Closing, Buyer or any of its Subsidiaries shall be required by court order, judgment or other proceeding, or any settlement of any such proceeding, to cease to operate or to vacate any Affected Store (other than a Store covered by a Delivered Lease) designated as a "Seller Store" on the Store Schedule (an "Eviction"), or shall reasonably agree to an increase in the rent (including, without limitation, rent retroactive to the Closing) (a "Rent Increase") for any Affected Store (other than a Store covered by a Delivered Lease) designated as a Seller Store on the Store Schedule, in either case as the result of any alleged breach or default under, or lessor's or sublessor's right of termination under, the lease or other Contractual Obligation applicable to such store primarily as a result of the transactions contemplated hereby, the parties agree as follows:

6.18.1. Evictions. In the case of an Eviction:

- (i) If such Affected Store is listed on the Store Schedule under the caption "Open Store Schedule," Seller shall, upon written notice thereof furnished to Seller by Buyer at any time within 30 days following the cessation of operations or vacation of such Affected Store, promptly pay to Buyer an amount in cash equal to 3 multiplied by the Store Contribution for such store but in no event less than \$250,000.
- (ii) If such Affected Store is listed on the Store Schedule under the caption "Closed Stores Schedule," Seller shall, subject to the provisions of this Section 6.18.1(ii) regarding dispute resolution, upon written notice thereof furnished to Seller by Buyer at any time or from time to time within two years

following any such cessation of operations or vacation of such Affected Store, promptly pay to Buyer an amount that constitutes 50% of Buyer's Closed Store Damages with respect to such cessation or vacation, but in no event more than \$250,000. If the parties fail to agree upon the amount payable from Seller to Buyer in the event of an occurrence subject to this clause (ii), such dispute shall be resolved in the manner and on the terms set forth for the resolution of disputes in Section 6.17.2 (with references therein to Section 6.17.2 being understood to apply to this clause (ii)).

6.18.2. Rent Increases. In the event of a Rent Increase:

- (i) If such Affected Store is listed on the Store Schedule under the caption "Open Store Schedule" Seller shall, upon written $\,$ notice thereof furnished to Seller by Buyer at any time within 30 days following the date Buyer, the Company, any Subsidiary or any of their respective Affiliates enters into an agreement providing for a Rent Increase with respect to such Affected Store, promptly pay to Buyer in cash an amount equal to 50% of the present value (applying a discount rate of 9%) of the increase in rent for the remainder of the current lease term and, only if the rent increase is applicable thereto, any available renewals thereof under the applicable lease. A rent increase shall not be deemed a Rent Increase unless the existing minimum or base rent (the "Original Minimum Rent") for the applicable lease is increased from such Original Minimum Rent to a new minimum rent (the "New Minimum Rent") more than the sum of the Original Minimum Rent and the percentage rent (if any) paid on such lease for the most recent twelve-month period ending prior to the Closing Date governing percentage rent and the breakpoint used to calculate the percentage rent is adjusted to reflect the New Minimum Rent. In no event shall the amount required to be paid under this Section 6.18.2(i) exceed the amount which Seller would be required to pay under Section 6.18.1(i) if the applicable store were subject to an Eviction.
- (ii) If such Affected Store is listed on the Store Schedule under the caption "Closed Stores Schedule," Seller shall have no liability for any Rent Increase.

Buyer agrees to keep Seller generally apprised of the status of any matter that may result in an event to which this Section 6.18 would apply. Buyer and Seller agree that the provisions of this Section 6.18 constitute a material inducement upon which each is relying and will rely in entering into this Agreement and any other agreements relating hereto or contemplated hereby, and that the Purchase Price has been determined in light of the provisions of this Section 6.18. Buyer and Seller acknowledge and agree that the amount of Losses that will be incurred by Buyer will be difficult or impossible to determine if an event occurs which is the subject matter of this Section 6.18 and that the payments required under

this Section 6.18 are intended to compensate Buyer for such Losses and to be the sole remedy for such Losses. Buyer shall have no duty to take any action in respect of any such event for the purpose of mitigating such Losses

- 6.19. No Solicitation or Employment. Except as provided by law, for a period beginning on the date hereof and ending two years after the Closing Date, neither the Seller nor any of its Affiliates shall solicit to employ or employ any individual who is an employee of the Company or any of its Subsidiaries on the date hereof, or at any time following the date hereof, and who on or prior to the Closing Date occupies a home office position (other than a secretarial or clerical position) or a management position, unless at least six months shall have elapsed following the Closing and following the cessation of such individual's employment with Buyer or any of its Affiliates. The restrictions of this Section 6.19 shall not apply to any individual listed on the document entitled "Listed Employees" initialed by the parties to the extent specified therein. Any individual specified therein shall be treated for purposes of Section 9 of this Agreement and all other purposes as between the parties in the manner specified in such document.
- 6.20. Manchester, CT and Staten Island, NY Stores; Etc. Prior to the Closing, Seller or any Affiliate of Seller and the Company shall enter into leases for the stores of the Company open or to be opened in Manchester, Connecticut and Staten Island, New York on the terms set forth in Schedule 6.20.
- 7. CONDITIONS TO THE OBLIGATION TO CLOSE OF BUYER. The obligations of Buyer at the Closing to purchase the Shares and to execute and deliver the Closing Agreements to which it is party are subject to the satisfaction, at or prior to the Closing, of all of the following conditions, compliance with which, or the occurrence of which, may be waived prior to the Closing in writing by Buyer in its sole discretion:
 - 7.1. Representations, Warranties and Covenants.
 - 7.1.1. Continued Accuracy of Representations and Warranties. All representations and warranties of Seller contained in this Agreement or any Closing Agreement that include qualifications as to materiality or Material Adverse Effect shall be true and correct as of the Closing and all other representations and warranties of Seller contained in this Agreement or any Closing Agreement shall be true and correct in all material respects as of the Closing, in each case with the same force and effect as if such representations and warranties were made at and as of the Closing.
 - 7.1.2. Performance of Agreements. Seller shall have performed and satisfied in all material respects all covenants and agreements required by this Agreement or any Closing Agreement to be performed or satisfied by it at or prior to the Closing.

- 7.1.3. Closing Certificate. At the Closing, Seller shall furnish to Buyer an unqualified certificate, signed by the President or Chief Financial Officer of Seller, dated the Closing Date, to the effect that the conditions specified in Sections 7.1.1 and 7.1.2 hereof have been satisfied.
- 7.2. Closing Agreements. At or prior to the Closing, the parties thereto other than Buyer shall have entered into each of the following Agreements (the "Closing Agreements"), in substantially the form thereof attached hereto without change other than such changes as may be reasonably satisfactory to Buyer:
 - (i) a Transitional Services Agreement having substantially the terms set forth in Schedule 7.2(i) (the "Transitional Services Agreement");
 - (ii) the Preferred Stock Subscription Agreement;
 - (iii) the Standstill and Registration Rights Agreement.
- 7.3. Legality; Governmental Authorization; Litigation. Buyer's purchase of and payment for the Shares, and the consummation of the other transactions contemplated hereby, shall not be prohibited by any Legal Requirement (other than any Permit). All necessary filings, if any, pursuant to the HSR Act shall have been made and all applicable waiting periods thereunder shall have expired or been terminated. No Action shall have been instituted at or prior to the Closing by any Governmental Authority that seeks to delay, enjoin or otherwise make illegal the consummation of the transactions contemplated hereby; provided that if such Action shall have been instituted by a non-federal Governmental Authority, there must be a reasonable likelihood that the result of such Action could be to delay, enjoin or otherwise make illegal Buyer's purchase of the Shares or the consummation of any other transaction contemplated hereby.
- 7.4. Affiliate Debt. There shall not be outstanding any Affiliate Debt. In addition, there shall not be outstanding any Debt or other advances owed to the Company or any Subsidiary by Seller or any of its Affiliates or by any present or former employee, stockholder or director of Seller.
- 7.5. Financing. Buyer shall have obtained the funds to be provided pursuant to the Commitment Letter, unless Buyer shall have failed to have obtained such funds exclusively as a result of Buyer's failure to pay any fees or expenses required to be paid by Buyer under the Commitment Letter.
- 7.6. Opinion of Counsel. Seller shall have furnished Buyer with favorable opinions of Davis Polk & Wardwell and Arthur V. Richards, general counsel of Seller, each dated the Closing Date, in substantially the forms of Exhibits C-1 and C-2 hereto.

- 7.7. General. Seller shall have furnished Buyer with such officers' certificates, good standing certificates, incumbency certificates and other customary closing documents as it may reasonably request in connection with the transactions contemplated hereby, including, without limitation (i) an instrument of assumption by Seller of the Excluded Liabilities, (ii) either a "sworn affidavit" or a "qualifying statement" that complies with Section 1445 of the Code and (iii) such director and officer resignation letters as Buyer may reasonably have requested of any of the officers or directors of the Company or any Subsidiary prior to Closing.
- 8. CONDITIONS TO THE OBLIGATION TO CLOSE OF SELLER. The obligations of Seller at the Closing to sell and transfer the Shares and to execute and deliver the Closing Agreements to which it is party are subject to the satisfaction, at or prior to the Closing, of all of the following conditions, compliance with which, or the occurrence of which, may be waived prior to the Closing in writing by Seller in its sole discretion:
 - 8.1. Representations, Warranties and Covenants.
 - 8.1.1. Continued Accuracy of Representations and Warranties. All representations and warranties of Buyer contained in this Agreement or any Closing Agreement that include qualifications as to materiality shall be true and correct as of the Closing and all other representations and warranties of Buyer contained in this Agreement or any Closing Agreement shall be true and correct in all material respects as of the Closing, in each case with the same force and effect as if such representations and warranties were made at and as of the Closing.
 - 8.1.2. Performance of Agreements. Buyer shall have performed and satisfied in all material respects all covenants and agreements required by this Agreement or any Closing Agreement to be performed or satisfied by Buyer at or prior to the Closing.
 - 8.1.3. Officer's Certificate. At the Closing, Buyer shall furnish to Seller an unqualified certificate signed by the President or Senior Vice President Finance and Chief Financial Officer of Buyer dated the Closing Date, to the effect that the conditions specified in Sections 8.1.1 and 8.1.2 hereof have been satisfied.
- 8.2. Closing Agreements; Buyer Stock. At or prior to the Closing, Buyer shall have entered into each of the Closing Agreements to which it is party, such agreements being in substantially the form attached hereto without change other than such changes as may be reasonably satisfactory to Seller, and Buyer shall have issued to Seller the Buyer Stock.
- 8.3. Legality; Government Authorization; Litigation. Seller's sale of the Shares, and the consummation of the other transactions contemplated hereby, shall not be prohibited by any Legal Requirement (other than any Permit). All necessary filings, if any, pursuant to the HSR Act shall have been made and all applicable waiting periods thereunder shall have

expired or been terminated. No Action shall have been instituted at or prior to the Closing by any Governmental Authority that seeks to delay, enjoin or otherwise make illegal the consummation of the transactions contemplated hereby; provided that if such Action shall have been instituted by a non-federal Governmental Authority there must be a reasonable likelihood that the result of such Action could be to delay, enjoin or otherwise make illegal Seller's sale of the Shares or the consummation of any other transaction contemplated hereby.

- 8.4. Opinion of Counsel. Buyer shall have furnished Seller with favorable opinions of Ropes & Gray and Jay Meltzer, Senior Vice President, General Counsel and Secretary of Buyer, each dated the Closing Date, in substantially the forms of Exhibits D-1 and D-2 hereto.
- 8.5. General. Seller shall have received copies of such officers' certificates, good standing certificate, incumbency certificates and other customary closing documents as it may reasonably request in connection with the transactions contemplated hereby.
 - EMPLOYMENT AND EMPLOYEE BENEFITS ARRANGEMENTS.
 - 9.1. Definitions. As used in this Agreement:
 - (a) the term "Employee" means an individual who, as of the Closing Date, is employed (including persons absent from active service by reason of layoff, illness, disability, or leave of absence, whether paid or unpaid) by the Company or any of its Subsidiaries.
 - (b) the term "Former Employee" means an individual who at any time prior to the Closing Date was employed by the Company or any of its Subsidiaries, but excluding any such person who is an Employee.
 - (c) the term "Company Plan" means each "employee benefit plan" as defined at Section 3(3) of ERISA which (i) is maintained, sponsored, contributed to, or participated in by Seller, the Company, any Subsidiary, any other Person in the same Controlled Group as any of the foregoing, or any of their respective predecessors, and (ii) covers or benefits one or more Employees or their spouses or dependents or with respect to which the Company or any of its Subsidiaries has or may have a material Liability.
 - (d) the term "Benefit Arrangement" means each plan, arrangement, contract, policy or practice (any of the foregoing, an "arrangement") of Seller, the Company, any Subsidiary, any other Person in the same Controlled Group as any of the foregoing, or any of their respective predecessors, which arrangement (i) provides any of the following benefits, coverages or insurance: pension, profit-sharing, savings, bonus, stock bonus, supplemental pension, deferred compensation (including so-called "excess" or "top-hat" deferred compensation), health (including dental, vision,

prescription drug and hospitalization), life insurance, short-term disability, long-term disability, severance, salary continuation, holiday, vacation, sick leave, scholarship, tuition assistance, dependent care spending, employee assistance, relocation, company car or automobile allowance, stock options, stock purchase, restricted stock, stock appreciation rights, phantom stock, or any other retirement, welfare, fringe benefit, or other employment- or service-related benefit (including any arrangement that facilitates the provision of such benefits, such as a "cafeteria plan" or spending account under Section 125 of the Code); (ii) covers or benefits one or more Employees or their spouses or dependents or with respect to which the Company or any Subsidiary has or may have a material Liability; and (iii) is not a Company Plan.

- (e) the term "Seller's Employee Stock Ownership Plan" means the Melville Corporation and Subsidiaries Employee Stock Ownership Plan, as amended from time to time.
- (f) the term "Seller's 401(k) Profit Sharing Plan" means the 401k Profit Sharing Plan of Melville Corporation and Affiliated Companies.
- 9.2. Disposition of Company Plans and Benefit Arrangements. Except as otherwise provided in this Agreement, Seller shall take such steps as are necessary to ensure that, as of the Closing, the Company and its Subsidiaries shall cease to participate in all Company Plans and Benefit Arrangements, other than those Company Plans and Benefit Arrangements, if any, sponsored and maintained solely by the Company or its Subsidiaries. At Buyer's request, Seller shall continue to cover Employees (and eligible spouses or dependents) under the Company Plans which provide medical, life insurance and disability benefits, from the Closing Date until such day after the Closing Date not later than April 30, 1996 as Buyer shall specify (the "Benefit Transition Period"). Buyer shall pay the cost (including any employee contributions received by Buyer) of such coverage on the basis of an equitable allocation determined in accordance with the Transitional Services Agreement. Seller acknowledges Buyer's interest in integrating Employees into benefit programs currently maintained or to be established by Buyer, and to that end agrees promptly to transfer to Buyer such employee- and benefits-related records and other information relating to Employees and Former Employees as Buyer may reasonably request, and otherwise to cooperate with Buyer. If so requested by Buyer, Seller shall make reasonable endeavors, including correspondence and cooperation with insurers or service providers to facilitate the creation or continuation by Buyer of transition plans or arrangements for Employees similar to those Company Plans and Benefit Arrangements which provide welfare benefits or insurance Except as provided in Section 9.3, Section 9.4 or Section 9.5, after coverage. the Benefit Transition Period, Buyer shall make available to Employees medical, life-insurance and disability benefits on terms substantially comparable to those generally available to other employees of Buyer and its subsidiaries or to those available under the applicable Company Plan, without limitation for preexisting conditions other than any such condition or limitation (including, without limitation, preexisting condition exclusions, waiting periods, activelyat-work requirements, and other

similar exclusions and conditions) as to which the relevant Company Plan provided only a conditional waiver and as to which the Employee (or in the case of his or her spouse or dependents, such spouse or dependents) had not, as of the Closing, satisfied the relevant conditions for such waiver. Buyer may provide, at Buyer's expense, or cause the Company and its Subsidiaries to continue or provide, at the Company's or its Subsidiaries' expense, such other benefits, if any, as Buyer in its sole discretion shall determine.

- Employment. Buyer may identify by separate letter to Seller individuals who will not continue in the employ of the Company or its Subsidiaries after the Closing, but any such individual, if employed immediately prior to the Closing, shall be treated for purposes of this Agreement as having been terminated following the Closing. Individuals who are Employees but who for any reason are not in active service at the Closing ("Non-Active Employees"), other than individuals described in the preceding sentence, shall be entitled to resume work for the Company or its Subsidiaries if and when they are able and willing to return to active service, subject to such limitations $% \left(1\right) =\left(1\right) \left(1\right) \left($ and restrictions, consistent with applicable law, as may apply under the Company's or Buyer's employment policies as in effect at the time of such return. Buyer shall cause the Company to retain all obligations relating to all vacation and personal holidays to which each individual who is an Employee immediately prior to the Closing is entitled as of the Closing under the applicable Company Plan or Benefit Arrangement in effect on the date hereof, and each such Employee shall be entitled to use and to be compensated for such vacation, sick days and personal holidays for the period ending December 31, 1995 under such Company Plan or Benefit Arrangement and for the period commencing January 1, 1996 in accordance with Buyer's policies and practices.
- 9.4. Liabilities Retained or Assumed by Seller. From and after the Closing, the Seller shall, at its sole cost and expense, continue to provide and administer the benefits set forth below that are currently provided through Company Plans and Benefit Arrangements, and shall reimburse Buyer for other benefits to the extent specified below.
 - (a) With respect to all Employees and their dependents, Seller shall provide and be liable for any and all medical, dental, vision and life insurance claims incurred, and other welfare and fringe benefits that are incurred or payable or for which claims for payment are made, on or before the Closing Date. In the case of any individual described in the first or second sentences of Section 9.4(b) who is receiving long-term disability benefits or worker's compensation (or employer's liability) benefits at the Closing, and his or her spouse and eligible dependents, Seller shall also provide and be liable for any and all welfare benefits and fringe benefits in accordance with the terms of the applicable Company Plan or Benefit Arrangement (including workers' compensation program) or a comparable replacement plan or arrangement of Seller, including without limitation any extended benefits, waiver of premium, survivor health, COBRA, conversion and any other applicable continuation of coverage provision, until such time, if any, as such individual returns to active service with the Company and its Subsidiaries, and without limiting the foregoing Seller shall be liable for all claims

under such benefit programs that are incurred prior to the time such individual returns to active service with the Company and its Subsidiaries. In the case of any other individual who is described in the first or second sentences of Section 9.4(b) or who is not described in Section 9.4(b) but is receiving short-term disability (including salary continuation) benefits at the Closing (or who later receives short-term disability or salary continuation benefits in respect of an injury, illness or other condition sustained or occurring on or prior to the Closing), and the spouse and eligible dependents of any such individual, Seller shall reimburse Buyer for any and all welfare benefit claims (other than short-term disability payments) incurred and fringe benefits provided prior to such time, if any, as such individual returns to active service with the Company and its Subsidiaries. Following the return to active service with the Company and its Subsidiaries of an individual described in the second sentence of $% \left(1\right) =\left(1\right) \left(1$ Section 9.4(b), Seller shall continue to be liable for any continuing worker's compensation medical benefits to the individual, but Buyer shall be liable for other welfare and fringe benefits (including dependent coverage) and for any future workers' compensation claims with respect to the individual. For purpose of this Section, a claim will be deemed to have been incurred when an individual is provided with medical, dental, vision or other services that are covered expenses and give rise to the claim; provided that a claim for life insurance or similar death benefits will be deemed to have been incurred at time of death (except that any life insurance waiver-of-premium claims shall be deemed to have been incurred as of the date of disability).

- (b) Seller from and after the Closing shall continue to provide and be liable for long-term disability benefits in accordance with the terms of the applicable Company Plan or Benefit Arrangement (or under a comparable plan or arrangement of Seller) to each individual who as of the Closing is receiving long-term disability benefits or who later becomes entitled to long-term disability benefits relating to an injury, illness or other condition sustained or occurring on or prior to the Closing Date. Seller from and after the Closing shall continue to provide and be liable for workers' compensation benefits and employer's liability benefits in accordance with the terms of the applicable workers' compensation program and applicable law to each individual who as of the Closing is receiving workers' compensation benefits or employer's liability benefits or who later becomes entitled to workers' compensation benefits or employer's liability benefits relating to an injury, illness or other condition sustained or occurring on or prior to the Closing Date.
- (c) Without limiting the foregoing, Seller shall provide and remain liable for any and all continuation of coverage required under Sections 601 through 608 of ERISA and Section 4980B of the Code with respect to any person as to whom the qualifying event (as defined at Section 6503 of ERISA) occurred at or prior to the Closing, other than individuals identified pursuant to the first sentence of Section 9.3.

- (d) Without limiting the foregoing, Seller shall be liable for all Compensation (including deferred compensation and severance) and benefits related claims and liabilities arising prior to or on or after the Closing Date, including without limitation all claims and liabilities under Company Plans and Benefit Arrangements, with respect to any Person (and such Person's spouse and dependents and beneficiaries) who is or becomes a Former Employee or who dies prior to the Closing.
- (e) Any Liability for welfare and fringe benefits for which Seller is liable pursuant to this Section 9 shall not be reflected in the Closing Balance Sheet.
- 9.5. Certain Severance Arrangements. Buyer agrees that any Employee whose employment is involuntarily terminated by Buyer other than for cause during the period commencing on the Closing Date and ending on the day which is six months after the Closing Date and who furnishes Buyer with a fully executed release satisfactory to Buyer (and the period, if any, within which such release may be revoked has expired) shall receive a severance benefit consisting of (i) cash severance determined in accordance with Schedule 9.5, plus (ii) any non-cash benefits to which the Employee would be entitled if the terms of Schedule 9.5A rather than the terms of Schedule 9.5 were applied, in each case based on the Employee's rate of pay, salary grade (or position) and service as of the Closing Date; provided, that Seller shall be liable and shall reimburse Buyer for the amount by which any cash severance determined in accordance with Schedule 9.5 in respect of an involuntary termination other than for cause during such six-month period exceeds the applicable cash severance amount set forth in Schedule 9.5A, in each case based on the Employee's rate of pay, salary grade (or position) and service as of the Closing Date.
- 9.6. Qualified Plans. Buyer shall take such steps as are necessary to enable Employees entitled to receive an eligible rollover distribution (as defined at Section 402(c)(4) of the Code) from Seller's Employee Stock Ownership Plan in connection with the Purchase to transfer the distribution in accordance with Section 402(c) of the Code to a tax-qualified plan maintained by Buyer. the case of Seller's 401(k) Profit Sharing Plan, Seller shall cause account balances for affected participants to be transferred on a non-elective basis pursuant to section 414(1) of the Code to one or more 401(k) plan or plans designated by Buyer. Buyer shall not be obligated to cause its plans to accept (i) any eligible rollover distribution or other transfer in respect of an individual who is not employed by Buyer or its subsidiaries at the time of the transfer, (ii) any assets other than cash or other property acceptable to the trustee of the recipient plan, or (iii) any transfer unless and until Buyer is reasonably satisfied that the transferor plan is, and Seller is reasonably satisfied that the transferee plan is, at the time of transfer, qualified under section 401(a) of the Code. In respect of any Employee described in clause (i) of the preceding sentence who becomes entitled to a distribution under Seller's 401(k) Profit Sharing Plan upon termination of employment after the Closing by the Company and its Subsidiaries, and as to any other Employee with respect to records under Buyer's control, Buyer agrees to cooperate with Seller in obtaining all necessary consents in respect of any distribution to be made from Seller's 401(k) Profit Sharing plan to such individual.

- 9.7. Service Credit. Buyer shall take such steps as are necessary to ensure that for purposes of determining service for eligibility and, for qualified plans other than any defined benefit plan maintained by Buyer, vesting service under any of Buyer's employee benefit plans for which an Employee is otherwise eligible, an Employee's service shall include his or her service for the Company and the Persons in the same Controlled Group as the Company prior to Closing. Nothing herein shall be construed as requiring that pre-Closing service be credited under Buyer's plans for purposes of determining benefit accrual except for purposes of determining any individual's benefit under a vacation or severance plan.
- 9.8. WARN Act. Buyer shall indemnify Seller and its Affiliates and hold each of them harmless from and against any Losses which may be incurred or suffered by any of them under the Worker Adjustment and Retraining Notification Act or any state or Puerto Rico plant closing or notification law arising out of, or relating to, any actions taken by Buyer or the Company or any Subsidiary after the Closing. Seller shall indemnify Buyer and its Affiliates (including, without limitation, the Company and each Subsidiary of the Company from and after the Closing) and hold each of them harmless from and against any Losses that may be incurred or suffered by any of them under the Worker Adjustment and Retraining Notification Act or any state or Puerto Rico plant closing or notification law arising out of, or relating to, any actions taken by Seller or the Company or any Subsidiary of the Company before the Closing.

10. INDEMNIFICATION.

10.1. Indemnification. Seller (in its capacity as indemnifying party, the "Indemnifying Party") hereby agrees to indemnify each of Buyer and its Affiliates (including, without limitation, the Company and each Subsidiary of the Company from and after the Closing) (each in its capacity as indemnified party, an "Indemnitee") and hold each of Buyer and such Affiliates harmless, and Buyer (in its capacity as indemnifying party, the "Indemnifying Party") hereby agrees to indemnify Seller and its Affiliates other than the Company or any Subsidiary of the Company (each in its capacity as indemnified party, an "Indemnitee") and hold each of Seller and such Affiliates harmless, from, against and in respect of any and all Losses arising from or related to any of the following:

10.1.1. Seller. In the case of Seller as Indemnifying Party:

(i) any breach of or inaccuracy in (or any allegation by any third party of facts which, if true as alleged, would constitute such a breach or inaccuracy in) any representation or warranty made by or on behalf of Seller in this Agreement (including, without limitation, the Schedules hereto) or in any Closing Agreement or other document, instrument or certificate delivered pursuant hereto (as such representation or warranty would read if all qualifications as to knowledge, materiality or Material Adverse Effect were deleted therefrom);

- (ii) any breach or violation of any covenant or agreement made by Seller in this Agreement (including, without limitation, the Schedules hereto) or in any document, instrument or certificate delivered pursuant hereto;
 - (iii) any Excluded Liability;
- (iv) any Liability of the Company or any of its Subsidiaries for Taxes (other than Taxes for which Seller indemnifies Buyer pursuant to Section 6.7) of any Person other than the Company or such Subsidiaries;
- (v) any Liability of the Company related to Company Plans and Benefit Arrangements to the extent they relate to Persons other than Employees; or $\frac{1}{2}$
- (vi) any Liability of the Company or any of its Subsidiaries arising solely as a result of the Company or any of its Subsidiaries being a member of a group of companies or other entities controlled by Seller or any other Person (other than the Company or any of its Subsidiaries) prior to the Closing; or
- (vii) any Liability asserted by any Governmental Authority against the Company or any Subsidiary with respect to the escheat of any unclaimed funds or asset which are asserted to have become payable to such Governmental Authority (x) prior to the Closing Date, or (y) subsequent to the Closing Date as a consequence of specific transactions or events that occurred prior to the Closing Date (such as the sale of gift certificates, etc.) to the extent such amounts claimed by such Governmental Authority, when treated in the same manner as other Liabilities in such category on the Company Net Assets Statement (i.e., the obligation to redeem gift certificates),

result in the total obligations of the Company and its Subsidiaries exceeding the amount carried on the Company Net Assets Statement for such category of Liability.

- 10.1.2. Buyer. In the case of Buyer as Indemnifying Party:
- (i) any breach of or inaccuracy in (or any allegation by any third party of facts which, if true as alleged, would constitute such a breach or inaccuracy in) any representation or warranty made by or on behalf of Buyer in this Agreement (including, without limitation, the Schedules hereto) or in any Closing Agreement or other document, instrument or certificate delivered pursuant hereto;
- (ii) any breach or violation of any covenant or agreement made by Buyer in this Agreement (including, without limitation, the Schedules hereto) or in any document, instrument or certificate delivered pursuant hereto;

- (iii) any Liability of Seller or any of its Affiliates arising out of, with respect to or in connection with any Guarantee of any Lease of the Company or any of its Subsidiaries other than to the extent the obligations under such Guarantee or Lease constitute an Excluded Liability; or
- (iv) any Liability incurred by Seller or any of its Affiliates at any time after the Closing arising out of, with respect to or in connection with the Business or any matter or circumstance involving the Company or any of its Subsidiaries, other than (a) any Excluded Liabilities (b) any Losses covered by Section 6.7 or the indemnity in Section 10.1.1. or (c) any Losses arising out of an illegal or tortious course of conduct on the part of Seller or any of its Affiliates.
- 10.2. Time Limitation on Indemnification. Notwithstanding the foregoing, no claim may be made or suit instituted under any provision of this Section 10 $\overline{\mbox{more}}$ than 90 days following the availability of the audited financial statements of Buyer for its fiscal year ending nearest to January 31, 1997 (the "General Survival Period") except for Reserved Claims. The term "Reserved Claims" shall mean (a) all claims as to which any Indemnitee has given any Indemnifying Party notice on or prior to the end of the General Survival Period or, if later, for a claim referred to in clause (c) or (d) of this sentence, on or prior to the end of the period by which such claim may be made in accordance with this Section 10.2, (b) all claims by any Indemnitee based upon an alleged or actual breach of or inaccuracy in the representations or warranties contained in Sections 4.1, 4.19, 5.1 or 5.7 or contained in Section 3.1.2 or 3.1.3(b) of the Preferred Stock Subscription Agreement (but only insofar as said Section 3.1.3(b) relates to the General Corporation Law of the State of Delaware), (c) all claims by any Indemnitee based upon an alleged or actual breach of or inaccuracy in the representations or warranties contained in Section 4.15, (d) all claims based upon an alleged or actual breach of or inaccuracy in the representations or warranties contained in Section 4.14 and all claims pursuant to clause (vii) of Section 10.1.1, (e) all claims pursuant to clauses (ii), (iii), (iv), (v) or (vi) of Section 10.1.1 or clause (ii), (iii) or (iv) of Section 10.1.2 and (f) all claims based upon fraud. As to the Reserved Claims referred to in clause (c) of the definition thereof, no claim may be made or suit instituted after five years from the Closing Date. As to Reserved Claims referred to in clause (d) of the definition thereof, no claim may be made or suit instituted after thirty (30) days after the expiration of the applicable statute of limitations. As to Reserved Claims referred to in clause (a), (b), (e) or (f) of the definition thereof, any claim may be made or suit instituted at any time without limitation.
- 10.3. Monetary Limitations on Seller's Indemnification. Except with respect to claims (i) arising out of the representations or warranties contained in Section 4.1, 4.14 or 4.19, (ii) under clauses (ii), (iii), (iv), (v) or (vi) of Section 10.1.1 hereof, or (iii) referred to in clause (f) of the definition of Reserved Claims, Seller as Indemnifying Party shall not have any obligation to indemnify Buyer or any of its Affiliates as Indemnitees under Section 10.1.1 in respect of any Loss incurred by Buyer and/or any of its Affiliates as Indemnitees unless the

aggregate cumulative total of all Losses (other than Losses arising out of claims referred to in clauses (i), (ii) and (iii) of this sentence) incurred by Buyer and/or any of its Affiliates as Indemnitees exceeds \$10,000,000, whereupon Buyer and each of its Affiliates as Indemnitees shall be entitled to indemnification for the aggregate cumulative amount of such Losses in excess of \$5,000,000. With respect to claims referred to in clauses (i), (ii) or (iii) of the first sentence of this Section 10.3, no such minimum dollar limitation shall apply.

- 10.4. Certain Matters of Construction. References in this Section 10 to claims with respect to or based upon a representation or warranty set forth in a particular Section or Schedule shall be deemed to include without limitation claims relating to such representations or warranties based upon the certificates to be furnished pursuant to Sections 7.1.3 and 8.1.3 hereof.
- 10.5. Third Party Claims. Promptly after the receipt by any Indemnitee of notice of the commencement of any Action against such Indemnitee by a third party (other than any Action relating to Taxes or any Tax Return, which shall be governed by Section 6.7) such Indemnitee shall, if a claim with respect thereto is or may be made against any Indemnifying Party pursuant to this Section 10, give such Indemnifying Party written notice thereof. The failure to give such notice shall not relieve any Indemnifying Party from any obligation hereunder except where, and then solely to the extent that, such failure actually and materially prejudices the rights of such Indemnifying Party. Such Indemnifying Party shall have the right to defend such Action, at such Indemnifying Party's expense and with counsel of its choice reasonably satisfactory to the Indemnitee, provided that the Indemnifying Party so notifies the Indemnitee within 15 days after receipt of such notice and then actually commences the defense of such Action, and otherwise the Indemnitee shall have the right to defend such Action and the Indemnifying Party will reimburse the Indemnitee promptly and periodically for the costs of defending against such Action, including attorneys' fees and expenses. If the Indemnifying Party is defending such Action, the Indemnitee may retain separate co-counsel at its sole cost and expense and may participate in defense of such Action. The Indemnifying Party will not be liable for any judgment or settlement with respect to such Action effected without its prior written consent (unless the Indemnifying Party is not conducting the defense of such Action pursuant to the provisions of this Section 10.5).
- 10.6. No Circular Recovery. Seller hereby agrees that it will not make any claim for indemnification against Buyer, Company or any of its Subsidiaries by reason of the fact that it or any of its officers, directors, agents or other representatives was a controlling person, director, officer, employee, agent or other representative of the Company or any of its Subsidiaries or was serving as such for another Person at the request of the Company or any Subsidiary of the Company (whether such claim is for Losses of any kind or otherwise and whether such claim is pursuant to any statute, Charter, By-law, Contractual Obligation or otherwise) with respect to any Action brought by Buyer or any of its Affiliates against Seller (whether such Action is pursuant to this Agreement, applicable law, or otherwise).

10.7. Nature of Indemnification Payments. Any and all indemnification payments pursuant to this Section 10 shall be deemed for all purposes to be adjustments to the Purchase Price.

10.8. Exclusive Remedy. Notwithstanding anything to the contrary in this Section 10, Section 6.7 hereof (and not this Section 10) shall be the exclusive any breach of or inaccuracy in any representation or warranty contained in Section 4.12 and any claim in respect of Taxes referred to in said Section 6.7, and no claim may be made or suit instituted under Section 6.7 after 30 days after the expiration of the applicable statute of limitations. After the Closing, this Section 10 (other than as set forth in the immediately preceding sentence) will provide the exclusive remedy for any breach of or inaccuracy in any representation or warranty referred to in this Section 10, any breach or violation of any covenant or other agreement referred to in this Section 10 (other than as provided in Section 3.4), or other claim arising out of this Agreement, any Closing Agreement or the transactions contemplated hereby or thereby. Without limiting the generality of the foregoing, Buyer and, effective at the Closing, the Company and its Subsidiaries hereby waive all rights for contribution or any other rights of recovery (except as otherwise provided in this Section 10) with respect to any Losses arising under or related to Environmental Laws that it might have by statute or otherwise against Seller insofar as such rights relate to the real property listed on Schedule 4.5.1 or any real property subject to any Lease. Notwithstanding anything in this Agreement to the contrary, (a) Seller shall not be responsible for indemnifying Buyer or any of its Affiliates (including the Company and any of its Subsidiaries after the Closing) for any Losses pursuant to this Section 10 (to the extent such indemnification relates to environmental conditions, contamination, release of Hazardous Substances or any other environmental matters (each, an "Environmental Condition")) unless (i) Buyer or any of its Affiliates is required to incur such Losses pursuant to any Environmental Law or any Governmental Order or by any Governmental Authority or (ii) in the reasonable judgment of Buyer, investigation, remediation, clean up or any other action is required in connection with such Environmental Condition due to a substantial risk to human health or the environment; provided that Buyer shall

exercise all reasonable efforts to (A) limit or reduce any such remediation or cleanup to the extent reasonable and (B) conduct such remediation or cleanup efficiently and taking into account economic considerations to the same extent as would a prudent business person. Prior to undertaking any remediation, cleanup or other action (other than investigation) referred to in clause (ii) of the immediately preceding sentence with respect to any Environmental Condition which has resulted from the actions of a lessee of the Company or any of its Subsidiaries or a known third party, Buyer shall notify such lessee or known third party of such Environmental Condition in writing and shall request therein that such lessee or known third party undertake or pay for such cleanup, remediation or other action. If such lessee or known third party (i) does not agree in writing, and in Buyer's reasonable judgment does not otherwise agree, to undertake or pay for such cleanup, remediation or other action within 60 days of receipt of such notice from Buyer, (ii) does so agree within such time period but in Buyer's reasonable judgment does not have the resources to comply with such agreement, or (iii) does so agree within such time period but does not

diligently pursue such cleanup, remediation or other action, Buyer may undertake such cleanup, remediation or other action. Buyer shall, prior to or after commencing cleanup, remediation or other action (other than investigation) in accordance with this Section 10.8, take such actions as Seller shall reasonably request, at the cost and expense of Seller, for the purpose of making any such lessee or known third party undertake or pay for such cleanup, remediation or other action, including without limitation pursuing the remedies available to the Company or any of its Subsidiaries under the Stock Purchase Agreement dated as of April 20, 1994 between Marshalls, Inc. and NYDS Holding Limited.

11. CONSENT TO JURISDICTION; JURY TRIAL WAIVER.

- 11.1. Consent to Jurisdiction. Each party to this Agreement, by its execution hereof, (i) hereby irrevocably submits, and agrees to cause each of its Subsidiaries to submit, to the exclusive jurisdiction of the Federal courts located in the County of New York, State of New York, and in the event that such Federal courts shall not have subject matter jurisdiction over the relevant proceeding, then of the state courts located in the County of New York, State of New York, for the purpose of any Action arising out of or based upon this Agreement or any Closing Agreement or relating to the subject matter hereof or thereof or the transactions contemplated hereby or thereby, (ii) hereby waives, and agrees to cause each of its Subsidiaries and Affiliates to waive, to the extent not prohibited by applicable law, and agrees not to assert, and agrees not to allow any of its Subsidiaries and Affiliates to assert, by way of motion, as a defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such proceeding brought in one of the above-named courts is improper, or that this Agreement or any other Closing Agreement, or the subject matter hereof or thereof, may not be enforced in or by such court and (iii) hereby agrees not to commence or to permit any of its Subsidiaries or Affiliates to commence any Action arising out of or based upon this Agreement or any Closing Agreement or relating to the subject matter hereof or thereof other than before one of the above-named courts nor to make any motion or take any other action seeking or intending to cause the transfer or removal of any such Action to any court other than one of the above-named court whether on the grounds of inconvenient forum or otherwise. Each party hereby consents to service of process in any such proceeding in any manner permitted by New York law, and agrees that service of process by registered or certified mail, return receipt requested, at its address specified pursuant to Section 13.9 hereof is reasonably calculated to give actual notice.
- 11.2. WAIVER OF JURY TRIAL. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LAW THAT CANNOT BE WAIVED, EACH OF THE PARTIES HERETO HEREBY WAIVES, AND AGREES TO CAUSE EACH OF ITS SUBSIDIARIES AND AFFILIATES TO WAIVE, AND COVENANTS THAT NEITHER IT NOR ANY OF ITS SUBSIDIARIES OR AFFILIATES WILL ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE) ANY RIGHT TO TRIAL BY JURY IN ANY FORUM IN RESPECT OF ANY ISSUE OR ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT OR ANY OTHER CLOSING AGREEMENT OR THE SUBJECT MATTER HEREOF OR THEREOF OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE TRANSACTIONS

CONTEMPLATED HEREBY, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING. SELLER ACKNOWLEDGES THAT IT HAS BEEN INFORMED BY BUYER THAT THIS SECTION 11.2 CONSTITUTES A MATERIAL INDUCEMENT UPON WHICH BUYER IS RELYING AND WILL RELY IN ENTERING INTO THIS AGREEMENT AND ANY OTHER AGREEMENTS RELATING HERETO OR CONTEMPLATED HEREBY. ANY PARTY HERETO MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION 11.2 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF EACH SUCH PARTY TO THE WAIVER OF ITS RIGHT TO TRIAL BY JURY.

TERMINATION.

- 12.1. Termination of Agreement. This Agreement may be terminated by the parties only as provided below:
 - (a) Buyer and Seller may terminate this Agreement by mutual written consent at any time prior to the Closing.
 - (b) Buyer may terminate this Agreement by giving written notice to Seller at any time prior to the Closing (i) in the event that any representation or warranty made by or on behalf of Seller herein or pursuant hereto or in any Closing Agreement containing qualifications as to materiality or Material Adverse Effect shall have been inaccurate when made, or any other representation or warranty made by or on behalf of Seller herein or in any Closing Agreement shall have been inaccurate in any material respect when made, or in each case if then made would be so inaccurate, and such inaccuracy is not capable of cure or if capable of cure is not so cured within a reasonable period following notice of such inaccuracy, (ii) in the event that Seller materially breaches or violates any covenant or agreement contained herein or in any Closing Agreement to be performed by Seller and such breach or violation is not capable of cure or if capable of cure is not so cured within a reasonable period following notice of such breach or violation, or (iii) if the Closing shall not have occurred on or before January 31, 1996 by reason of the failure of any condition set forth in Section 7 hereof to be satisfied (unless the failure results primarily from the failure of any representation or warranty made by or on behalf of Buyer herein or in any Closing Agreement containing qualifications as to materiality or Material Adverse Effect to be true and correct or any other representation or warranty made by or on behalf of Buyer herein or in any Closing Agreement to be true and correct in all material respects or from the material breach or violation by Buyer of any covenant or agreement contained herein or in any Closing Agreement.
 - (c) Seller may terminate this Agreement by giving written notice to Buyer at any time prior to the Closing (i) in the event that any representation or warranty made by or on behalf of Buyer herein or pursuant hereto or in any Closing Agreement containing qualifications as to materiality or Material Adverse Effect shall have been inaccurate when made, or any other representation or warranty made by or on behalf of

Buyer herein or in any Closing Agreement shall have been inaccurate in any material respect when made, or in each case if then made would be so inaccurate, and such inaccuracy is not capable of cure or if capable of cure is not so cured within a reasonable period following notice of such inaccuracy, (ii) in the event that Buyer materially breaches or violates any covenant or agreement contained herein or in any Closing Agreement to be performed by Buyer and such breach or violation is not capable of cure or if capable of cure is not so cured within a reasonable period following notice of such breach or violation, or (iii) if the Closing shall not have occurred on or before January 31, 1996 by reason of the failure of any condition set forth in Section 8 hereof to be satisfied (unless the failure results primarily from the failure of any representation or warranty made by or on behalf of Seller herein or in any Closing Agreement containing qualifications as to materiality or Material Adverse Effect to be true and correct or any other representation or warranty made by or on behalf of Seller herein or in any Closing Agreement to be true and correct in all material respects or from the material breach or violation by Seller of any covenant or agreement contained herein or in any Closing Agreement).

12.2. Effect of Termination. In the event of the termination of this Agreement pursuant to Section 12.1, all obligations of the parties hereunder (other than the obligations under Sections 6.3, 6.8, 6.9 (insofar as Section 6.9 relates to Buyer and its Subsidiaries), 11.1, 11.2, 13.1, 13.2, 13.8, 13.9, 13.11 and 13.13, which shall survive termination) shall terminate without any liability of any party to any other party; provided, however, that no termination shall relieve any party from any liability arising from or relating to breach prior to termination.

13. MISCELLANEOUS.

- 13.1. Entire Agreement; Waivers. This Agreement, the Closing Agreements and the Confidentiality Agreement constitute the entire agreement among the parties hereto pertaining to the subject matter hereof and thereof and supersede all prior and contemporaneous agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to such subject matter. No waiver of any provision of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), shall constitute a continuing waiver unless otherwise expressly provided nor shall be effective unless in writing and executed (i) in the case of a waiver by Buyer, by Buyer and (ii) in the case of a waiver by Seller, by Seller.
- 13.2. Amendment or Modification. The parties hereto may not amend or modify this Agreement except in such manner as may be agreed upon by a written instrument executed by Buyer and Seller.
- 13.3. Survival, etc. All representations, warranties, covenants and agreements made by or on behalf of any party hereto in this Agreement (including, without limitation, the

Schedules hereto), or pursuant to any document, certificate or other instrument referred to herein or delivered in connection with the transactions contemplated hereby, shall be deemed to have been material and relied upon by the parties hereto, notwithstanding any investigation made by or on behalf of any of the parties hereto or any opportunity therefor (including without limitation the availability for review of any document), and, subject to the provisions of Section 10, shall survive the execution and delivery of this Agreement and the Closing. Neither the period of survival nor the liability of any party with respect to such party's representations, warranties covenants and agreements shall be reduced by any investigation made at any time by or on behalf of any party. If written notice of a claim has been given prior to the expiration of any time period set forth herein for any such notice by a party in whose favor such representations, warranties, covenants or agreements have been made to any party that made such representations, warranties, covenants or agreements, then the relevant representations, warranties, covenants or agreements shall survive as to such claim until such claims have been finally resolved.

- 13.4. Independence of Representations and Warranties. The parties hereto intend that each representation, warranty, covenant and agreement contained herein shall have independent significance. If any party has breached any representation, warranty, covenant or agreement contained herein in any respect, the fact that there exists any other representation, warranty, covenant or agreement relating to the same subject matter (regardless of the relative levels of specificity) that the party has not breached shall not detract from or mitigate the fact that such party is in breach of the first representation, warranty, covenant or agreement.
- 13.5. Severability. In the event that any provision hereof (including, without limitation, any of the provisions of Section 6.9 hereof) would, under applicable law, be invalid or unenforceable in any respect, such provision shall (to the extent permitted under applicable law) be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable law. The provisions hereof (including, without limitation, each of the provisions of Section 6.9 hereof) are severable, and in the event any provision hereof should be held invalid or unenforceable in any respect, it shall not invalidate, render unenforceable or otherwise affect any other provision hereof.
- 13.6. Knowledge. Whenever reference is made herein to the knowledge of any Person with respect to any matter, it is understood that such knowledge extends only to the officers of such Person (and in the case of Seller, the officers of the Company or any of its Subsidiaries) having responsibility for the areas of such Person's (and in the case of Seller, also the Company's or any of its Subsidiaries') business covering such matter, which officers have made an inquiry that is reasonably appropriate to determine the accuracy of the statement in question or, in the case of Actions, have made an inquiry that is reasonably appropriate to determine the existence of Actions threatened in writing against such Person (and in the case of Seller, against the Company or any of its Subsidiaries).

- 13.7. Successors and Assigns. All of the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective transferees, successors and assigns (each of which such transferees, successors and assigns shall be deemed to be a party hereto for all purposes hereof); provided, however, that (i) Seller may not assign or transfer (by operation of law or otherwise) any of its rights or obligations hereunder, (ii) Buyer may assign its rights and obligations hereunder (other than to issue the Buyer Stock) to an Affiliate of Buyer (but Buyer shall not be relieved of its obligations hereunder), and (iii) no transfer or assignment by any party shall relieve such party of any of its obligations hereunder.
- 13.8. Notices. Any notices or other communications required or permitted hereunder shall be sufficiently given if in writing (including telecopy or similar teletransmission), addressed as follows:

If to Seller, to it at: Melville Corporation

One Theall Road Rye, New York 10580 Telecopier: 914-925-4052

Attention: Chief Executive Officer, Chief

Financial

Officer and General Counsel

With a copy to: Davis Polk & Wardwell

450 Lexington Avenue New York, New York 10017 Telecopier: 212-450-5744 Attention: Dennis S. Hersch

If to Buyer, to it at: The TJX Companies, Inc.

770 Cochituate Road Framingham, MA 01701 Telecopier: 508-390-2457 Attention: President and General Counsel

With a copy to: Ropes & Gray

One International Place Boston, MA 02110 Telecopier: 617-951-7050

Attention: Arthur G. Siler, Esq.

Unless otherwise specified herein, such notices or other communications shall be deemed received (a) in the case of any notice or communication sent other than by mail, on the date actually delivered to such address (evidenced, in the case of delivery by overnight courier, by confirmation of delivery from the overnight courier service making such delivery, and in the

case of a telecopy, by receipt of a transmission confirmation form or the addressee's confirmation of receipt), or (b) in the case of any notice or communication sent by mail, three Business Days after being sent, if sent by registered or certified mail, with first-class postage prepaid. Each of the parties hereto shall be entitled to specify a different address by giving notice as aforesaid to each of the other parties hereto.

- 13.9. Public Announcements. At all times at or before the Closing, no party hereto will issue or make any reports, statements or releases to the public or generally to the suppliers or other Persons to whom Buyer or the Company sells goods or provides services or with whom Buyer or the Company otherwise has significant business relationships with respect to this Agreement or the transactions contemplated hereby without the consent of the other party hereto, which consent shall not be unreasonably withheld. If any party hereto is unable to obtain, after reasonable effort, the approval of its public report, statement or release from the other parties hereto and such report, statement or release is, in the opinion of legal counsel to such party, required by law in order to discharge such party's disclosure obligations, then such party may make or issue the legally required report, statement or release and promptly furnish the other parties with a copy thereof. Each party hereto will also obtain the prior approval by the other parties hereto of any press release to be issued immediately following the Closing announcing the consummation of the transactions contemplated by this Agreement.
- 13.10. Headings, etc. Section and subsection headings are not to be considered part of this Agreement, are included solely for convenience, are not intended to be full or accurate descriptions of the content thereof and shall not affect the construction hereof.
- 13.11. Third Party Beneficiaries. Except as otherwise provided in Section 10, nothing in this Agreement is intended or shall be construed to entitle any Person other than the parties, the Company or their respective transferees, successors and assigns permitted hereby to any claim, cause of action, remedy or right of any kind.
- 13.12. Counterparts. This Agreement may be executed in any number of counterparts and by the different parties on separate counterparts each of which shall be deemed an original, but all of which together shall constitute but one and the same instrument.
- 13.13. Governing Law. This Agreement shall be governed by and construed in accordance with the domestic substantive law of the State of New York, without giving effect to any choice or conflict of law provision or rule that would cause the application of the law of any other jurisdiction.

IN WITNESS WHEREOF, the parties hereto, intending to be legally bound hereby, have caused this Stock Purchase Agreement to be executed, as of the date first above written by their respective officers thereunto duly authorized.

SELLER: MELVILLE CORPORATION

By /s/ H. Rosenthal
Name: Harvey Rosenthal
Title: President

BUYER: THE TJX COMPANIES, INC.

> By /s/ Donald G. Campbell Name: Donald G. Campbell Title: Chief Financial Officer

	EXHIBIT 10	. 1			
PREFERRED STOCK SUBSCRIPTION AGREEMENT					
hetween					
The TJX Companies, Inc.					
as Issuer					
and					
Melville Corporation,					
as Subscriber					
Dated as of					
	between The TJX Companies, Inc. as Issuer and Melville Corporation, as Subscriber	PREFERRED STOCK SUBSCRIPTION AGREEMENT between The TJX Companies, Inc. as Issuer and Melville Corporation, as Subscriber			

TABLE OF CONTENTS

1.	1.1. 1.2. 1.3.	Incorporation of Purchase Agreement Definitions	1 1
2.	ISSUANCE OF 2.1. 2.2. 2.3.	PREFERRED SHARES. Issuance Time and Place of Closing. Delivery.	3
3.	REPRESENTATI 3.1. 3.2. 3.3. 3.4. 3.5.	IONS AND WARRANTIES OF ISSUER. Corporate Matters, etc. Litigation, etc. Disclosure. Change in Condition. Tax Matters.	3 6 6
4.	REPRESENTATI 4.1. 4.2. 4.3. 4.4. 4.5.	IONS AND WARRANTIES OF SUBSCRIBER. Corporate Matters	9 9 9 10
5.	ACCESS TO PF	REMISES AND INFORMATION	11
6.	CONDITIONS 7 6.1. 6.2. 6.3.	TO THE OBLIGATION TO CLOSE OF ISSUER	12 12
7.	CONDITIONS 7.1. 7.2. 7.3. 7.4.	TO THE OBLIGATION TO CLOSE OF SUBSCRIBER	12 13 13
8.	MISCELLANEOU 8.1. 8.2. 8.3.	JS Entire Agreement; Waivers Amendment or Modification Survival	13 13

8.4.	Knowledge	14
8.5.	Successors and Assigns	14
8.6.	Notices	14
8.7.	Headings, etc	15
	Third-Party Beneficiaries	
8.9.	Preparation for Closing	15
8.10.	Counterparts	15
	Governing Law	
8.12.	Termination	15

EXHIBITS

 ${\bf Exhibit} \ {\bf A} \ {\bf \cdot} \ {\bf Certificate} \ {\bf of} \ {\bf Designation} \ {\bf of} \ {\bf Series} \ {\bf D} \ {\bf Cumulative} \ {\bf Preferred} \ {\bf Stock}$

 ${\bf Exhibit} \ {\bf B} \ {\bf \cdot} \ {\bf Certificate} \ {\bf of} \ {\bf Designation} \ {\bf of} \ {\bf Series} \ {\bf E} \ {\bf Cumulative} \ {\bf Preferred} \ {\bf Stock}$

PREFERRED STOCK SUBSCRIPTION AGREEMENT

This PREFERRED STOCK SUBSCRIPTION AGREEMENT (this "Agreement") is made as of the ___ day of _____, 1995, between The TJX Companies, Inc., a Delaware corporation (the "Issuer"), and Melville Corporation, a New York corporation (the "Subscriber").

Recitals

- 1. Issuer and Subscriber are parties to a Stock Purchase Agreement dated as of October 14, 1995 (the "Purchase Agreement") pursuant to which Issuer has agreed to purchase from the Subscriber, and the Subscriber has agreed to sell to Issuer, all of the issued and outstanding shares of capital stock (the "Target Shares") of Marshalls of Roseville, Minn., Inc., a Minnesota corporation.
- 2. As part of the Purchase Price for the Target Shares, Issuer has agreed to issue and sell to the Subscriber, and the Subscriber has agreed to acquire and accept from Issuer, the Preferred Shares (as defined below).

Agreement

Therefore, in consideration of the foregoing and the mutual agreements and covenants set forth below and in the Purchase Agreement, the parties hereto hereby agree as follows:

1. DEFINITIONS. For purposes of this Agreement:

Term

- 1.1. Incorporation of Purchase Agreement Definitions. Capitalized terms used in this Agreement and not otherwise defined herein shall have the same meanings herein as the meanings ascribed to such terms in the Purchase Agreement.
- 1.2. Cross Reference Table. The following terms defined in this Agreement in the Sections set forth below shall have the respective meanings therein defined:

Definition

"Conversion Shares" "Issuer"	Section 3.1. Preamble
"Purchase Agreement" "Securities Act"	Recitals Section 4.4(
"Subscriber" "Target Shares"	Preamble Recitals

- - 1.3.1. Common Stock. The term "Common Stock" shall mean the Common Stock, par value \$1 per share, of Issuer.
 - 1.3.2. Commission. The term "Commission" shall mean the Securities and Exchange Commission.
 - 1.3.3. Exchange Act. The term "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
 - 1.3.4. Exchange Act Documents. The term "Exchange Act Documents" shall mean each of the following:
 - (i) Issuer's Annual Reports on Form 10-K for each of its fiscal years ended January 28, 1995, January 29, 1994 and January 30, 1993:
 - (ii) Issuer's Quarterly Reports on Form 10-Q for the fiscal quarters ended April 29, 1995 and July 29, 1995;
 - (iii) Issuer's proxy or information statement relating to meetings of, or actions taken without a meeting by, the stockholders of Issuer since January 28, 1995; and
 - (iv) the Current Reports on Form 8-K and all other reports, statements and schedules filed by Issuer under the Exchange Act since January 28, 1995.
 - 1.3.5. Preferred Shares. The term "Preferred Shares" shall mean the Series D Shares and the Series E Shares.
 - 1.3.6. Securities Act. The term "Securities Act" is defined in Section 3.2.
 - 1.3.7. Series D Preferred Stock. The term "Series D Preferred Stock" shall mean the Series D Cumulative Convertible Preferred Stock, par value \$1 per share, of Issuer containing the terms and provisions set forth in the Certificate of Designation attached hereto as Exhibit A.
 - 1.3.8. Series E Preferred Stock. The term "Series E Preferred Stock" shall mean the Series E Cumulative Convertible Preferred Stock, par value \$1 per share, of Issuer containing the terms and provisions set forth in the Certificate of Designation attached hereto as Exhibit B.

- 1.3.9. Series D Shares. The term "Series D Shares" shall mean 250,000 shares of Series D Preferred Stock.
- 1.3.10. Series E Shares. The term "Series E Shares" shall mean 1,500,000 shares of Series E Preferred Stock.

ISSUANCE OF PREFERRED SHARES.

- 2.1. Issuance. Upon the terms, subject to the conditions, and in reliance on the representations, warranties and covenants set forth herein, Issuer agrees to issue and sell to Subscriber, and Subscriber agrees to acquire and accept from Issuer, the Preferred Shares at the Closing as part of the Purchase Price.
- 2.2. Time and Place of Closing. The closing of the issuance and sale of the Preferred Shares (the "Closing") shall take place at the same location as, and contemporaneously with, the closing under the Purchase Agreement (the day on which the Closing takes place being referred to herein as the "Closing Date").
- 2.3. Delivery. At the Closing, Issuer will deliver to Subscriber a certificate or certificates evidencing all of the Series D Shares registered in the name of Subscriber and a separate certificate or certificates evidencing all of the Series E Shares registered in the name of Subscriber.
- 3. REPRESENTATIONS AND WARRANTIES OF ISSUER. In order to induce Subscriber to enter into and perform this Agreement and to consummate the transactions contemplated hereby, Issuer represents and warrants to Subscriber as follows:

3.1. Corporate Matters, etc.

3.1.1. Incorporation and Authority of Issuer. Issuer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority, corporate and otherwise, to enter into this Agreement, to carry out and perform its obligations hereunder, to consummate the transactions contemplated hereby and to carry on its business as currently conducted, except for the failure to have any power and authority (other than corporate power and authority) that has not had and could not reasonably be expected to have a Material Adverse Effect on Issuer. Issuer is duly qualified or licensed to do business as a foreign corporation, and is in good standing as such, in all jurisdictions where the nature of Issuer's activities or its ownership or leasing of property require such qualification, except for such failures to be so qualified as have not had and will not have a Material Adverse Effect on Issuer. Issuer has heretofore furnished to Subscriber a true and complete copy of the Charter and By-laws of Issuer in the form currently in effect and as will be in effect as of the Closing.

3.1.2. Authorization and Enforceability. This Agreement has been duly authorized, executed and delivered by Issuer and is Enforceable against Issuer. When issued in accordance with the terms of this Agreement, the Preferred Shares will be duly authorized, validly issued, fully paid and nonassessable and will be free and clear of any Lien or other right or claim (or any restriction on transfer or voting) and will not be subject to any preemptive or similar rights. Issuer has authorized and reserved for issuance upon conversion or redemption of the Preferred Shares a sufficient number of shares of Common Stock (the "Conversion Shares"), and the Conversion Shares will, upon such issuance in accordance with the terms of the Charter of Issuer, be duly authorized, validly issued, fully paid and nonassessable and will be free and clear of any Lien or other right or claim (or any restriction on transfer or voting) and will not be subject to any preemptive or similar rights.

3.1.3. Non-Contravention, etc.

(a) No approval, consent, waiver, authorization or other order of, and no filing, registration, qualification or recording with, any Governmental Authority or any other Person is required to be obtained or made by or on behalf of Issuer or any of its Subsidiaries in connection with the execution, delivery or performance of this Agreement and the consummation of the transactions contemplated hereby, except for (i) satisfaction of the requirements of the Hart-Scott-Rodino Antitrust Improvements of 1976, as amended (the "HSR Act") and (ii) any item required to be obtained from or made with any Person other than a Governmental Authority the failure to obtain or make which, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect on Issuer nor a material adverse effect on the ability of Issuer to consummate the transactions contemplated hereby. Neither the execution, delivery and performance of this Agreement nor the consummation of any of the transactions contemplated hereby (including, without limitation, performance by Issuer of its obligations under the other Closing Agreements and its obligations in respect of the Preferred Shares in accordance with their terms) does or will constitute, result in or give rise to (i) a breach or violation or default under any Legal Requirement applicable to Issuer or any of its Subsidiaries (assuming the accuracy of the representations and warranties of Subscriber in Section 4.4), (ii) a breach of or a default under any Charter or By-Laws provision of Issuer or any of its Subsidiaries, (iii) the acceleration of the time for performance of any obligation under any Contractual Obligation of Issuer or any of its Subsidiaries, (iv) the imposition of any Lien upon or the forfeiture of any asset of Issuer or any of its Subsidiaries, (v) a breach of or a default under any Contractual Obligation of Issuer or any of its Subsidiaries, or (vi) termination, right of termination, modification of terms or change in benefits or burdens under any Contractual Obligation of Issuer or any of its Subsidiaries, other than in the case of clauses (i), (iii), (iv), (v) and (vi) such as, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect on Issuer nor a material adverse effect on the ability of Issuer to consummate the transactions contemplated hereby (including, without limitation, Issuer's ability to

-4-

perform its obligations under the other Closing Agreements and its obligations in respect of the Preferred Shares in accordance with their terms).

- (b) Neither the execution, delivery or performance of this Agreement, the consummation of the transactions contemplated hereby (including, without limitation, performance by Issuer of its obligations under the other Closing Agreements and its obligations in respect of the Preferred Shares in accordance with their terms), nor the terms of the Preferred Shares, do or will constitute, result in or give rise to a breach or violation of or default under the Series A Agreements (as defined in the Standstill and Registration Rights Agreement) or any provision of the General Corporation Law of the State of Delaware.
- ${\tt 3.1.4.} \quad {\tt Capitalization.} \quad {\tt The \ authorized \ capital \ stock \ of \ Issuer}$ consists of 150,000,000 shares of Common Stock and 5,000,000 shares of preferred stock, par value \$1 per share ("Issuer Preferred Stock"), and of such shares of Issuer Preferred Stock 250,000 shares have been designated as Series A Cumulative Convertible Preferred Stock (the "Series A Preferred Stock"), 1,650,000 shares have been designated as Series C Cumulative Convertible Preferred Stock (the "Series C Preferred Stock"), 250,000 shares have been designated as Series D Preferred Stock and 1,500,000 shares have been designated as Series E Preferred Stock. As of September 30, 1995, there were issued and outstanding 72,407,253 shares of Common Stock, 250,000 shares of Series A Stock and 1,650,000 shares of Series C Stock. Other than the issued and outstanding Issuer Preferred Stock and other than equity securities of Issuer issued to directors, officers or employees of Issuer or its Subsidiaries in connection with their service to Issuer or its Subsidiaries ("Employee Securities"), as of September 30, 1995, there were no other securities of Issuer outstanding convertible into or exchangeable for capital stock or other voting securities of Issuer or any other outstanding options or rights to acquire capital stock, voting securities or securities convertible into or exchangeable for capital stock or voting securities of Issuer. All outstanding shares of capital stock of Issuer have been duly authorized and validly issued and are fully paid and nonassessable. Except as contemplated by this Agreement or as set forth in this Section and except for changes since September 30, 1995 resulting from the grant or exercise of Employee Securities since such date, there are outstanding (a) no shares of capital stock or other voting securities of Issuer, (b) no securities of Issuer convertible into or exchangeable for shares of capital stock or voting securities of Issuer, and (c) no options or other rights to acquire from Issuer, and no obligation of Issuer to issue, any capital stock, voting securities or securities convertible into or exchangeable for capital stock or voting securities of Issuer (the items in clauses (a), (b) and (c) being referred to collectively as the Securities"). Except as set forth in the Charter of Issuer or in connection with Employee Securities, there are no outstanding obligations of Issuer or any of its Subsidiaries to repurchase, redeem or otherwise acquire any Issuer Securities.

- 3.2. Litigation, etc. There is no Action against Issuer or any of its Subsidiaries, pending or, to the knowledge of Issuer, threatened, which could reasonably be expected to have a Material Adverse Effect on Issuer. There is no Action pending or, to the knowledge of Issuer, threatened, that seeks recission of, seeks to enjoin the consummation of, or otherwise relates to, this Agreement or any of the transactions contemplated hereby and that could reasonably be expected to have a Material Adverse Effect on Issuer or a material adverse effect on Issuer's ability to consummate the transactions contemplated hereby. No Governmental Order specifically directed at Issuer or any of its Subsidiaries has been issued which has had or could reasonably be expected to have a Material Adverse Effect on Issuer.
- 3.3. Disclosure. Issuer has heretofore delivered to Subscriber true and complete copies of each of the Exchange Act Documents and each registration statement of Issuer filed pursuant to the Securities Act of 1933, as amended (the "Securities Act"), since January 28, 1995. Each Exchange Act Document, when filed with the Commission, conformed in all material respects to the requirements of the Exchange Act and none of such documents contained an untrue statement of a material fact or omitted to state a material fact necessary to make the statements therein, in the light of the circumstances under which they are made, not misleading. Each such registration statement, as amended or supplemented, if applicable, filed pursuant to the Securities Act as of the date such registration statement or amendment thereto became effective did not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading. The consolidated financial statements, and the related notes thereto, included in the Exchange Act Documents were prepared in accordance with generally accepted accounting principles consistently applied throughout the periods specified therein (except for such changes as are noted therein, all of which changes have been concurred in by Coopers) and present fairly in all material respects the financial position of Issuer and its Subsidiaries as of the dates indicated therein and the results of their operations and cash flows for the periods covered thereby, subject in the case of interim financial statements to an absence of footnotes and to normal year-end audit adjustments which will not in the aggregate be material.
- 3.4. Change in Condition. Except for the matters set forth in Schedule 3.4 or in connection with the sale of all of the issued and outstanding capital stock of Hit or Miss, Inc. or as contemplated by this Agreement, the Purchase Agreement or any other Closing Agreement, since July 29, 1995:
 - (a) The business of Issuer has been conducted only in the ordinary course of business consistent with past practice;
 - (b) Neither Issuer nor any of its Subsidiaries has, except in the ordinary course of business consistent with past practice:
 - (i) incurred or otherwise become liable in respect of any Debt or become liable in respect of any Guarantee, other than Debt or any Guarantee

between Issuer and its wholly owned Subsidiaries or between wholly owned Subsidiaries of Issuer;

- (ii) mortgaged or pledged any Asset or subjected any material Asset to any Lien;
- (iii) declared or made any Distribution (other than (A) cash dividends on its capital stock in usual amounts, (B) repurchases of outstanding capital stock through Issuer's publicly announced share repurchase program and other repurchases not material in amount and (C) Distributions from any Subsidiary of Issuer to Issuer);
- (iv) sold, leased to others or otherwise disposed of any of its material Assets except as contemplated by clause (iii) of this Section 3.4(b);
- (v) purchased a material amount of Equity Securities of any Person other than of a direct or indirect wholly owned Subsidiary of Issuer, or a material amount of assets (other than inventory) of any Person or assets constituting a business, or been party to any merger, consolidation or other business combination or entered into any Contractual Obligation relating to any such purchase, merger, consolidation or business combination;
- (vi) made any loan, advance or capital contribution to or investment in any Person material in amount other than loans, advances or capital contributions to or investments in or to its wholly owned Subsidiaries and other than advances to suppliers in the ordinary course of business;
- (vii) canceled or compromised any Debt or claim in any material amount other than those between Issuer and a wholly owned Subsidiary or between wholly owned Subsidiaries of Issuer or owed by Issuer or any wholly owned Subsidiary of Issuer;
- (viii) sold, transferred, licensed or otherwise disposed of any material intellectual or intangible property rights;
- (ix) waived or released or permitted to lapse any right of material value; or
- $\mbox{\ensuremath{(x)}}$ instituted, settled or agreed to settle any material Action;
- (c) Neither Issuer nor any of its Subsidiaries has had any change in its relationships with its employees, agents, customers or suppliers, except as has not and could not reasonably be expected to have a Material Adverse Effect;

- (d) Except for Employee Securities, there has been no amendment of any material provision of any Equity Security of Issuer;
- (e) Neither Issuer nor any of its Affiliates has entered into any Contractual Obligation to do any of the things referred to in clauses (a) through (d) above; and
 - (f) No Material Adverse Effect on Issuer has occurred.

3.5. Tax Matters.

- 3.5.1. To Issuer's knowledge, (a) all material Tax Returns required to be filed on or before the date hereof by, or with respect to Issuer or any Subsidiary have been duly and timely filed (taking into account extensions); (b) no position is reflected in a Tax Return referred to in (a) for which the applicable limitation period has not expired (and for which a closing agreement has not been entered into) which (x) was not, at the time such Tax Return was filed, supported by substantial authority (as determined for purposes of Section 6662 of the Code, or any predecessor provision, and any comparable provisions of applicable federal, state, or local tax statutes, rules or regulations) and (y) would have a Material Adverse Effect if decided against the taxpayer; (c) Issuer and its Subsidiaries have timely paid, withheld or made provision for all Taxes shown as due and payable on any Tax Return and have timely paid, withheld, or made provision for all material Taxes, whether or not shown on any Tax Return; (d) no Liens for Taxes upon the assets of Issuer or any Subsidiary exist; (e) no claim has ever been made by an authority in a jurisdiction where any of Issuer and its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
- 3.5.2. Issuer and each of its Subsidiaries is a member of the affiliated group, within the meaning of Section 1504(a) of the Code, of which Issuer is the common parent (the "Affiliated Group"), and such Affiliated Group files a consolidated federal Income Tax Return. Neither Issuer nor any of its Subsidiaries has at any time been a member of an affiliated group filing a consolidated federal Income Tax Return other than a group, the common parent of which is Issuer. To Issuer's knowledge, all Income Taxes shown on any Tax Return of the Affiliated Group have been paid for each taxable period during which any of Issuer and its Subsidiaries was a member of the Affiliated Group.
- 3.5.3. To Issuer's knowledge, each of Issuer and its Subsidiaries has withheld and paid all material Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder, foreign person, or other third party.
- $3.5.4.\,$ Other than the proposed assessment made by the IRS regarding the receipt of construction allowances, there is no dispute or claim concerning any material

Tax liability of any of Issuer and its Subsidiaries either (A) claimed or raised by any taxing authority in writing or (B) as to which Issuer has knowledge based upon personal contact with any agent of such taxing authority.

- 4. REPRESENTATIONS AND WARRANTIES OF SUBSCRIBER. In order to induce Issuer to enter into and perform this Agreement and to consummate the transactions contemplated hereby, Subscriber represents and warrants to Issuer as follows (and, as provided in Section 4.4, agrees with Issuer):
- 4.1. Corporate Matters. Subscriber is a corporation duly organized, validly existing and in good standing under the laws of the State of New York and has all requisite corporate power and authority to enter into this Agreement, to carry out and perform its obligations hereunder and to consummate the transactions contemplated hereby.
- 4.2. Authorization and Enforceability. This Agreement has been duly authorized, executed and delivered by Subscriber and is Enforceable against Subscriber.
- 4.3. Non-Contravention, etc. No approval, consent, waiver, authorization or other order of, and no filing, registration, qualification or recording with, any Governmental Authority or any other Person (other than any party to any Lease-In other than the Company or any Subsidiary) is required to be obtained or made by or on behalf of Subscriber or the Company or any of its Subsidiaries in connection with the execution, delivery or performance of this Agreement and the consummation of the transactions contemplated hereby, except for (i) satisfaction of the requirements of the HSR Act, (ii) items listed on Schedule 4.1.4 to the Purchase Agreement, which shall have been obtained or made and shall be in full force and effect at the Closing (subject to the materiality exception set forth at the end of the next sentence) and (iii) any other of the foregoing items required to be obtained from or made with any Person other than any Governmental Authority the failure to obtain or make which, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect nor a material adverse effect on the ability of Subscriber to consummate the transactions contemplated hereby. Except as set forth on Schedule 4.1.4 to the Purchase Agreement, neither the execution, delivery and performance of this Agreement nor the consummation of any of the transactions contemplated hereby (including, without limitation, the execution, delivery and performance of the other Closing Agreements) does or will constitute, result in or give rise to (i) a breach or violation or default under any Legal Requirement applicable to Subscriber, the Company or any of its Subsidiaries, (ii) a breach of or a default under any Charter or By-Laws provision of Subscriber, the Company or any of its Subsidiaries, (iii) the acceleration of the time for performance of any obligation under any Contractual Obligation (other than any Lease-In) of Subscriber, the Company or any of its Subsidiaries, (iv) the imposition of any Lien upon or the forfeiture of any Asset, other than any Asset held under any Lease-In, (v) a breach of or a default under any Contractual Obligation (other than any Lease-In) of Subscriber, the Company or any of its Subsidiaries, or (vi) right to any severance payments other than by operation of law (including without limitation if such payments become due only if employment is terminated following the

Closing), termination, right of termination, modification of terms or change in benefits or burdens under any Contractual Obligation (other than any Lease-In), other than in the case of clauses (i) through (vi) such as, individually or in the aggregate, have and could reasonably be expected to have neither a Material Adverse Effect nor a material adverse effect on the ability of Subscriber to consummate the transactions contemplated hereby.

4.4. Investment Intent.

- (a) Subscriber is acquiring the Preferred Shares hereunder for its own account, for investment, and not with a view to, or for sale in connection with, any distribution thereof within the meaning of the Securities Act (as hereafter defined).
- (b) Subscriber understands and agrees that the Preferred Shares and, if any are issued, the Conversion Shares will not be registered or qualified under the Securities Act or state "blue-sky" or other securities laws and therefore cannot be resold unless they are registered under the Securities Act and applicable state laws or unless an exemption from such registration requirement is available.
- (c) Subscriber is able to bear the economic risk of holding the Preferred Shares and, if any are issued, the Conversion Shares for an indefinite period of time and is experienced and has such knowledge and experience in financial and business matters that it is capable of evaluating the risks and merits of acquiring the Preferred Shares, and, if any are issued, the Conversion Shares. Subscriber acknowledges that the Preferred Shares and, if any are issued, the Conversion Shares will bear a legend to the effect that transfers are restricted unless (i) the transfer is exempt from the registration requirements under the Securities Act and Issuer receives an opinion of counsel reasonably satisfactory to Issuer to that effect or (ii) the transfer is made pursuant to an effective registration statement under the Securities Act.
- (d) Subscriber understands that Issuer is under no obligation to effect a registration of the Preferred Shares or, if any are issued, the Conversion Shares under the Securities Act, except to the extent set forth in the Standstill and Registration Rights Agreement.
- (e) Subscriber is an Accredited Investor within the definition set forth in Rule 501(a) of the Securities Act.
- (f) Nothing in this Section 4.4 shall limit or qualify the representations, warranties, covenants or agreements made by Issuer herein or in the Purchase Agreement or any other Closing Agreement or in any certificate or document delivered pursuant hereto or thereto. Issuer acknowledges and agrees that the purpose of this Section 4.4 is solely to ensure that the sale of the Preferred Shares pursuant hereto complies with the distribution restrictions of the Securities Act and the distribution restrictions of other applicable securities laws.

- 4.5. Litigation. There is no Action against the Company or any Subsidiary, pending or, to the knowledge of Subscriber, threatened, which could reasonably be expected to have a Material Adverse Effect, except for such of the foregoing as are described in Schedule 4.17 to the Purchase Agreement. Except as set forth on Schedule 4.17 to the Purchase Agreement, there is no Action pending or, to the knowledge of Subscriber, threatened with respect to which Subscriber or any of its Affiliates, on the one hand, and the Company or any of its Subsidiaries, on the other hand, are or would be parties. There is no Action pending or, to the knowledge of Subscriber, threatened, that seeks rescission of, seeks to enjoin the consummation of, or otherwise relates to, this Agreement or any of the transactions contemplated hereby and that could reasonably be expected to have a Material Adverse Effect or a material adverse effect on Subscriber's ability to consummate the transactions contemplated hereby. No Governmental Order specifically directed at the Company or any of its Subsidiaries has been issued which has had or could reasonably be expected to have a Material Adverse Effect.
- 5. ACCESS TO PREMISES AND INFORMATION. Prior to the Closing, Issuer will permit Subscriber and its representatives to have access to its premises and documents, books and records and to make copies during normal business hours (or to have copies made and delivered to Subscriber) of such financial and operating data and other information with respect to Issuer and its Subsidiaries as Subscriber or any of its representatives shall reasonably request; provided, however, that Subscriber and its representatives shall not have any such access to, or right to copies of, any competitively sensitive information of Issuer or any of its Affiliates. In addition, Issuer shall cause its management (including senior officers) to be available to Subscriber at such times, and from time to time, as Subscriber may reasonably request in connection with the transactions contemplated hereby and to discuss the business and affairs of Issuer and its Subsidiaries; provided, however, that Subscriber shall not be entitled to receive as a result of such availability any competitively sensitive information of Issuer or any of its Affiliates. In addition, so long as the Standstill and Registration Rights Agreement remains in effect, Issuer will deliver to Subscriber copies of all of Issuer's filings with the Commission pursuant to the Exchange Act or the Securities Act no later than five Business Days following the date on which the same are filed with the Commission.
- 6. CONDITIONS TO THE OBLIGATION TO CLOSE OF ISSUER. The obligation of Issuer at the Closing to issue and sell the Preferred Shares is subject to the satisfaction, at or prior to the Closing, of all the following conditions, compliance with which, or the occurrence of which, may be waived prior to the Closing in writing by Issuer in its sole discretion:
 - 6.1. Representations, Warranties and Covenants.
 - 6.1.1. Continued Accuracy of Representations and Warranties. All representations and warranties of Subscriber contained in this Agreement that include qualifications as to materiality or Material Adverse Effect shall be true and correct as of the Closing and all other representations and warranties of Subscriber contained in

this Agreement shall be true and correct in all material respects as of the Closing, in each case with the same force and effect as if such representations and warranties were made at and as of the Closing.

- 6.1.2. Performance of Agreements. Subscriber shall have performed and satisfied in all material respects all covenants and agreements required by this Agreement to be performed or satisfied by it at or prior to the Closing.
- 6.1.3. Closing Certificate. At the Closing, Subscriber shall furnish to Issuer an unqualified certificate, signed by the President or Chief Financial Officer of Subscriber, dated the Closing Date, to the effect that the conditions specified in Sections 6.1.1 and 6.1.2 hereof have been satisfied.
- 6.2. Conditions under Purchase Agreement. All conditions to the closing under the Purchase Agreement set forth in Section 7 thereof shall have been satisfied or waived in accordance with the provisions of said Section 7.
- 6.3. Other Agreements. At or prior to the Closing, Subscriber shall have entered into the Standstill and Registration Rights Agreement, such agreement being in substantially the form thereof attached as an exhibit to the Purchase Agreement without change other than such changes as may be reasonably satisfactory to Issuer.
- 7. CONDITIONS TO THE OBLIGATION TO CLOSE OF SUBSCRIBER. The obligations of Subscriber at the Closing to acquire and accept the Preferred Shares is subject to the satisfaction, at or prior to the Closing, of all of the following conditions, compliance with which, or the occurrence of which, may be waived prior to the Closing in writing by Subscriber in its sole discretion:
 - 7.1. Representations, Warranties and Covenants.
 - 7.1.1. Continued Accuracy of Representations and Warranties. All representations and warranties of Issuer contained in this Agreement that include qualifications as to materiality or Material Adverse Effect shall be true and correct as of the Closing and all other representations and warranties of Issuer contained in this Agreement shall be true and correct in all material respects as of the Closing, in each case with the same force and effect as if such representations and warranties were made at and as of the Closing.
 - 7.1.2. Performance of Agreements. Issuer shall have performed and satisfied in all material respects all covenants and agreements required by this Agreement to be performed or satisfied by Issuer at or prior to the Closing.
 - 7.1.3. Officer's Certificate. At the Closing, Issuer shall furnish to Subscriber an unqualified certificate signed by the President or Senior Vice President Finance

and Chief Financial Officer of Issuer dated the Closing Date, to the effect that the conditions specified in Sections 7.1.1 and 7.1.2 hereof have been satisfied

- 7.2. Conditions under Purchase Agreement. All conditions to the closing under the Purchase Agreement set forth in Section 8 thereof shall have been satisfied or waived in accordance with the provisions of said Section 8.
- 7.3. Other Agreements. At or prior to the Closing, Issuer shall have entered into the Standstill Agreement and Registration Rights Agreement, such agreement being in substantially the form thereof attached as an exhibit to the Purchase Agreement without change other than such changes as may be reasonably satisfactory to Subscriber.
- 7.4. Opinion of Counsel. Issuer shall have furnished Subscriber with favorable opinions of Ropes & Gray and Jay Meltzer, Senior Vice President, General Counsel and Secretary of Issuer, each dated the Closing Date in substantially the forms of Exhibits D-1 and D-2 to the Purchase Agreement.

8. MISCELLANEOUS.

- 8.1. Entire Agreement; Waivers. This Agreement, the other Closing Agreements, the Confidentiality Agreement and the Purchase Agreement constitute the entire agreement among the parties hereto pertaining to the subject matter hereof and thereof and supersede all prior and contemporaneous agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to such subject matter. No waiver of any provision of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), shall constitute a continuing waiver unless otherwise expressly provided nor shall be effective unless in writing and executed (i) in the case of a waiver by Issuer, by Issuer and (ii) in the case of a waiver by Subscriber,
- 8.2. Amendment or Modification. The parties hereto may not amend or modify this Agreement except in such manner as may be agreed upon by a written instrument executed by Issuer and Subscriber.
- 8.3. Survival. All representations, warranties, covenants and agreements made by or on behalf of any party hereto in this Agreement or pursuant to any document, certificate or other instrument referred to herein or delivered in connection with the transactions contemplated hereby shall survive for the General Survival Period as specified in Section 10 of the Purchase Agreement except for the representations and warranties in Sections 3.1.2 and 3.1.3(b) (but only insofar as Section 3.1.3(b) relates to the General Corporation Law of the State of Delaware).
- 8.4. Knowledge. Whenever reference is made in this Agreement to the knowledge of any Person with respect to any matter, it is understood that such knowledge extends only to the officers of such Person (and in the case of Subscriber, the officers of the Company or any

of its Subsidiaries) having responsibility for the areas of such Person's (and in the case of Subscriber, also the Company's or any of its Subsidiaries') business covering such matter, which officers have made an inquiry that is reasonably appropriate to determine the accuracy of the statement in question.

- $8.5.\,$ Successors and Assigns. All the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective transferees, successors and permitted assigns (each of which such transferees, successors and assigns shall be deemed to be a party hereto for all purposes hereof); provided, however, that (i) neither Issuer nor Subscriber may assign or transfer (by operation of law or otherwise) any of its rights or obligations hereunder without the prior written consent of the other and (iii) no transfer or assignment by any party shall relieve such party of any of its obligations hereunder.
- 8.6. Notices. Any notices or other communications required or permitted hereunder shall be sufficiently given if in writing (including telecopy or similar teletransmission), addressed as follows:

If to Subscriber to it at: Melville Corporation

One Theall Road Rye, New York 10580 Telecopier: 914-925-4052

Attention: Chief Executive Officer, Chief Financial

Officer and General Counsel

With a copy to: Davis Polk & Wardwell

450 Lexington Avenue New York, New York 10017 Telecopier: 212-450-5744 Attention: Dennis S. Hersch

If to Issuer to it at: The TJX Companies, Inc.

770 Cochituate Road Framingham, MA 01701 Telecopier: 508-390-2457

Attn: President and General Counsel

With a copy to: Ropes & Gray

One International Place Boston, MA 02110

Telecopier: 617-951-7050 Attention: Arthur G. Siler, Esq.

Unless otherwise specified herein, such notices or other communications shall be deemed received (a) in the case of any notice or communication sent other than by mail, on the date

actually delivered to such address (evidenced, in the case of delivery by overnight courier, by confirmation of delivery from the overnight courier service making such delivery, and in the case of a telecopy, by receipt of a transmission confirmation form or the addressee's confirmation of receipt), or (b) in the case of any notice or communication sent by mail, three Business Days after being sent, if sent by registered or certified mail, with first-class postage prepaid. Each of the parties hereto shall be entitled to specify a different address by giving notice as aforesaid to each of the other parties hereto.

- 8.7. Headings, etc. Section and subsection headings are not to be considered part of this Agreement, are included solely for convenience, are not intended to be full or accurate descriptions of the content thereof and shall not affect the construction hereof.
- 8.8. Third-Party Beneficiaries. Nothing in this Agreement is intended or shall be construed to entitle any Person other than the parties or their respective permitted transferees, successors and assigns hereby to any claim, cause of action, remedy or right of any kind.
- 8.9. Preparation for Closing. Each party will use its reasonable best efforts to bring about the fulfillment of each of the conditions precedent to the obligations of the other parties hereto set forth in this Agreement.
- 8.10. Counterparts. This Agreement may be executed in any number of counterparts and by the different parties on separate counterparts each of which shall be deemed an original, but all of which together shall constitute but one and the same instrument.
- 8.11. Governing Law. This Agreement shall be governed by and construed in accordance with the domestic substantive law of the State of New York, without giving effect to any choice or conflict of law provision or rule that would cause the application of the law of any other jurisdiction.
- 8.12. Termination. This Agreement may be terminated by Issuer and Subscriber by mutual written consent at any time prior to the Closing, and this Agreement shall automatically terminate immediately upon the termination of the Purchase Agreement in accordance with its terms.

hereby, have caused this Preferred	es hereto, intending to be legally bound Stock Subscription Agreement to be executed, by their respective officers thereunto duly
SUBSCRIBER:	MELVILLE CORPORATION
	By Name: Title:
ISSUER:	THE TJX COMPANIES, INC.
	By Name: Title:

-16-

CERTIFICATE OF DESIGNATIONS, PREFERENCES AND RIGHTS OF SERIES D CUMULATIVE CONVERTIBLE PREFERRED STOCK

\$1.00 PAR VALUE PER SHARE

0F

THE TJX COMPANIES, INC.

Pursuant to Section 151(g) of the General Corporation Law of the State of Delaware

We, Steven R. Wishner, Vice President - Finance, and Jay H. Meltzer, Secretary, of The TJX Companies, Inc. (hereinafter called the "Corporation"), a corporation organized and existing under and by virtue of the provisions of the General Corporation Law of the State of Delaware,

Do HEREBY CERTIFY:

FIRST: The restated certificate of incorporation, as amended (the "Certificate of Incorporation"), of the corporation authorizes the issuance of 5,000,000 shares of Preferred Stock, \$1.00 par value per share ("Preferred Stock"), in one or more series, and further authorizes the Board of Directors from time to time to provide by resolution for the issuance of shares of Preferred Stock in one or more series not exceeding the aggregate number of shares of Preferred Stock authorized by the Certificate of Incorporation and to determine with respect to each such series, the voting powers, if any (which voting powers if granted may be full or limited), designations, preferences, the relative, participating, optional or other rights, and the qualifications, limitations and restrictions appertaining thereto.

RESOLVED: that the Finance [Executive] Committee of the Board of Directors, pursuant to authority conferred on such Committee by the Board of Directors (which fixed the voting rights with respect to the shares designated herein) by the provisions of the Second Restated Certificate of Incorporation, as amended (the "Certificate of Incorporation"), of the Corporation, hereby authorizes the issuance of a series of cumulative convertible Preferred Stock of the Corporation and hereby fixes the voting powers, designations, preferences, the relative, participating, optional and other rights, and the qualifications, limitations and restrictions appertaining thereto in addition to those set forth in said Certificate of Incorporation, as follows:

1. DESIGNATION AND NUMBER. The designation of Preferred Stock created by this resolution shall be Series D Cumulative Convertible Preferred Stock, \$1.00 par value per share, of The TJX Companies, Inc. (the "Corporation") (hereinafter referred to as the "Series D Preferred Stock"), and the number of shares constituting such series shall be 250,000, which number may not be increased but may be decreased (but not below the number of shares of Series D Preferred Stock then outstanding) from time to time by the Board of Directors.

All shares of Series D Preferred Stock which shall have been issued and reacquired in any manner by the Corporation (excluding, until the Corporation elects to retire them, shares which are held as treasury shares but including shares redeemed, shares purchased and retired, shares converted pursuant to Section 5 hereof and shares exchanged for any other security of the Corporation) shall not be reissued and shall, upon the making of any necessary filing with the Secretary of State of Delaware have the status of authorized but unissued shares of the Corporation's Preferred Stock, without designation as to series, and thereafter may be issued, but not as shares of Series D Preferred Stock.

DIVIDEND RIGHTS.

(a) General. The holders of shares of Series D Preferred Stock shall be entitled to receive, in preference to the holders of shares of Common Stock and any other stock ranking as to dividends junior to the Series D Preferred Stock, when and as declared by the Board of Directors, out of funds legally available therefor, cumulative cash dividends, accruing from and after the date of original issuance of the Series D Preferred Stock at an annual rate of

- /1/ per share, and no more, as long as shares of Series D Preferred Stock remain outstanding. Dividends shall be payable quarterly in arrears, on January 1, April 1, July 1 and October 1 in each year commencing on the first of such four dates which follows the date of initial issuance of the Series D Preferred Stock (each, a "Dividend Payment Date"). Each dividend will be payable to holders of record as they appear on the stock register of the Corporation on the record date therefor, not exceeding 60 days nor less than 10 days preceding the payment date thereof, as shall be fixed by the Board of Directors. Dividends in arrears may be declared and paid at any time, without reference to any Dividend Payment Date, to holders of record on such date, not exceeding 60 days preceding the payment date thereof, as may be fixed by the Board of Directors of the Corporation. Dividends payable on the Series D Preferred Stock (i) for any period greater or less than a full dividend period, shall be computed on the basis of a 360-day year consisting of twelve 30-day months and (ii) for each full quarterly dividend period, shall be computed by dividing the annual dividend rate by four. Dividends on shares of Series D Preferred Stock shall be cumulative and shall accrue on a daily basis from the date of original issuance thereof whether or not there shall be funds legally available for the payment thereof and whether or not such dividends are declared. Holders of shares of the Series D Preferred Stock shall not be entitled to any dividend, whether payable in cash, property or stock, in excess of Full Cumulative Dividends on such shares. No interest or sum of money in lieu of interest shall be payable in respect of any dividend payment or payments which may be in arrears.
- (b) Requirements for Dividends on Senior Preferred Stock. The Corporation shall not (i) declare or pay or set apart for payment any dividends or distributions on shares of Series D Preferred Stock (other than dividends paid in shares of stock ranking junior to any series of Preferred Stock ranking senior to the Series D Preferred Stock as to dividends) or (ii) make any purchase or redemption of, or any sinking fund payment for the purchase or redemption of, shares of Series D Preferred Stock (other than a purchase or redemption made by issue or delivery of any stock ranking junior to any series of Preferred Stock ranking senior to the Series D Preferred Stock as to dividends or upon liquidation, dissolution or winding up) unless Full Cumulative Dividends on all outstanding shares of any series of Preferred Stock ranking senior to Series D Preferred Stock through the most recent dividend payment date prior to the date of payment of such dividend or distribution, or effective date of such purchase, redemption or sinking fund payment, shall have been paid in full or declared and a sufficient sum set apart for payment thereof.
- (c) Requirements for Dividends on Parity Preferred Stock. If there shall be outstanding shares of any other class or series of Preferred Stock ranking on a parity with the Series D Preferred Stock as to dividends, no dividends, except as described in the next sentence, shall be declared or paid or set apart for payment on any such other series for any

^{/1/} An amount per \$100 that is equal to (\$0.28 divided by the Initial Common Stock Price) x 100.

period unless Full Cumulative Dividends on the Series D Preferred Stock through the most recent Dividend Payment Date have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof is set apart for such payment. If dividends on the Series D Preferred Stock and on any other series of Preferred Stock ranking on a parity as to dividends with the Series D Preferred Stock are in arrears, all dividends declared upon shares of the Series D Preferred Stock and all dividends declared upon such other series shall be declared pro rata so that the amounts of dividends per share declared on the Series D Preferred Stock and such other series shall in all cases bear to each other the same ratio that Full Cumulative Dividends per share at the time on the shares of Series D Preferred Stock and on such other series bear to each other.

(d) Requirements for Dividends on Junior Stock. The Corporation shall not (i) declare or pay or set apart for payment any dividends or distributions on any stock ranking as to dividends junior to the Series D Preferred Stock (other than dividends paid in shares of stock ranking junior to the Series D Preferred Stock as to dividends) or (ii) make any purchase or redemption of, or any sinking fund payment for the purchase or redemption of, any stock ranking as to dividends or upon liquidation, dissolution or winding up junior to the Series D Preferred Stock (other than a purchase or redemption made by issue or delivery of any stock ranking junior to the Series D Preferred Stock as to dividends or upon liquidation, dissolution or winding up) unless Full Cumulative Dividends on all outstanding shares of Series D Preferred Stock through the most recent Dividend Payment Date prior to the date of payment of such dividend or distribution, or effective date of such purchase, redemption or sinking fund payment, shall have been paid in full or declared and a sufficient sum set apart for payment thereof; provided, however, that unless prohibited by the terms of any other outstanding series of Preferred Stock, any moneys theretofore deposited in any sinking fund with respect to any Preferred Stock of the Corporation in compliance with this Section 2(d) and the provisions of such sinking fund may thereafter be applied to the purchase or redemption of such Preferred Stock in accordance with the terms of such sinking fund regardless of whether at the time of such application Full Cumulative Dividends on all outstanding shares of Series D Preferred Stock through the most recent Dividend Payment Date shall have been paid in full or declared and a sufficient sum set apart for payment thereof.

LIQUIDATION PREFERENCES.

(a) Senior Preferred Stock. In the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, before any payment or distribution of the assets of the Corporation (whether from capital or surplus) shall be made to or set apart for the holders of the Series D Preferred Stock upon liquidation, dissolution or winding up, the holders of each class or series of Preferred Stock ranking senior to the Series D Preferred Stock upon liquidation, dissolution or winding up shall be entitled to receive full payment of their liquidation preferences.

Δ-4

(b) Order of Payments among Parity Preferred Stock. In the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, before any payment or distribution of the assets of the Corporation (whether from capital or surplus) shall be made to or set apart for the holders of any class or series of stock of the Corporation ranking junior to the Series D Preferred Stock upon liquidation, dissolution or winding up, the holders of the shares of Series D Preferred Stock and the holders of each other class or series of Preferred Stock ranking on a parity with Series D Preferred Stock upon liquidation, dissolution or winding up shall be entitled to receive liquidation payments according to the following priorities:

First,

The holders of the shares of Series D Preferred Stock shall receive \$100 per share and the holders of shares of each such other class or series of Preferred Stock shall receive the full respective liquidation preferences (including any premiums) to which they are entitled; and

Second,

The holders of shares of Series D Preferred Stock and the holders of shares of each such other class or series of Preferred Stock shall each receive an amount equal to Full Cumulative Dividends with respect to their respective shares through and including the date of final distribution to such holders, but such holders shall not be entitled to any further payment.

No payment (in either of the First step or Second step provided above) on account of any liquidation, dissolution or winding up of the Corporation shall be made to holders of any such other class or series of Preferred Stock or to the holders of Series D Preferred Stock unless there shall likewise be paid at the same time to the holders of the Series D Preferred Stock and the holders of each such other class or series of Preferred Stock like proportionate amounts of the same payments (as to each of the First step or the Second step above), such proportionate amounts to be determined ratably in proportion to the full amounts to which the holders of all outstanding shares of Series D Preferred Stock and the holders of all outstanding shares of each such other class or series of Preferred Stock are respectively entitled (in either the First step or the Second step, as the case may be) with respect to such distribution.

For purposes of this Section 3, neither a consolidation or merger of the Corporation with or into another corporation nor a merger of any other corporation with or into the Corporation or a sale or transfer of all or any part of the Corporation's assets for cash, securities or other property will be deemed a liquidation, dissolution or winding up of the Corporation.

(c) Junior Stock. After payment shall have been made in full to the holders of Series D Preferred Stock and to the holders of each such other class or series of Preferred Stock as provided in this Section 3 upon liquidation, dissolution or winding up of the Corporation, any other series or class or classes of stock ranking junior to the Series D Preferred Stock upon

liquidation, dissolution or winding up shall, subject to the respective terms and provisions (if any) applying thereto, be entitled to receive any and all assets remaining to be paid or distributed upon such liquidation, dissolution or winding up, and the holders of Series D Preferred Stock shall not be entitled to share therein.

4. MANDATORY REDEMPTION BY THE CORPORATION.

- (a) Obligation to Redeem Out of Asset Sale or Stock Sale Proceeds. Except as provided by this Section 4, shares of Series D Preferred are not redeemable by the Corporation. If at any time after the Initial Issuance Date and not less than 10 Business Days before the Automatic Conversion Date the Corporation shall consummate any Sale, then the Corporation shall apply the full amount of the Sale Proceeds received by the Corporation in respect of such Sale to redeem all then outstanding shares of Series D Preferred Stock (or, if fewer, as many such shares as can be redeemed at the Call Price out of such Sale Proceeds). Upon any such redemption, the Corporation shall deliver to the holders of shares of Series D Preferred Stock, in accordance with the provisions of this Certificate, in exchange for each share so redeemed, cash in an amount equal to the sum of (i) the Call Price in effect on the date of redemption plus (ii) Full Cumulative Dividends thereon to the date fixed for redemption. If fewer than all the outstanding shares of Series D Preferred Stock are to be redeemed, shares to be redeemed shall be selected by the Corporation from outstanding shares of this Series not previously redeemed by lot or pro rata (as nearly as may be practicable) or by any other method determined by the Board of Directors of the Corporation in its sole discretion to be equitable.
- (b) Notice of Redemption. The Corporation will provide notice of any redemption of shares of Series D Preferred Stock to holders of record of the Series D Preferred Stock to be redeemed not less than 10 nor more than 60 days prior to the date fixed for such redemption. Such notice shall be provided by first-class mail postage prepaid, to each holder of record of the Series D Preferred Stock to be redeemed, at such holder's address as it appears on the stock register of the Corporation; provided, however, that failure to give such notice or any defect therein shall not affect the validity of the proceeding for redemption of any of the shares of Series D Preferred Stock. Each such mailed notice shall state, as appropriate, the following:
- (i) the redemption date (which shall be no later than the Automatic Conversion Date);
 - (ii) the amount of the Sale Proceeds;
- (iii) the number of shares of Series D Preferred Stock to be redeemed and, if less than all the shares held by any holder are to be redeemed, the number of such shares to be redeemed from such holder;

- (iv) the Call Price;
- (v) the place or places where certificates for such shares are to be surrendered for redemption;
- (vi) the amount of Full Cumulative Dividends per share of Series D Preferred Stock to be redeemed through and including such redemption date, and that dividends on shares of Series D Preferred Stock to be redeemed will cease to accrue on such redemption date unless the Corporation shall default in payment of the Call Price plus such Full Cumulative Dividends thereon;
- (vii) the Exchange Rate as of the date of such notice (it being understood that the Exchange Rate may thereafter fluctuate in accordance with the terms set forth in the definition thereof), and that the right of holders to convert shares of Series D Preferred Stock to be redeemed will terminate at the close of business on the Business Day next preceding the date fixed for redemption (unless the Corporation shall default in the payment of the Call Price plus such Full Cumulative Dividends thereon).
 - (c) Mechanics of Redemption.
- (i) Upon surrender in accordance with the aforesaid notice of the certificate for any shares so redeemed (duly endorsed or accompanied by appropriate instruments of transfer), the holders of record of such shares shall be entitled to receive an amount of cash constituting the Call Price plus Full Cumulative Dividends thereon, without interest.
- The Corporation's obligation to provide funds upon redemption in accordance with this Section 4 shall be deemed fulfilled if, on or before a redemption date, the Corporation shall deposit with a bank or trust company, an affiliate of a bank or trust company, having an office or agency in New York, New York and having a capital and surplus of at least \$50,000,000 according to its last published statement of condition, or shall set aside or make other reasonable provision for the payment of cash required to be delivered by the Corporation pursuant to this Section 4 upon the occurrence of the related redemption of Series D Preferred Stock and for cash required to pay Full Cumulative Dividends and cash in lieu of fractional shares on the shares of Series D Preferred Stock to be redeemed as required by this Section 4, in trust for the account of the holders of such shares of Series D Preferred Stock to be redeemed (and so as to be and continue to be available therefor), with (in the case of deposits with a bank or trust company) irrevocable instructions and authority to such bank or trust company that such funds be delivered upon redemption of the shares of Series D Preferred Stock so called for redemption. If such notice of redemption shall have been given, and if on the date fixed for redemption funds necessary for the redemption shall have been irrevocably either set aside by the Company separate and apart from its other funds or assets in trust for the account of the holders of the shares of Series D Preferred Stock to be redeemed (and so as to be and continue to be available therefor) or the Company shall have made other

reasonable provision therefor, then, notwithstanding that the certificates evidencing any shares of the Series D Preferred Stock so called for redemption shall not have been surrendered, the shares represented thereby shall be deemed no longer outstanding, dividends with respect to such shares shall cease to accrue on the date fixed for redemption (provided that holders of shares of Series D Preferred Stock at the close of business on a record date for any payment of dividends shall be entitled to receive Full Cumulative Dividends payable on such shares on the corresponding Dividend Payment Date notwithstanding the redemption of such shares following such record date and prior to such Dividend Payment Date) and all rights with respect to such shares shall forthwith after such date cease and terminate, except for the rights of the holders to receive the funds payable pursuant to this Section 4 without interest upon surrender of their certificates therefor.

(iii) Each redemption of shares of Series D Preferred Stock pursuant to this Section 4 shall be deemed to have been made as of the close of business on the applicable redemption date, so that the rights of the holder of such shares of Series D Preferred Stock shall, to the extent of such redemption, cease at such time.

CONVERSION.

- (a) Automatic Conversion. Unless earlier converted pursuant to Section 5(b) at the option of the holder, on the Automatic Conversion Date each outstanding share of the Series D Preferred Stock shall convert automatically (the "Automatic Conversion") into (i) shares of Common Stock at the Exchange Rate in effect on the Automatic Conversion Date and (ii) the right to receive an amount in cash equal to Full Cumulative Dividends on such share to the Automatic Conversion Date.
- (b) Optional Conversion by Holder. Shares of Series D Preferred Stock may be converted, in whole or in part, at the option of the holder thereof ("Optional Conversion"), at any time after the giving of any notice of redemption by the Corporation pursuant to Section 4 and not later than the close of business on the Business Day prior to the Automatic Conversion Date, into (i) shares of Common Stock at the Exchange Rate in effect on the Optional Conversion Date and (ii) the right to receive an amount in cash equal to Full Cumulative Dividends on such share to the Optional Conversion Date; provided that only the shares of Series D Preferred Stock that were subject to such notice of redemption may be converted in an Optional Conversion. Notwithstanding the foregoing, the Corporation may, at its option, in lieu of delivering shares of Common Stock on the Optional Conversion Date, deliver cash in an aggregate amount equal to the aggregate Closing Price (on the Trading Day preceding the Optional Conversion Date) of the number of shares of Common Stock otherwise so deliverable (together, in any event, with Full Cumulative Dividends thereon to the Optional Conversion Date).

Optional Conversion of shares of Series D Preferred Stock may be effected by delivering certificates evidencing such shares, together with written notice of conversion and a $\,$

proper assignment of such certificates to the Corporation or in blank (and, if applicable, payment of an amount equal to the dividend payable on such shares), to the office of any transfer agent for the Series D Preferred Stock or to any other office or agency maintained by the Corporation for that purpose and otherwise in accordance with the Optional Conversion procedures established by the Corporation. Each Optional Conversion shall be deemed to have been effected immediately prior to the close of business on the date on which the foregoing requirements shall have been satisfied (the "Optional Conversion Date").

If any shares of Series D Preferred Stock shall be called for redemption, the right to convert the shares designated for redemption shall terminate at the close of business on the Business Day preceding the date fixed for redemption.

- (c) Mechanics of Conversion.
- (i) Upon surrender in accordance with the aforesaid provisions of the certificate for any shares so converted (duly endorsed or accompanied by appropriate instruments of transfer), the holder of record of such shares shall be entitled to receive the applicable number of shares of Common Stock (calculated to the nearest 1/1,000,000th of a share) (and cash representing fractional share settlements in respect thereof) at the applicable Exchange Rate plus Full Cumulative Dividends thereon, without interest.
- Before any holder of shares of Series D Preferred Stock shall receive certificates for shares of Common Stock in respect of the conversion of shares of Series D Preferred Stock (or cash representing fractional share settlements in respect thereof) such holder shall surrender the certificate or certificates of shares of Series D Preferred Stock duly endorsed if required by the Corporation, at the office of the Corporation and, if certificates for shares of Common Stock are to be received by such holder, shall state in writing the name or names and the denominations in which such holder wishes the certificate or certificates for the Common Stock to be issued. The Corporation $% \left(1\right) =\left(1\right) \left(1\right) \left($ will, as soon as practicable after receipt thereof, issue and deliver to such holder, or such holder's designee or designees, a certificate or certificates for the number of shares of Common Stock to which such holder shall be entitled as aforesaid, together with a certificate or certificates representing any shares of Series D Preferred Stock which are not to be converted, but which shall have constituted part of the certificate or certificates for shares of Series D Preferred Stock so surrendered.
- (iii) The Corporation's obligation to deliver shares of Common Stock and provide funds upon conversion in accordance with this Section 5 shall be deemed fulfilled if, on or before a conversion date, the Corporation shall deposit with a bank or trust company, or an affiliate of a bank or trust company, having an office or agency in New York, New York and having a capital and surplus of at least \$50,000,000 according to its last published statement of condition, or shall set aside or make other reasonable provision for the issuance of, such number of shares of Common Stock as are required to be delivered by the Corporation pursuant to this Section 5 upon the occurrence of the related conversion of Series D Preferred

Stock (and for the payment of cash required to be delivered by the Corporation pursuant to this Section 5 if the Corportion so elects in the case of an Optional Conversion of Series D Preferred Stock) and for cash required to be paid in lieu of the issuance of fractional share amounts and to pay Full Cumulative Dividends on the shares of Series D Preferred Stock to be converted as required by this Section 5, in trust for the account of the holders of such shares of Series D Preferred Stock to be converted (and so as to be and continue to be available therefor), with (in the case of deposits with a bank or trust company) irrevocable instructions and authority to such bank or trust company that such shares and funds be delivered upon conversion of the shares of Series D Preferred Stock so to be converted. If on the Automatic Conversion Date shares of Common Stock and funds (if any) necessary for the conversion shall have been irrevocably either set aside by the Company separate and apart from its other funds or assets in trust for the account of the holders of the shares of Series D Preferred Stock to be converted (and so as to be and continue to be available therefor) or the Company shall have made other reasonable provision therefor, then, notwithstanding that the certificates evidencing any shares of the Series D Preferred Stock so subject to conversion shall not have been surrendered, the shares represented thereby shall be deemed no longer outstanding, dividends with respect to such shares shall cease to accrue on the date fixed for conversion (provided that holders of shares of Series D Preferred Stock at the close of business on a record date for any payment of dividends shall be entitled to receive the Full Cumulative Dividend payable on such shares on the corresponding Dividend Payment Date notwithstanding the conversion of such shares following such record date and prior to such Dividend Payment Date) and all rights with respect to such shares shall forthwith after such date cease and terminate, except for the rights of the holders to receive the shares of Common Stock and funds (if any) payable pursuant to this Section 5 without interest upon surrender of their certificates therefor. Holders of shares of Series D Preferred Stock at the close of business on a dividend payment record date shall be entitled to receive the dividend payable on such shares on the corresponding Dividend Payment Date notwithstanding the Optional Conversion of such shares following such record date and prior to such Dividend Payment Date. However, shares of Series D Preferred Stock surrendered for Optional Conversion after the close of business on a dividend payment record date and before the opening of business on the corresponding Dividend Payment Date (except shares converted after the issuance of a notice of redemption with respect to a redemption date during such period) must be accompanied by payment in cash of an amount equal to the dividend payable on such shares on such Dividend Payment Date. A holder of shares of Series D Preferred Stock on a dividend record date who (or whose transferee) surrenders any such shares for conversion into shares of Common Stock on the corresponding Dividend Payment Date will receive the dividend payable by the Corporation on such shares of Series D Preferred Stock on such Dividend Payment Date, and the converting holder need not include payment of the amount of such dividend upon surrender of shares of Series D Preferred Stock for conversion. Except as provided above, upon any conversion of shares of Series D Preferred Stock, the Corporation shall make no payment or allowance for unpaid dividends, whether or not in arrears, on such shares of Series D Preferred Stock as to which conversion has been effected or for dividends or distributions on the shares of Common Stock issued upon such conversion.

A-10

- (iv) Holders of shares of Series D Preferred Stock that are converted shall not be entitled to receive dividends declared and paid on such shares of Common Stock, and such shares of Common Stock shall not be entitled to vote, until such shares of Common Stock are issued upon the surrender of the certificates representing such shares of Series D Preferred Stock and upon such surrender such holders shall be entitled to receive such dividends declared and paid on such shares of Common Stock subsequent to such conversion date. Amounts payable in cash in respect of the shares of Series D Preferred Stock or in respect of such shares of Common Stock shall not bear interest.
- (v) Each conversion of shares of Series D Preferred Stock into Common Stock shall be deemed to have been made as of the close of business on the applicable conversion date, so that the rights of the holder of such shares of Series D Preferred Stock shall, to the extent of such conversion, cease at such time and the person or persons entitled to receive shares of the Common Stock upon conversion of such shares shall be treated for all purposes as having become the record holder or holders of the Common Stock at such time; provided, however, that if an event that results in an adjustment to the Exchange Rate is declared or occurs with respect to the shares of Common Stock, and the record date for any such action is on or after the close of business on the date on which notice of such conversion is given, but prior to the close of business on the date of such conversion, then the person or persons entitled to receive shares of the Common Stock upon conversion of shares of Series D Preferred Stock shall be treated for purposes of such action as having become the record holder or holders of the Common Stock at the close of business on the Trading Day next preceding the date on which notice of such conversion is given.
- (vi) The Corporation will pay any and all taxes that may be payable in respect of the issuance or delivery of shares of Common Stock upon conversion of shares of Series D Preferred Stock pursuant hereto. The Corporation shall not, however, be required to pay any tax which may be payable in respect of any transfer involved in the delivery of shares registered in a name other than the name in which such shares of Series D Preferred Stock were formerly registered, and no such issue or delivery shall be made unless and until the person requesting such issue or delivery has paid to the Corporation the amount of any such tax, or has established, to the satisfaction of the Corporation, that such tax has been paid.
- (d) Adjustments to the Exchange Rate. The Exchange Rate shall be subject to adjustment from time to time as provided below in this paragraph (d).
 - (i) If the Corporation shall pay or make a dividend or other distribution with respect to its Common Stock in shares of Common Stock (including by way of reclassification of any shares of its Common Stock) to all holders of Common Stock, the Exchange Rate in effect at the opening of business on the day following the date fixed for the determination of stockholders entitled to receive such dividend or other distribution shall be increased by multiplying such Exchange Rate by a fraction of

which the numerator shall be the sum of the number of shares of Common Stock outstanding at the close of business on the date fixed for such determination plus the total number of shares of Common Stock constituting such dividend or other distribution, and of which the denominator shall be the number of shares of Common Stock outstanding at the close of business on the date fixed for such determination, such increase to become effective immediately after the opening of business on the day following the date fixed for such determination.

- (ii) In case outstanding shares of Common Stock shall be subdivided into a greater number of shares of Common Stock, the Exchange Rate in effect at the opening of business on the day following the day upon which such subdivision becomes effective shall be proportionately increased, and, conversely, in case outstanding shares of Common Stock shall be combined into a smaller number of shares of Common Stock, the Exchange Rate in effect at the opening of business on the day following the day upon which such combination becomes effective shall be proportionately reduced, such increases or reductions, as the case may be, to become effective immediately after the opening of business on the day following the day upon which such subdivision or combination becomes effective.
- (iii) If the Corporation shall, after the date hereof, issue rights or warrants, in each case other than the Rights, to all holders of its Common Stock entitling them (for a period not exceeding 45 days from the date of such issuance) to subscribe for or purchase shares of Common Stock at a price per share less than the Fair Market Value of the Common Stock on the record date for the determination of stockholders entitled to receive such rights or warrants, then in each case the Exchange Rate shall be adjusted by multiplying the Exchange Rate in effect on such record date, by a fraction of which the numerator shall be the number of shares of Common Stock outstanding on the date of issuance of such rights or warrants, immediately prior to such issuance, plus the number of additional shares of Common Stock offered for subscription or purchase pursuant to such rights or warrants, and of which the denominator shall be the number of shares of Common Stock outstanding on the date of issuance of such rights or warrants, immediately prior to such issuance, plus the number of shares of Common Stock which the aggregate offering price of the total number of shares of Common Stock so offered for subscription or purchase pursuant to such rights or warrants would purchase at such Fair Market Value (determined by multiplying such total number of shares by the exercise price of such rights or warrants and dividing the product so obtained by such Fair Market Value). Such adjustment shall become effective at the opening of business on the Business Day next following the record date for the determination of stockholders entitled to receive such rights or warrants. To the extent that shares of Common Stock are not delivered after the expiration of such rights or warrants, the Exchange Rate shall be readjusted to the Exchange Rate which would then be in effect had the adjustments made upon the issuance of such rights or warrants been made upon the basis of the issuance of rights or warrants in respect of only

A-12

- If the Corporation shall pay a dividend or make a distribution to all holders of its Common Stock consisting of evidences of its indebtedness or other assets (including shares of capital stock of the Corporation other than Common Stock but excluding any cash dividends or any dividends or other distributions referred to in clauses (i) and (ii) above), or shall issue to all holders of its Common Stock rights or warrants to subscribe for or purchase any of its securities (other than those referred to in clause (iii) above and other than Rights), then in each such case the Exchange Rate shall be adjusted by multiplying the Exchange Rate in effect on the record date for such dividend or distribution or for the determination of stockholders entitled to receive such rights or warrants, as the case may be, by a fraction of which the numerator shall be the Fair Market Value per share of the Common Stock on such record date, and of which the denominator shall be such Fair Market Value per share of Common Stock less the fair market value (as determined by the Board of Directors, whose determination shall be conclusive) as of such record date of the portion of the assets or evidences of indebtedness so distributed, or of such subscription rights or warrants, applicable to one share of Common Stock. Such adjustment shall become effective on the opening of business on the Business Day next following the record date for such dividend or distribution or for the determination of stockholders entitled to receive such rights or warrants, as the case may be.
- (v) Any share of Common Stock issuable in payment of a dividend or other distribution shall be deemed to have been issued immediately prior to the close of business on the record date for such dividend or other distribution for purposes of calculating the number of outstanding shares of Common Stock under subparagraph (ii) above.
- (vi) Anything in this paragraph (d) notwithstanding, the Corporation shall be entitled to make such upward adjustments in the Exchange Rate, in addition to those required by this paragraph (d), as the Corporation in its sole discretion shall determine to be advisable, in order that any stock dividends, subdivision of shares, distribution of rights to purchase stock or securities, or distribution of securities convertible into or exchangeable for stock (or any transaction which could be treated as any of the foregoing transactions pursuant to Section 305 of the Internal Revenue Code of 1986, as amended) hereafter made by the Corporation to its stockholders shall not be taxable.
- (vii) In any case in which this paragraph (d) shall require that an adjustment as a result of any event become effective at the opening of business on the Business Day next following a record date and the date fixed for conversion pursuant to paragraph (a) occurs after such record date, but before the occurrence of such event, the Corporation may in its sole discretion elect to defer the following until after the occurrence of such event: (A) issuing to the holder of any shares of Series D Preferred Stock surrendered

for conversion the additional shares of Common Stock issuable upon such conversion over the shares of Common Stock issuable before giving effect to such adjustment; and (B) paying to such holder any amount in cash in lieu of a fractional share of Common Stock pursuant to Section 6(d).

- (viii) For purposes hereof, an "adjustment in the Exchange Rate" means, and shall be implemented by, an adjustment of the nature and amount specified, effected in the manner specified, in each of the Upper Exchange Rate, the Middle Exchange Rate and the Lower Exchange Rate. If an adjustment is made to the Exchange Rate pursuant to this paragraph (d), a proportionate adjustment in the same direction shall also be made on the Automatic Conversion Date to the Current Market Price solely to determine which of clauses (a), (b) or (c) of the definition of Exchange Rate will apply on the Automatic Conversion Date. Such adjustment shall be made by multiplying the Current Market Price by a fraction of which the numerator shall be the Exchange Rate immediately after such adjustment pursuant to this paragraph (d) and the denominator shall be the Exchange Rate immediately before such adjustment. All adjustments to the Exchange Rate shall be calculated to the nearest 1/1,000,000th of a share of Common Stock. No adjustment in the Exchange Rate shall be required unless such adjustment would require an increase or decrease of at least one percent in the Exchange Rate; provided, however, that any adjustments which by reason of this subparagraph are not required to be made shall be carried forward and taken into account in any subsequent adjustment. All adjustments to the Exchange Rate shall be made successively.
- (ix) Before taking any action that would cause an adjustment increasing the Exchange Rate such that the conversion price (for purposes of this paragraph (d), an amount equal to the liquidation value per share of Series D Preferred Stock divided by the Upper Exchange Rate as in effect from time to time) would be below the then par value of the Common Stock, the Corporation will take any corporate action which may, in the opinion of its counsel, be necessary in order that the Corporation may validly and legally issue fully paid and nonassessable shares of Common Stock at the Upper Exchange Rate as so adjusted.
- (e) Adjustment for Certain Consolidations or Mergers. In case of any consolidation or merger to which the Corporation is a party (other than a merger or consolidation in which the Corporation is the continuing corporation and in which the Common Stock outstanding immediately prior to the merger or consolidation remains unchanged), or in case of any sale or transfer to another corporation of the property of the Corporation as an entirety or substantially as an entirety, or in case of any statutory exchange of securities with another corporation (other than in connection with a merger or acquisition), proper provision shall be made so that each share of the Series D Preferred Stock shall, after consummation of such transaction, be subject to (i) conversion at the option of the holder into the kind and amount of securities, cash or other property receivable upon consummation of such transaction by a holder of the number of shares of Common Stock into which such share of Series D Preferred

Stock would have been converted if the conversion had occurred immediately prior to consummation of such transaction (based on the Exchange Rate in effect immediately prior to such consummation), (ii) conversion on the Automatic Conversion Date into the kind and amount of securities, cash or other property receivable upon consummation of such transaction by a holder of the number of shares of Common Stock into which such share of Series D Preferred Stock would have been converted if the conversion on the Automatic Conversion Date had occurred immediately prior to the date of consummation of such transaction (based on the Exchange Rate in effect immediately prior to such consummation) and (iii) redemption on any redemption date in exchange for the kind and amount of securities, cash or other property receivable upon consummation of such transaction by a holder of the number of shares of Common Stock that would have been issuable at the Call Price in effect on such redemption date upon a redemption of such share of Series D Preferred Stock immediately prior to consummation of such transaction; assuming in each case that such holder of Common Stock failed to exercise rights of election, if any, as to the kind or amount of securities, cash or other property receivable upon consummation of such transaction (provided that if the kind or amount of securities, cash or other property receivable upon consummation of such transaction is not the same for each nonelecting share, then the kind and amount of securities, cash or other property receivable upon consummation of such transaction for each nonelecting share shall be deemed to be the kind and amount so receivable per share by a plurality of the nonelecting shares). The kind and amount of securities into which the shares of the Series D Preferred Stock shall be convertible after consummation of such transaction shall be subject to adjustment as described in paragraph (d) following the date of consummation of such transaction. The Corporation may not become a party to any such transaction unless the terms thereof are consistent with the foregoing.

- (f) Notice of Adjustments. Whenever the Exchange Rate is adjusted as provided in paragraph (d), the Corporation shall:
 - (i) Forthwith compute the adjusted Exchange Rate and prepare a certificate signed by the Chief Financial Officer, any Vice President, the Treasurer or the Controller of the Corporation setting forth the adjusted Exchange Rate, the method of calculation thereof in reasonable detail and the facts requiring such adjustment and upon which such adjustment is based, which certificate shall be prima facie evidence of the correctness of the adjustment, and file such certificate forthwith with the Transfer Agent;
 - (ii) Make a prompt public announcement stating that the Exchange Rate has been adjusted and setting forth the adjusted Exchange Rate; and
 - (iii) Promptly mail a notice (stating that the Exchange Rate has been adjusted and the facts requiring such adjustment and upon which such adjustment is based and setting forth the adjusted Exchange Rate) to the holders of record of the outstanding shares of the Series D Preferred Stock at or prior to the time the Corporation mails an interim statement to its stockholders covering the fiscal quarter during which the facts

A-15

requiring such adjustment occurred but in any event within 45 days of the end of such fiscal quarter.

- (g) Prior Notice of Certain Events. In case:
- (i) the Corporation shall (1) declare any dividend (or any other distribution) on its Common Stock, other than a dividend payable solely in cash in an amount such that the aggregate cash dividend per share of Common Stock in any fiscal quarter does not exceed 3.75% of the Current Market Price of the Common Stock on the Trading Day next preceding the date of declaration of such dividend, or (2) declare or authorize a redemption or repurchase of in excess of 10% of the then outstanding shares of Common Stock; or
- (ii) the Corporation shall authorize the granting to all holders of Common Stock of rights or warrants to subscribe for or purchase any shares of stock of any class or of any other rights or warrants (other than Rights); or
- (iii) of any reclassification of Common Stock (other than a subdivision or combination of the outstanding Common Stock, or a change in par value, or from par value to no par value, or from no par value to par value), or of any consolidation or merger to which the Corporation is a party and for which approval of any stockholders of the Corporation shall be required, or of the sale or transfer of all or substantially all of the assets of the Corporation or of any compulsory share exchange where the Common Stock is converted into other securities, cash or other property; or
- (iv) of the voluntary or involuntary liquidation, dissolution or winding up of the Corporation;

then the Corporation shall cause to be filed with the Transfer Agent and each office or agency maintained for conversion of shares of Series D Preferred Stock, and shall cause to be mailed to the holders of record of the Series D Preferred Stock, at their last addresses as they shall appear upon the stock transfer books of the Corporation, at least 15 days prior to the applicable record date hereinafter specified, a notice stating (x) the date on which a record (if any) is to be taken for the purpose of such dividend, distribution, redemption, repurchase or granting of rights or warrants or, if a record is not to be taken, the date as of which the holders of Common Stock of record to be entitled to such dividend, distribution, redemption, rights or warrants are to be determined or (y) the date on which such reclassification, consolidation, merger, sale, transfer, share exchange, liquidation, dissolution or winding up is expected to become effective, and the date as of which it is expected that holders of Common Stock of record shall be entitled to exchange their shares of Common Stock for securities or other property (including cash) deliverable upon such reclassification, consolidation, merger, sale, transfer, share exchange, liquidation, dissolution or winding up. No failure to mail such notice or any defect therein or in the mailing thereof shall affect the validity of the corporate action required to be specified in such notice.

- (h) Dividend or Interest Reinvestment Plans; Other. Notwithstanding the foregoing provisions, the issuance of any shares of Common Stock pursuant to any plan providing for the reinvestment of dividends or interest payable on securities of the Corporation and the investment of additional optional amounts in shares of Common Stock under any such plan, and the issuance of any shares of Common Stock or options or rights to purchase such shares pursuant to any employee benefit plan or program of the Corporation, or pursuant to any option, warrant, right or exercisable, exchangeable or convertible security outstanding as of the date the Series D Preferred Stock was first designated, shall not be deemed to constitute an issuance of Common Stock or exercisable, exchangeable or convertible securities by the Corporation to which any of the adjustment provisions described above applies. There shall be no adjustment of the Exchange Rate in case of the issuance of any stock (or securities convertible into or exchangeable for stock) of the corporation except as described in this Section 5. Except as expressly set forth in this Section 5, if any action would require adjustment of the Exchange Rate pursuant to more than one of the provisions described above, only one adjustment shall be made and such adjustment shall be the amount of adjustment which has the highest absolute value.
- (i) For purposes of this Section 5, the number of shares of Common Stock at any time outstanding shall not include any shares of Common Stock then owned or held (directly or indirectly through a subsidiary) by or for the account of the Corporation.
- 6. RESERVATION OF SHARES; LISTING OF SHARES, ETC.
- (a) Reservation of Shares. The Corporation shall at all times reserve and keep available, out of its authorized and unissued stock, solely for the purpose of effecting the conversion of the Series D Preferred Stock, the full number of shares of its Common Stock deliverable upon conversion of all shares of Series D Preferred Stock not theretofore converted.
- (b) Listing of Shares. If any shares of Common Stock required to be reserved for purposes of conversion of the Series D Preferred Stock hereunder require registration with or approval of any governmental authority under any Federal or State law before such shares may be issued upon conversion, the Corporation will in good faith and as expeditiously as possible endeavor to cause such shares to be duly registered or approved, as the case may be. If the Common Stock is listed on the New York Stock Exchange or any other national securities exchange, the Corporation will, as expeditiously as possible, if permitted by the rules of such exchange, cause to be listed and keep listed on such exchange, upon official notice of issuance, all shares of Common Stock issuable upon conversion of the Series D Preferred Stock.
- (c) Shares Issued on Conversion to be Fully Paid, Etc. The shares of Common Stock issuable upon conversion of the shares of Series D Preferred Stock, when the same shall be issued in accordance with the terms hereof, are hereby declared to be and shall be fully paid and nonassessable shares of Common Stock in the hands of the holders thereof.

- (d) No Fractional Shares. No fractional shares or scrip representing fractional shares of Common Stock shall be issued upon conversion of Series D Preferred Stock. Instead of any fractional share of Common Stock that would otherwise be issuable upon conversion of any shares of Series D Preferred Stock, the Corporation shall pay a cash adjustment in respect of such fractional interest in an amount equal to the same fraction of the Closing Price of a share of Common Stock (or, if there is no such Closing Price, the fair market value of a share of Common Stock, as determined or prescribed by the Board of Directors) at the close of business on the Trading Day immediately preceding the date of conversion.
- (e) Other Action. If the Corporation shall take any action affecting the Common Stock, other than action described in Section 5, that in the opinion of the Board of Directors would materially adversely affect the conversion rights of the holders of the shares of Series D Preferred Stock, the Exchange Rate for the Series D Preferred Stock may be adjusted, to the extent permitted by law, in such manner, if any, and at such time, as the Board of Directors may determine to be equitable in the circumstances.
- 7. VOTING RIGHTS. Other than as required by applicable law, the Series D Preferred Stock shall not have any voting powers either general or special except that:
- (a) Unless a greater vote or consent shall then be required by law, the affirmative vote or consent of two-thirds of the votes to which the holders of the outstanding shares of the Series D Preferred Stock, and each other series of Preferred Stock of the Corporation similarly affected, if any, voting together as a single class, are entitled shall be necessary for authorizing, effecting or validating the amendment, alteration or repeal of any of the provisions of the Certificate of Incorporation (including any Certificate of Designations, Preferences and Rights or any similar document relating to any series of Preferred Stock) of the Corporation, including any amendment or supplement thereto, if such would materially and adversely affect the preferences, rights, powers or privileges, qualification, limitations and restrictions of the Series D Preferred Stock and any such other series of Preferred Stock; provided, however, that the creation, issuance or increase in the amount of authorized shares of any series of Preferred Stock ranking on a parity with or junior to the Series D Preferred Stock as to the payment of dividends or upon liquidation, dissolution or winding up will not be deemed to materially and adversely affect such rights, powers or privileges, qualification, limitations and restrictions.
- (b) Unless the vote or consent of the holders of a greater number of shares shall then be required by law, the affirmative vote or consent of two-thirds of the votes to which the holders of the outstanding shares of the Series D Preferred Stock, and all other series of Preferred Stock of the Corporation ranking on parity with shares of the Series D Preferred Stock (either as to dividends or upon liquidation, dissolution or winding up) as to which like voting rights have been conferred, voting together as a single class, are entitled shall be necessary to create, authorize or issue, or reclassify any authorized stock of the Corporation

into, or create, authorize or issue any obligation or security convertible into or evidencing a right to purchase, any shares of any class or series of stock of the Corporation ranking prior to the Series D Preferred Stock or ranking prior to any other class or series of Preferred Stock of the Corporation which ranks on a parity with the Series D Preferred Stock as to dividends or upon liquidation, dissolution or winding up.

- (c) Whenever, at any time or times, dividends payable on the shares of Series D Preferred Stock shall be in arrears in an amount equal to at least six full quarterly dividends on shares of the Series D Preferred Stock at the time outstanding, the holders of the outstanding shares of Series D Preferred Stock shall have the exclusive right, voting together as a class with holders of shares of any one or more other series of Preferred Stock ranking on a parity with the Series D Preferred Stock (either as to dividends or upon liquidation, dissolution or winding up) upon which like voting rights have been conferred and are then exercisable, to elect two (2) directors of the Corporation for one-year terms at the Corporation's next annual meeting of stockholders and at each subsequent annual meeting of stockholders. If the right to elect directors shall have accrued to the holders of the Series D Preferred Stock more than 90 days prior to the date established for the next annual meeting of stockholders, the President of the Corporation shall, within 20 days after delivery to the Corporation at its principal office of a written request for a special meeting signed by the holders of at least 10% of all outstanding shares of the Series D Preferred Stock, call a special meeting of the holders of Series D Preferred Stock to be held within 60 days after the delivery of such request for the purpose of electing such additional directors. Upon the vesting of such right of the holders of Series D Preferred Stock, the maximum authorized number of members of the Board of Directors shall automatically be increased by two and the two vacancies so created shall be filled by vote of the holders of the outstanding shares of Series D Preferred Stock (either alone or together with the holders of shares of any one or more other such series of Preferred Stock entitled to vote in such election) as set forth above. The right of the holders of Series D Preferred Stock to elect members of the Board of Directors of the Corporation as aforesaid shall continue until such time as all dividends in arrears on the Series D Preferred Stock shall have been paid in full or declared and set apart for payment, at which time such right shall terminate, except as herein or by law expressly provided, subject to revesting in the event of each and every subsequent default of the character above described.
- (d) Upon termination of such special voting rights attributable to all holders of the Series D Preferred Stock and any other such series of Preferred Stock ranking on a parity with the Series D Preferred Stock as to dividends or upon liquidation, dissolution or winding up and upon which like voting rights have been conferred and are exercisable, the term of office of each director elected by the holders of shares of Series D Preferred Stock and such parity Preferred Stock (a "Preferred Stock Director") pursuant to such special voting rights shall immediately terminate and the number of directors constituting the entire Board of Directors shall be reduced by the number of Preferred Stock Directors. Any Preferred Stock Director may be removed by, and shall not be removed otherwise than by, a majority of the votes to which the holders of the outstanding shares of Series D Preferred Stock and all other such

series of Preferred Stock ranking on a parity with the Series D Preferred Stock with respect to dividends who were entitled to participate in such Preferred Stock Directors election, voting as a single class, are entitled. If the office of any Preferred Stock Director becomes vacant by reason of death, resignation, retirement, disqualification, removal from office, or otherwise, the remaining Preferred Stock Director may choose a successor who shall hold office for the unexpired term in respect of which such vacancy occurred.

- (e) In connection with any right to vote, each holder of Series D Preferred Stock shall be entitled to one vote for each share held (the holders of shares of any other series of Preferred Stock being entitled to such number of votes, if any, for each share of stock held as may be granted to them).
- 8. RANKING. The Common Stock shall rank junior to the Series D Preferred Stock as to dividends and upon liquidation, dissolution or winding up, as described in Sections 2 and 3. The Series A Preferred Stock and Series C Preferred Stock shall rank senior to the Series D Preferred Stock, and the Series D Preferred Stock shall rank on a parity with the Series E Preferred Stock, as to dividends and upon liquidation, dissolution or winding up, in each case as described in Section 2 or 3, respectively, provided that the Series D Preferred Stock shall so rank on a parity with the Series C Preferred Stock at such times as there shall be no shares of Series A Preferred Stock outstanding. Any other class or series of stock of the Corporation shall be deemed to rank:
- (a) prior to the Series D Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in Section 3, respectively, if the holders of such class shall be entitled to the receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in preference or priority to the holders of the Series D Preferred Stock;
- (b) on a parity with the Series D Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in section 3, respectively, whether or not the dividend rates, dividend payment dates or redemption or liquidation prices per share thereof be different from those of the Series D Preferred Stock, if the holders of such class of stock and the Series D Preferred Stock shall be entitled to the receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in proportion to their respective amounts of accrued and unpaid dividends per share or liquidation prices, without preference or priority one over the other; and
- (c) junior to the Series D Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in section 3, respectively, if the holders of Series D Preferred Stock shall be entitled to receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in preference or priority to the holders of shares of such stock.

- 9. DEFINITIONS. For purposes of this Certificate of Designations, Preferences and Rights of Series D Preferred Stock, the following terms shall have the meanings indicated:
 - (a) "Automatic Conversion" is defined in Section 5(a).
- (c) "Base Number" shall mean the number derived from dividing $$100\ by$ the Initial Common Stock Price.
- (d) "Business Day" shall mean any day other than a Saturday, Sunday, or a day on which banking institutions in the State of New York or The Commonwealth of Massachusetts are authorized or obligated by law or executive order to close or a day which is or is declared a national or New York or Massachusetts state holiday.
- (i) "Call Price" of each share of Series D Preferred Stock shall mean an amount equal to 100% of the Initial Preferred Stock Price.
- (e) "Closing Price" with respect to any securities on any day shall mean the closing sale price regular way on such day or, in case no such sale takes place on such day, the average of the reported closing bid and asked prices, regular way, in each case on the New York Stock Exchange, or, if such security is not listed or admitted to trading on such Exchange, on the principal national securities exchange or quotation system on which such security is quoted or listed or admitted to trading, or, if not quoted or listed or admitted to trading on any national securities exchange or quotation system, the average of the closing bid and asked prices of such security on the over-the-counter market on the day in question as reported by the National Association of Securities Dealers, Inc. Automated Quotation System, or a similarly generally accepted reporting service, or if not so available, in such manner as furnished by any New York Stock Exchange member firm selected from time to time by the Board of Directors for that purpose.
- (f) "Current Market Price" shall mean the average of the daily Closing Prices per share of Common Stock for the ten consecutive Trading Days immediately prior to the date in question, provided, however, that, if any event that results in an adjustment of the Exchange Rate occurs during the period beginning on the first day of such ten-day period and ending on the applicable conversion or redemption date, the Current Market Price as determined pursuant to the foregoing shall be appropriately adjusted to reflect the occurrence of such event.
- (g) The "Exchange Rate" shall be equal to (a) if the Current Market Price on the date of determination is equal to or greater than 120% of the Initial Common Stock Price (the "Threshold Common Stock Price"), the number of shares of Common Stock equal to 0.83333333 of the Base Number (the "Upper Exchange Rate"), (b) if the Current Market Price

on the date of determination is less than the Threshold Common Stock Price but greater than 80% of the Initial Common Stock Price, the number of shares of Common Stock having a value (determined at the Current Market Price) equal to the Initial Preferred Stock Price (the "Middle Exchange Rate"), and (c) if the Current Market Price on the date of determination is equal to or less than 80% of the Initial Common Stock Price, a number of shares of Common Stock (the "Lower Exchange Rate") equal to 1.25 multiplied by the Base Number. The Exchange Rate is subject to adjustment as set forth in Section 5(d).

- (h) "Fair Market Value" on any day shall mean the average of the daily Closing Prices of a share of Common Stock of the Corporation on the five (5) consecutive Trading Days selected by the Corporation commencing not more than 20 Trading Days before, and ending not later than, the earlier of the day in question and the day before the "ex" date with respect to the issuance or distribution requiring such computation. The term "'ex' date", when used with respect to any issuance or distribution, means the first day on which the Common Stock trades regular way, without the right to receive such issuance or distribution, on the exchange or in the market, as the case may be, used to determine that day's Closing Price.
- (i) "Full Cumulative Dividends" shall mean, with respect to the Series D Preferred Stock, or any other capital stock of the Corporation, as of any date the aggregate amount of all then accumulated, accrued and unpaid dividends payable on such shares of Series D Preferred Stock, or other capital stock, as the case may be, in cash, whether or not earned or declared and whether or not there shall be funds legally available for the payment thereof.
- (j) "Initial Common Stock Price" shall mean $\$ ____ per share of Common Stock./2/
- (k) "Initial Issuance Date" shall mean the date on which shares of Series D Preferred Stock are initially issued by the Company.
 - (1) "Initial Preferred Stock Price" shall mean \$100 per share.
- (m) "Lower Exchange Rate" is defined in the definition of "Exchange Rate".
- (n) "Middle Exchange Rate" is defined in the definition of "Exchange Rate".
 - (0) "Optional Conversion" is defined in Section 5(b).
 - (p) "Optional Conversion Date" is defined in Section 5(b).

/2/ Average Closing Price for 5 Trading Days prior to Closing Date, but in no event less than \$13.4375 per share or greater than \$15.4375 per share.

- (q) "Record Date" shall mean, with respect to any dividend, distribution or other transaction or event in which the holders of Common Stock have the right to receive any cash, securities or other property or in which the Common Stock (or other applicable security) is exchanged or converted into any combination of cash, securities or other property, the date fixed for determination of stockholders entitled to receive such cash, securities of other property (whether such dated is fixed by the Board of Directors or by statute, contract or otherwise), and with respect to any subdivision or combination of the Common Stock, the effective date of such subdivision or combination.
- (r) "Rights" shall mean the rights of the Corporation which are issuable under the Rights Agreement, or rights to purchase any capital stock of the Corporation under any successor shareholder rights plan or plan adopted in replacement of the Rights Agreement.
- (s) "Rights Agreement" shall mean any agreement similar to the Corporation's previous Rights Agreement dated as of April 26, 1988 between the Corporation and State Street Bank and Trust Company, as Rights Agent, as the same may be amended from time to time.
- (t) "Sale" shall mean a sale of all or substantially all of the assets or stock of an operating division or subsidiary of the Corporation other than TJ Maxx or Marshall's at a value of not less than a \$25 million premium over the book value of such assets or stock.
- (u) "Sale Proceeds" shall mean the net cash proceeds, if any (after subtracting all fees and expenses related to such transaction), received by the Corporation in respect of any Sale.
- (v) "Series A Preferred Stock" shall mean the Corporation's New Series A Cumulative Convertible Preferred Stock.
- (w) "Series C Preferred Stock" shall mean the Corporation's \$3.125 Series C Cumulative Convertible Preferred Stock.
- $\mbox{(x)}$ "Series E Preferred Stock" shall mean the Corporation's Series E Cumulative Convertible Preferred Stock.
- $\mbox{(y)}$ "Threshold Common Stock Price" is defined in the definition of "Exchange Rate".
- (z) "Trading Day" shall mean (x) if the applicable security is listed or admitted for trading on the New York Stock Exchange or another national securities exchange, a day on which the New York Stock Exchange or such other national securities exchange is open for business or (y) if the applicable security is quoted on the National Market System of the

National Association of Securities Dealers Automated Quotation System, a day on which trades may be made on such National Market System or (z) if the applicable security is not so listed, admitted for trading or quoted, any day other than a Saturday or Sunday or a day on which banking institutions in the State of New York are authorized or obligated by law or executive order to close.

- (aa) "Transfer Agent" shall mean State Street Bank and Trust Company, or any other national or state bank or trust company having combined capital and surplus of at least \$100,000,000 and designated by the Corporation as the transfer agent and/or registrar of the Series D Preferred Stock, or if no such designation is made, the Corporation.
- (bb) "Upper Exchange Rate" is defined in the definition of "Exchange Rate".

IN WITNESS WHEREOF, The TJX Companies, Inc of Designation to be signed by its Vice President this day of, 1995.	
THE T	JX COMPANIES, INC.
By:	
	Steven R. Wishner
	Vice President - Finance
Attest:	
Jay H. Meltzer	
Secretary	

A-25

CERTIFICATE OF DESIGNATIONS, PREFERENCES AND RIGHTS OF SERIES E CUMULATIVE CONVERTIBLE PREFERRED STOCK

\$1.00 PAR VALUE PER SHARE

0F

THE TJX COMPANIES, INC.

Pursuant to Section 151(g) of the General Corporation Law of the State of Delaware

We, Steven R. Wishner, Vice President - Finance, and Jay H. Meltzer, Secretary, of The TJX Companies, Inc. (hereinafter called the "Corporation"), a corporation organized and existing under and by virtue of the provisions of the General Corporation Law of the State of Delaware,

Do HEREBY CERTIFY:

FIRST: The restated certificate of incorporation, as amended (the "Certificate of Incorporation"), of the corporation authorizes the issuance of 5,000,000 shares of Preferred Stock, \$1.00 par value per share ("Preferred Stock"), in one or more series, and further authorizes the Board of Directors from time to time to provide by Resolution for the issuance of shares of Preferred Stock in one or more series not exceeding the aggregate number of shares of Preferred Stock authorized by the Certificate of Incorporation and to determine with respect to each such series, the voting powers, if any (which voting powers if granted may be full or limited), designations, preferences, the relative, participating, optional or other rights, and the qualifications, limitations and restrictions appertaining thereto.

SECOND: The Finance [Executive] Committee of the Board of Directors of the corporation, pursuant to authority conferred on such committee by the Board of Directors (which fixed the voting rights with respect to the shares designated herein), at a meeting duly called and held on _______ did duly adopt the following Resolution authorizing the creation and issuance of a series of said Preferred Stock to be known as "Series E Cumulative Convertible Preferred Stock," said Series E Cumulative Convertible Preferred Stock to be convertible into the common stock, \$1.00 par value per share (the "Common Stock"), of the corporation:

RESOLVED: that the Finance [Executive] Committee of the Board of Directors, pursuant to authority conferred on such Committee by the Board of Directors (which fixed the voting rights with respect to the shares designated herein) by the provisions of the Second Restated Certificate of Incorporation, as amended (the "Certificate of Incorporation"), of the Corporation, hereby authorizes the issuance of a series of cumulative convertible Preferred Stock of the Corporation and hereby fixes the voting powers, designations, preferences, the relative, participating, optional and other rights, and the qualifications, limitations and restrictions appertaining thereto in addition to those set forth in said Certificate of Incorporation, as follows:

1. DESIGNATION AND NUMBER. The designation of Preferred Stock created by this Resolution shall be Series E Cumulative Convertible Preferred Stock, \$1.00 par value per share, of The TJX Companies, Inc. (the "Corporation") (hereinafter referred to as the "Series E Preferred Stock"), and the number of shares constituting such series shall be 1,500,000, which number may not be increased but may be decreased (but not below the number of shares of Series E Preferred Stock then outstanding) from time to time by the Board of Directors.

All shares of Series E Preferred Stock which shall have been issued and reacquired in any manner by the Corporation (excluding, until the Corporation elects to retire them, shares which are held as treasury shares but including shares redeemed, shares purchased and retired, shares converted pursuant to Section 4 hereof and shares exchanged for any other security of the Corporation) shall not be reissued and shall, upon the making of any necessary filing with the Secretary of State of Delaware have the status of authorized but unissued shares of the Corporation's Preferred Stock, without designation as to series, and thereafter may be issued, but not as shares of Series E Preferred Stock.

DIVIDEND RIGHTS.

(a) General. The holders of shares of Series E Preferred Stock shall be entitled to receive, in preference to the holders of shares of Common Stock and any other stock ranking as to dividends junior to the Series E Preferred Stock, when and as declared by the Board of Directors, out of funds legally available therefor, cumulative cash dividends, accruing from and after the date of original issuance of the Series E Preferred Stock at an annual rate of \$7.00 per share, and no more, as long as shares of Series E Preferred Stock remain

outstanding. Dividends shall be payable quarterly in arrears, on January 1, April 1, July 1 and October 1 in each year commencing on the first of such four dates which follows the date of initial issuance of the Series E Preferred Stock (each, a "Dividend Payment Date"). Each dividend will be payable to holders of record as they appear on the stock register of the Corporation on the record date therefor, not exceeding 60 days nor less than 10 days preceding the payment date thereof, as shall be fixed by the Board of Directors. Dividends in arrears may be declared and paid at any time, without reference to any Dividend Payment Date, to holders of record on such date, not exceeding 60 days preceding the payment date thereof, as may be fixed by the Board of Directors of the Corporation. Dividends payable on the Series E Preferred Stock (i) for any period greater or less than a full dividend period, shall be computed on the basis of a 360-day year consisting of twelve 30-day months and (ii) for each full quarterly dividend period, shall be computed by dividing the annual dividend rate by four. Dividends on shares of Series E Preferred Stock shall be cumulative and shall accrue on a daily basis from the date of original issuance thereof whether or not there shall be funds legally available for the payment thereof and whether or not such dividends are declared. Holders of shares of the Series E Preferred Stock shall not be entitled to any dividend, whether payable in cash, property or stock, in excess of Full Cumulative Dividends on such shares. No interest or sum of money in lieu of interest shall be payable in respect of any dividend payment or payments which may be in arrears.

- (b) Requirements for Dividends on Senior Preferred Stock. The Corporation shall not (i) declare or pay or set apart for payment any dividends or distributions on shares of Series E Preferred Stock (other than dividends paid in shares of stock ranking junior to any series of Preferred Stock ranking senior to the Series E Preferred Stock as to dividends) or (ii) make any purchase or redemption of, or any sinking fund payment for the purchase or redemption of, shares of Series E Preferred Stock (other than a purchase or redemption made by issue or delivery of any stock ranking junior to any series of Preferred Stock ranking senior to the Series E Preferred Stock as to dividends or upon liquidation, dissolution or winding up) unless Full Cumulative Dividends on all outstanding shares of any series of Preferred Stock ranking senior to Series E Preferred Stock through the most recent dividend payment date prior to the date of payment of such dividend or distribution, or effective date of such purchase, redemption or sinking fund payment, shall have been paid in full or declared and a sufficient sum set apart for payment thereof.
- (c) Requirements for Dividends on Parity Preferred Stock. If there shall be outstanding shares of any other class or series of Preferred Stock ranking on a parity with the Series E Preferred Stock as to dividends, no dividends, except as described in the next sentence, shall be declared or paid or set apart for payment on any such other series for any period unless Full Cumulative Dividends on the Series E Preferred Stock through the most recent Dividend Payment Date have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof is set apart for such payment. If dividends on the Series E Preferred Stock and on any other series of Preferred Stock ranking on a parity as to dividends with the Series E Preferred Stock are in arrears, all dividends

declared upon shares of the Series E Preferred Stock and all dividends declared upon such other series shall be declared pro rata so that the amounts of dividends per share declared on the Series E Preferred Stock and such other series shall in all cases bear to each other the same ratio that Full Cumulative Dividends per share at the time on the shares of Series E Preferred Stock and on such other series bear to each other.

Requirements for Dividends on Junior Stock. The Corporation shall not (i) declare or pay or set apart for payment any dividends or distributions on any stock ranking as to dividends junior to the Series E Preferred Stock (other than dividends paid in shares of stock ranking junior to the Series E Preferred Stock as to dividends) or (ii) make any purchase or redemption of, or any sinking fund payment for the purchase or redemption of, any stock ranking as to dividends or upon liquidation, dissolution or winding up junior to the Series E Preferred Stock (other than a purchase or redemption made by issue or delivery of any stock ranking junior to the Series E Preferred Stock as to dividends or upon liquidation, dissolution or winding up) unless Full Cumulative Dividends on all outstanding shares of Series E Preferred Stock through the most recent Dividend Payment Date prior to the date of payment of such dividend or distribution, or effective date of such purchase, redemption or sinking fund payment, shall have been paid in full or declared and a sufficient sum set apart for payment thereof; provided, however, that unless prohibited by the terms of any other outstanding series of Preferred Stock, any moneys theretofore deposited in any sinking fund with respect to any Preferred Stock of the Corporation in compliance with this Section 2(d) and the provisions of such sinking fund may thereafter be applied to the purchase or redemption of such Preferred Stock in accordance with the terms of such sinking fund regardless of whether at the time of such application Full Cumulative Dividends on all outstanding shares of Series E Preferred Stock through the most recent Dividend Payment Date shall have been paid in full or declared and a sufficient sum set apart for payment thereof.

LIQUIDATION PREFERENCES.

- (a) Senior Preferred Stock. In the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, before any payment or distribution of the assets of the Corporation (whether from capital or surplus) shall be made to or set apart for the holders of the Series E Preferred Stock upon liquidation, dissolution or winding up, the holders of each class or series of Preferred Stock ranking senior to the Series E Preferred Stock upon liquidation, dissolution or winding up shall be entitled to receive full payment of their liquidation preferences.
- (b) Order of Payments among Parity Preferred Stock. In the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, before any payment or distribution of the assets of the Corporation (whether from capital or surplus) shall be made to or set apart for the holders of any class or series of stock of the Corporation ranking junior to the Series E Preferred Stock upon liquidation, dissolution or winding up, the

holders of the shares of Series E Preferred Stock and the holders of each other class or series of Preferred Stock ranking on a parity with Series E Preferred Stock upon liquidation, dissolution or winding up shall be entitled to receive liquidation payments according to the following priorities:

First,

The holders of the shares of Series E Preferred Stock shall receive \$100 per share and the holders of shares of each such other class or series of Preferred Stock shall receive the full respective liquidation preferences (including any premiums) to which they are entitled; and

Second,

The holders of shares of Series E Preferred Stock and the holders of shares of each such other class or series of Preferred Stock shall each receive an amount equal to Full Cumulative Dividends with respect to their respective shares through and including the date of final distribution to such holders, but such holders shall not be entitled to any further payment.

No payment (in either of the First step or Second step provided above) on account of any liquidation, dissolution or winding up of the Corporation shall be made to holders of any such other class or series of Preferred Stock or to the holders of Series E Preferred Stock unless there shall likewise be paid at the same time to the holders of the Series E Preferred Stock and the holders of each such other class or series of Preferred Stock like proportionate amounts of the same payments (as to each of the First step or the Second step above), such proportionate amounts to be determined ratably in proportion to the full amounts to which the holders of all outstanding shares of Series E Preferred Stock and the holders of all outstanding shares of each such other class or series of Preferred Stock are respectively entitled (in either the First step or the Second step, as the case may be) with respect to such distribution.

For purposes of this Section 3, neither a consolidation or merger of the Corporation with or into another corporation nor a merger of any other corporation with or into the Corporation or a sale or transfer of all or any part of the Corporation's assets for cash, securities or other property will be deemed a liquidation, dissolution or winding up of the Corporation.

(c) Junior Stock. After payment shall have been made in full to the holders of Series E Preferred Stock and to the holders of each such other class or series of Preferred Stock as provided in this Section 3 upon liquidation, dissolution or winding up of the Corporation, any other series or class or classes of stock ranking junior to the Series E Preferred Stock upon liquidation, dissolution or winding up shall, subject to the respective terms and provisions (if any) applying thereto, be entitled to receive any and all assets remaining to be paid or distributed upon such liquidation, dissolution or winding up, and the holders of Series E Preferred Stock shall not be entitled to share therein.

4. CONVERSION.

- (a) Automatic Conversion. Unless earlier converted at the option of the holder in accordance with the provisions of Section 4(b), on the Automatic Conversion Date each outstanding share of the Series E Preferred Stock shall convert automatically (the "Automatic Conversion") into (i) shares of Common Stock at the Exchange Rate in effect on the Automatic Conversion Date and (ii) the right to receive an amount in cash equal to Full Cumulative Dividends on such share to the Automatic Conversion Date.
- (b) Optional Conversion by Holder. Shares of Series E Preferred Stock may be converted, in whole or in part, at the option of the holder thereof ("Optional Conversion"), at any time after the Initial Issuance Date and not later than the close of business on the Business Day prior to the Automatic Conversion Date, into shares of Common Stock at the Upper Exchange Rate.

Optional Conversion of shares of Series E Preferred Stock may be effected by delivering certificates evidencing such shares, together with written notice of conversion and a proper assignment of such certificates to the Corporation or in blank (and, if applicable, payment of an amount equal to the dividend payable on such shares), to the office of any transfer agent for the Series E Preferred Stock or to any other office or agency maintained by the Corporation for that purpose and otherwise in accordance with the Optional Conversion procedures established by the Corporation. Each Optional Conversion shall be deemed to have been effected immediately prior to the close of business on the date on which the foregoing requirements shall have been satisfied (the "Optional Conversion Date").

- (c) Mechanics of Conversion.
- (i) Upon surrender in accordance with the aforesaid provisions of the certificate for any shares so converted (duly endorsed or accompanied by appropriate instruments of transfer), the holder of record of such shares shall be entitled to receive the applicable number of shares of Common Stock (calculated to the nearest 1/1,000,000th of a share) (and cash representing fractional share settlements in respect thereof) at the applicable Exchange Rate plus Full Cumulative Dividends thereon, without interest.
- (ii) Before any holder of shares of Series E Preferred Stock shall receive certificates for shares of Common Stock in respect of the conversion of shares of Series E Preferred Stock (or cash representing fractional share settlements in respect thereof) such holder shall surrender the certificate or certificates of shares of Series E Preferred Stock duly endorsed if required by the Corporation, at the office of the Corporation and, if certificates for shares of Common Stock are to be received by such holder, shall state in writing the name or names and the denominations in which such holder wishes the certificate or certificates for the Common Stock to be issued. The Corporation will, as soon as practicable after receipt thereof, issue and deliver to such holder, or such holder's designee or designees, a certificate or certificates

for the number of shares of Common Stock to which such holder shall be entitled as aforesaid, together with a certificate or certificates representing any shares of Series E Preferred Stock which are not to be converted, but which shall have constituted part of the certificate or certificates for shares of Series E Preferred Stock so surrendered.

(iii) The Corporation's obligation to deliver shares of Common Stock and provide funds upon conversion in accordance with this Section 4 shall be deemed fulfilled if, on or before a conversion date, the Corporation shall deposit with a bank or trust company, or an affiliate of a bank or trust company, having an office or agency in New York, New York and having a capital and surplus of at least \$50,000,000 according to its last published statement of condition, or shall set aside or make other reasonable provision for the issuance of, such number of shares of Common Stock as are required to be delivered by the Corporation pursuant to this Section 4 upon the occurrence of the related conversion of Series E Preferred Stock and for cash required to be paid in lieu of the issuance of fractional share amounts and Full Cumulative Dividends payable in cash on the shares of Series E Preferred Stock to be converted as required by this Section 4, in trust for the account of the holders of such shares of Series E Preferred Stock to be converted (and so as to be and continue to be available therefor), with (in the case of deposits with a bank or trust company) irrevocable instructions and authority to such bank or trust company that such shares and funds be delivered upon conversion of the shares of Series E Preferred Stock so to be converted. If on the Automatic Conversion Date shares of Common Stock and funds (if any) necessary for the conversion shall have been irrevocably either set aside by the Company separate and apart from its other funds or assets in trust for the account of the holders of the shares of Series E Preferred Stock to be converted (and so as to be and continue to be available therefor) or the Company shall have made other reasonable provision therefor, then, notwithstanding that the certificates evidencing any shares of the Series E Preferred Stock so subject to conversion shall not have been surrendered, the shares represented thereby shall be deemed no longer outstanding, dividends with respect to such shares shall cease to accrue on the date fixed for conversion (provided that holders of shares of Series E Preferred Stock at the close of business on a record date for any payment of dividends shall be entitled to receive Full Cumulative Dividends payable on such shares on the corresponding Dividend Payment Date notwithstanding the conversion of such shares following such record date and prior to such Dividend Payment Date) and all rights with respect to such shares shall forthwith after such date cease and terminate, except for the rights of the holders to receive the shares of Common Stock and funds (if any) payable pursuant to this Section 4 without interest upon surrender of their certificates therefor. Holders of shares of Series E Preferred Stock at the close of business on a dividend payment record date shall be entitled to receive the dividend payable on such shares on the corresponding Dividend Payment Date notwithstanding the Optional Conversion of such shares following such record date and prior to such Dividend Payment Date. However, shares of Series E Preferred Stock surrendered for Optional Conversion after the close of business on a dividend payment record date and before the opening of business on the corresponding Dividend Payment Date must be accompanied by payment in cash of an amount equal to the dividend payable on such shares on such Dividend Payment Date. A

holder of shares of Series E Preferred Stock on a dividend record date who (or whose transferee) surrenders any such shares for conversion into shares of Common Stock on the corresponding Dividend Payment Date will receive the dividend payable by the Corporation on such shares of Series E Preferred Stock on such Dividend Payment Date, and the converting holder need not include payment of the amount of such dividend upon surrender of shares of Series E Preferred Stock for conversion. Except as provided above, upon any conversion of shares of Series E Preferred Stock, the Corporation shall make no payment or allowance for unpaid dividends, whether or not in arrears, on such shares of Series E Preferred Stock as to which conversion has been effected or for dividends or distributions on the shares of Common Stock issued upon such conversion.

- (iv) Holders of shares of Series E Preferred Stock that are converted shall not be entitled to receive dividends declared and paid on such shares of Common Stock, and such shares of Common Stock shall not be entitled to vote, until such shares of Common Stock are issued upon the surrender of the certificates representing such shares of Series E Preferred Stock and upon such surrender such holders shall be entitled to receive such dividends declared and paid on such shares of Common Stock subsequent to such conversion date. Amounts payable in cash in respect of the shares of Series E Preferred Stock or in respect of such shares of Common Stock shall not bear interest.
- (v) Each conversion of shares of Series E Preferred Stock into Common Stock shall be deemed to have been made as of the close of business on the applicable conversion date, so that the rights of the holder of such shares of Series E Preferred Stock shall, to the extent of such conversion, cease at such time and the person or persons entitled to receive shares of the Common Stock upon conversion of such shares shall be treated for all purposes as having become the record holder or holders of the Common Stock at such time; provided, however, that if an event that results in an adjustment to the Exchange Rate is declared or occurs with respect to the shares of Common Stock, and the record date for any such action is on or after the close of business on the date on which notice of such conversion is given, but prior to the close of business on the date of such conversion, then the person or persons entitled to receive shares of the Common Stock upon conversion of shares of Series E Preferred Stock shall be treated for purposes of such action as having become the record holder or holders of the Common Stock at the close of business on the Trading Day next preceding the date on which notice of such conversion is given.
- (vi) The Corporation will pay any and all taxes that may be payable in respect of the issuance or delivery of shares of Common Stock upon conversion of shares of Series E Preferred Stock pursuant hereto. The Corporation shall not, however, be required to pay any tax which may be payable in respect of any transfer involved in the delivery of shares registered in a name other than the name in which such shares of Series E Preferred Stock were formerly registered, and no such issue or delivery shall be made unless and until the person requesting such issue or delivery has paid to the Corporation the amount of any such tax, or has established, to the satisfaction of the Corporation, that such tax has been paid.

- (d) Adjustments to the Exchange Rate. The Exchange Rate shall be subject to adjustment from time to time as provided below in this paragraph (d).
 - (i) If the Corporation shall pay or make a dividend or other distribution with respect to its Common Stock in shares of Common Stock (including by way of reclassification of any shares of its Common Stock) to all holders of Common Stock, the Exchange Rate in effect at the opening of business on the day following the date fixed for the determination of stockholders entitled to receive such dividend or other distribution shall be increased by multiplying such Exchange Rate by a fraction of which the numerator shall be the sum of the number of shares of Common Stock outstanding at the close of business on the date fixed for such determination plus the total number of shares of Common Stock constituting such dividend or other distribution, and of which the denominator shall be the number of shares of Common Stock outstanding at the close of business on the date fixed for such determination, such increase to become effective immediately after the opening of business on the day following the date fixed for such determination.
 - (ii) In case outstanding shares of Common Stock shall be subdivided into a greater number of shares of Common Stock, the Exchange Rate in effect at the opening of business on the day following the day upon which such subdivision becomes effective shall be proportionately increased, and, conversely, in case outstanding shares of Common Stock shall be combined into a smaller number of shares of Common Stock, the Exchange Rate in effect at the opening of business on the day following the day upon which such combination becomes effective shall be proportionately reduced, such increases or reductions, as the case may be, to become effective immediately after the opening of business on the day following the day upon which such subdivision or combination becomes effective.
 - (iii) If the Corporation shall, after the date hereof, issue rights or warrants, in each case other than the Rights, to all holders of its Common Stock entitling them (for a period not exceeding 45 days from the date of such issuance) to subscribe for or purchase shares of Common Stock at a price per share less than the Fair Market Value of the Common Stock on the record date for the determination of stockholders entitled $% \left(1\right) =\left(1\right) \left(1\right) \left($ to receive such rights or warrants, then in each case the Exchange Rate shall be adjusted by multiplying the Exchange Rate in effect on such record date, by a fraction of which the numerator shall be the number of shares of Common Stock outstanding on the date of issuance of such rights or warrants, immediately prior to such issuance, plus the number of additional shares of Common Stock offered for subscription or purchase pursuant to such rights or warrants, and of which the denominator shall be the number of shares of Common Stock outstanding on the date of issuance of such rights or warrants, immediately prior to such issuance, plus the number of shares of Common Stock which the aggregate offering price of the total number of shares of Common

Stock so offered for subscription or purchase pursuant to such rights or warrants would purchase at such Fair Market Value (determined by multiplying such total number of shares by the exercise price of such rights or warrants and dividing the product so obtained by such Fair Market Value). Such adjustment shall become effective at the opening of business on the Business Day next following the record date for the determination of stockholders entitled to receive such rights or warrants. To the extent that shares of Common Stock are not delivered after the expiration of such rights or warrants, the Exchange Rate shall be readjusted to the Exchange Rate which would then be in effect had the adjustments made upon the issuance of such rights or warrants been made upon the basis of the issuance of rights or warrants in respect of only the number of shares of Common Stock actually delivered.

- (iv) If the Corporation shall pay a dividend or make a distribution to all holders of its Common Stock consisting of evidences of its indebtedness or other assets (including shares of capital stock of the Corporation other than Common Stock but excluding any cash dividends or any dividends or other distributions referred to in clauses (i) and (ii) above), or shall issue to all holders of its Common Stock rights or warrants to subscribe for or purchase any of its securities (other than those referred to in clause (iii) above and other than Rights), then in each such case the Exchange Rate shall be adjusted by multiplying the Exchange Rate in effect on the record date for such dividend or distribution or for the determination of stockholders entitled to receive such rights or warrants, as the case may be, by a fraction of which the numerator shall be the Fair Market Value per share of the Common Stock on such record date, and of which the denominator shall be such Fair Market Value per share of Common Stock less the fair market value (as determined by the Board of Directors, whose determination shall be conclusive) as of such record date of the portion of the assets or evidences of indebtedness so distributed, or of such subscription rights or warrants, applicable to one share of Common Stock. Such adjustment shall become effective on the opening of business on the Business Day next following the record date for such dividend or distribution or for the determination of stockholders entitled to receive such rights or warrants, as the case may be.
- (v) Any share of Common Stock issuable in payment of a dividend or other distribution shall be deemed to have been issued immediately prior to the close of business on the record date for such dividend or other distribution for purposes of calculating the number of outstanding shares of Common Stock under subparagraph (ii) above.
- (vi) Anything in this paragraph (d) notwithstanding, the Corporation shall be entitled to make such upward adjustments in the Exchange Rate, in addition to those required by this paragraph (d), as the Corporation in its sole discretion shall determine to be advisable, in order that any stock dividends, subdivision of shares, distribution of rights to purchase stock or securities, or distribution of securities convertible into or

exchangeable for stock (or any transaction which could be treated as any of the foregoing transactions pursuant to Section 305 of the Internal Revenue Code of 1986, as amended) hereafter made by the Corporation to its stockholders shall not be taxable.

(vii) In any case in which this paragraph (d) shall require that an adjustment as a result of any event become effective at the opening of business on the Business Day next following a record date and the date fixed for conversion pursuant to paragraph (a) occurs after such record date, but before the occurrence of such event, the Corporation may in its sole discretion elect to defer the following until after the occurrence of such event: (A) issuing to the holder of any shares of Series E Preferred Stock surrendered for conversion the additional shares of Common Stock issuable upon such conversion over the shares of Common Stock issuable before giving effect to such adjustment; and (B) paying to such holder any amount in cash in lieu of a fractional share of Common Stock pursuant to Section 5(d).

(viii) For purposes hereof, an "adjustment in the Exchange Rate" means, and shall be implemented by, an adjustment of the nature and amount specified, effected in the manner specified, in each of the Upper Exchange Rate, the Middle Exchange Rate and the Lower Exchange Rate. If an adjustment is made to the Exchange Rate pursuant to this paragraph (d), a proportionate adjustment in the same direction shall also be made on the Automatic Conversion Date to the Current Market Price solely to determine which of clauses (a), (b) or (c) of the definition of Exchange Rate will apply on the Automatic Conversion Date. Such adjustment shall be made by multiplying the Current Market Price by a fraction of which the numerator shall be the Exchange Rate immediately after such adjustment pursuant to this paragraph (d) and the denominator shall be the Exchange Rate immediately before such adjustment. All adjustments to the Exchange Rate shall be calculated to the nearest 1/1,000,000th of a share of Common Stock. No adjustment in the Exchange Rate shall be required unless such adjustment would require an increase or decrease of at least one percent in the Exchange Rate; provided, however, that any adjustments which by reason of this subparagraph are not required to be made shall be carried forward and taken into account in any subsequent adjustment. All adjustments to the Exchange Rate shall be made successively.

- (ix) Before taking any action that would cause an adjustment increasing the Exchange Rate such that the conversion price (for purposes of this paragraph (d), an amount equal to the liquidation value per share of Series E Preferred Stock divided by the Upper Exchange Rate as in effect from time to time) would be below the then par value of the Common Stock, the Corporation will take any corporate action which may, in the opinion of its counsel, be necessary in order that the Corporation may validly and legally issue fully paid and nonassessable shares of Common Stock at the Upper Exchange Rate as so adjusted.
- (e) Adjustment for Certain Consolidations or Mergers. In case of any consolidation $% \left(1\right) =\left(1\right) \left(1\right$

or merger to which the Corporation is a party (other than a merger or consolidation in which the Corporation is the continuing corporation and in which the Common Stock outstanding immediately prior to the merger or consolidation remains unchanged), or in case of any sale or transfer to another corporation of the property of the Corporation as an entirety or substantially as an entirety, or in case of any statutory exchange of securities with another corporation (other than in connection with a merger or acquisition), proper provision shall be made so that each share of the Series E Preferred Stock shall, after consummation of such transaction, be subject to (i) conversion at the option of the holder into the kind and amount of securities, cash or other property receivable upon consummation of such transaction by a holder of the number of shares of Common Stock into which such share of Series E Preferred Stock would have been converted if the conversion had occurred immediately prior to consummation of such transaction (based on the Exchange Rate in effect immediately prior to such consummation) and (ii) conversion on the Automatic Conversion Date into the kind and amount of securities, cash or other property receivable upon consummation of such transaction by a holder of the number of shares of Common Stock into which such share of Series E Preferred Stock would have been converted if the conversion on the Automatic Conversion Date had occurred immediately prior to the date of consummation of such transaction (based on the Exchange Rate in effect immediately prior to such consummation); assuming in each case that such holder of Common Stock failed to exercise rights of election, if any, as to the kind or amount of securities, cash or other property receivable upon consummation of such transaction (provided that if the kind or amount of securities, cash or other property receivable upon consummation of such transaction is not the same for each nonelecting share, then the kind and amount of securities, cash or other property receivable upon consummation of such transaction for each nonelecting share shall be deemed to be the kind and amount so receivable per share by a plurality of the nonelecting shares). The kind and amount of securities into which the shares of the Series E Preferred Stock shall be convertible after consummation of such transaction shall be subject to adjustment as described in paragraph (d) following the date of consummation of such transaction. The Corporation may not become a party to any such transaction unless the terms thereof are consistent with the foregoing.

- (f) Notice of Adjustments. Whenever the Exchange Rate is adjusted as provided in paragraph (d), the Corporation shall:
 - (i) Forthwith compute the adjusted Exchange Rate and prepare a certificate signed by the Chief Financial Officer, any Vice President, the Treasurer or the Controller of the Corporation setting forth the adjusted Exchange Rate, the method of calculation thereof in reasonable detail and the facts requiring such adjustment and upon which such adjustment is based, which certificate shall be prima facie evidence of the correctness of the adjustment, and file such certificate forthwith with the Transfer Agent;
 - (ii) Make a prompt public announcement stating that the Exchange Rate has been adjusted and setting forth the adjusted Exchange Rate; and

- (iii) Promptly mail a notice (stating that the Exchange Rate has been adjusted and the facts requiring such adjustment and upon which such adjustment is based and setting forth the adjusted Exchange Rate) to the holders of record of the outstanding shares of the Series E Preferred Stock at or prior to the time the Corporation mails an interim statement to its stockholders covering the fiscal quarter during which the facts requiring such adjustment occurred but in any event within 45 days of the end of such fiscal quarter.
- (g) Prior Notice of Certain Events. In case:
- (i) the Corporation shall (1) declare any dividend (or any other distribution) on its Common Stock, other than a dividend payable solely in cash in an amount such that the aggregate cash dividend per share of Common Stock in any fiscal quarter does not exceed 3.75% of the Current Market Price of the Common Stock on the Trading Day next preceding the date of declaration of such dividend, or (2) declare or authorize a redemption or repurchase of in excess of 10% of the then outstanding shares of Common Stock; or
- (ii) the Corporation shall authorize the granting to all holders of Common Stock of rights or warrants to subscribe for or purchase any shares of stock of any class or of any other rights or warrants (other than Rights); or
- (iii) of any reclassification of Common Stock (other than a subdivision or combination of the outstanding Common Stock, or a change in par value, or from par value to no par value, or from no par value to par value), or of any consolidation or merger to which the Corporation is a party and for which approval of any stockholders of the Corporation shall be required, or of the sale or transfer of all or substantially all of the assets of the Corporation or of any compulsory share exchange where the Common Stock is converted into other securities, cash or other property; or
- (iv) of the voluntary or involuntary liquidation, dissolution or winding up of the Corporation; $\;$

then the Corporation shall cause to be filed with the Transfer Agent and each office or agency maintained for conversion of shares of Series E Preferred Stock, and shall cause to be mailed to the holders of record of the Series E Preferred Stock, at their last addresses as they shall appear upon the stock transfer books of the Corporation, at least 15 days prior to the applicable record date hereinafter specified, a notice stating (x) the date on which a record (if any) is to be taken for the purpose of such dividend, distribution, redemption, repurchase or granting of rights or warrants or, if a record is not to be taken, the date as of which the holders of Common Stock of record to be entitled to such dividend, distribution, redemption, rights or warrants are to be determined or (y) the date on which such reclassification, consolidation, merger, sale, transfer, share exchange, liquidation, dissolution or winding up is expected to become effective, and the date as of which it is expected that holders of Common

Stock of record shall be entitled to exchange their shares of Common Stock for securities or other property (including cash) deliverable upon such reclassification, consolidation, merger, sale, transfer, share exchange, liquidation, dissolution or winding up. No failure to mail such notice or any defect therein or in the mailing thereof shall affect the validity of the corporate action required to be specified in such notice.

- Dividend or Interest Reinvestment Plans; Other. Notwithstanding the foregoing provisions, the issuance of any shares of Common Stock pursuant to any plan providing for the reinvestment of dividends or interest payable on securities of the Corporation and the investment of additional optional amounts in shares of Common Stock under any such plan, and the issuance of any shares of Common Stock or options or rights to purchase such shares pursuant to any employee benefit plan or program of the Corporation, or pursuant to any option, warrant, right or exercisable, exchangeable or convertible security outstanding as of the date the Series E Preferred Stock was first designated, shall not be deemed to constitute an issuance of Common Stock or exercisable, exchangeable or convertible securities by the Corporation to which any of the adjustment provisions described above applies. There shall be no adjustment of the Exchange Rate in case of the issuance of any stock (or securities convertible into or exchangeable for stock) of the corporation except as described in this Section 4. Except as expressly set forth in this Section 4, if any action would require adjustment of the Exchange Rate pursuant to more than one of the provisions described above, only one adjustment shall be made and such adjustment shall be the amount of adjustment which has the highest absolute value.
- (i) For purposes of this Section 4, the number of shares of Common Stock at any time outstanding shall not include any shares of Common Stock then owned or held, directly or indirectly through a subsidiary, by or for the account of the Corporation.
- 5. RESERVATION OF SHARES; LISTING OF SHARES, ETC.
- (a) Reservation of Shares. The Corporation shall at all times reserve and keep available, out of its authorized and unissued stock, solely for the purpose of effecting the conversion of the Series E Preferred Stock, the full number of shares of its Common Stock deliverable upon conversion of all shares of Series E Preferred Stock not theretofore converted.
- (b) Listing of Shares. If any shares of Common Stock required to be reserved for purposes of conversion of the Series E Preferred Stock hereunder require registration with or approval of any governmental authority under any Federal or State law before such shares may be issued upon conversion, the Corporation will in good faith and as expeditiously as possible endeavor to cause such shares to be duly registered or approved, as the case may be. If the Common Stock is listed on the New York Stock Exchange or any other national securities exchange, the Corporation will, as expeditiously as possible, if permitted by the rules of such exchange, cause to be listed and keep listed on such exchange, upon official notice of issuance,

all shares of Common Stock issuable upon conversion of the Series E Preferred

- (c) Shares Issued on Conversion to be Fully Paid, Etc. The shares of Common Stock issuable upon conversion of the shares of Series E Preferred Stock, when the same shall be issued in accordance with the terms hereof, are hereby declared to be and shall be fully paid and nonassessable shares of Common Stock in the hands of the holders thereof.
- (d) No Fractional Shares. No fractional shares or scrip representing fractional shares of Common Stock shall be issued upon conversion of Series E Preferred Stock. Instead of any fractional share of Common Stock that would otherwise be issuable upon conversion of any shares of Series E Preferred Stock, the Corporation shall pay a cash adjustment in respect of such fractional interest in an amount equal to the same fraction of the Closing Price of a share of Common Stock (or, if there is no such Closing Price, the fair market value of a share of Common Stock, as determined or prescribed by the Board of Directors) at the close of business on the Trading Day immediately preceding the date of conversion.
- (e) Other Action. If the Corporation shall take any action affecting the Common Stock, other than action described in Section 4, that in the opinion of the Board of Directors would materially adversely affect the conversion rights of the holders of the shares of Series E Preferred Stock, the Exchange Rate for the Series E Preferred Stock may be adjusted, to the extent permitted by law, in such manner, if any, and at such time, as the Board of Directors may determine to be equitable in the circumstances.
- 6. VOTING RIGHTS. Other than as required by applicable law, the Series E Preferred Stock shall not have any voting powers either general or special except that:
- (a) Unless a greater vote or consent shall then be required by law, the affirmative vote or consent of two-thirds of the votes to which the holders of the outstanding shares of the Series E Preferred Stock, and each other series of Preferred Stock of the Corporation similarly affected, if any, voting together as a single class, are entitled shall be necessary for authorizing, effecting or validating the amendment, alteration or repeal of any of the provisions of the Certificate of Incorporation (including any Certificate of Designations, Preferences and Rights or any similar document relating to any series of Preferred Stock) of the Corporation, including any amendment or supplement thereto, if such would materially and adversely affect the preferences, rights, powers or privileges, qualification, limitations and restrictions of the Series E Preferred Stock and any such other series of Preferred Stock; provided, however, that the creation, issuance or increase in the amount of authorized shares of any series of Preferred Stock ranking on a parity with or junior to the Series E Preferred Stock as to the payment of dividends or upon liquidation, dissolution or winding up will not be deemed to materially and adversely affect such rights, powers or privileges, qualification, limitations and restrictions.
- (b) Unless the vote or consent of the holders of a greater number of shares shall then $% \left(1\right) =\left\{ 1\right\} =\left\{$

be required by law, the affirmative vote or consent of two-thirds of the votes to which the holders of the outstanding shares of the Series E Preferred Stock, and all other series of Preferred Stock of the Corporation ranking on parity with shares of the Series E Preferred Stock (either as to dividends or upon liquidation, dissolution or winding up) as to which like voting rights have been conferred, voting together as a single class, are entitled shall be necessary to create, authorize or issue, or reclassify any authorized stock of the Corporation into, or create, authorize or issue any obligation or security convertible into or evidencing a right to purchase, any shares of any class or series of stock of the Corporation ranking prior to the Series E Preferred Stock or ranking prior to any other class or series of Preferred Stock of the Corporation which ranks on a parity with the Series E Preferred Stock as to dividends or upon liquidation, dissolution or winding up.

- Whenever, at any time or times, dividends payable on the shares of Series E Preferred Stock shall be in arrears in an amount equal to at least six full quarterly dividends on shares of the Series E Preferred Stock at the time outstanding, the holders of the outstanding shares of Series E Preferred Stock shall have the exclusive right, voting together as a class with holders of shares of any one or more other series of Preferred Stock ranking on a parity with the Series E Preferred Stock (either as to dividends or upon liquidation, dissolution or winding up) upon which like voting rights have been conferred and are then exercisable, to elect two (2) directors of the Corporation for one-year terms at the Corporation's next annual meeting of stockholders and at each subsequent annual meeting of stockholders. If the right to elect directors shall have accrued to the holders of the Series E Preferred Stock more than 90 days prior to the date established for the next annual meeting of stockholders, the President of the Corporation shall, within 20 days after delivery to the Corporation at its principal office of a written request for a special meeting signed by the holders of at least 10% of all outstanding shares of the Series E Preferred Stock, call a special meeting of the holders of Series E Preferred Stock to be held within 60 days after the delivery of such request for the purpose of electing such additional directors. Upon the vesting of such right of the holders of Series E Preferred Stock, the maximum authorized number of members of the Board of Directors shall automatically be increased by two and the two vacancies so created shall be filled by vote of the holders of the outstanding shares of Series E Preferred Stock (either alone or together with the holders of shares of any one or more other such series of Preferred Stock entitled to vote in such election) as set forth above. The right of the holders of Series E Preferred Stock to elect members of the Board of Directors of the Corporation as aforesaid shall continue until such time as all dividends in arrears on the Series E Preferred Stock shall have been paid in full or declared and set apart for payment, at which time such right shall terminate, except as herein or by law expressly provided, subject to revesting in the event of each and every subsequent default of the character above described.
- (d) Upon termination of such special voting rights attributable to all holders of the Series E Preferred Stock and any other such series of Preferred Stock ranking on a parity with the Series E Preferred Stock as to dividends or upon liquidation, dissolution or winding up and upon which like voting rights have been conferred and are exercisable, the term of office of

each director elected by the holders of shares of Series E Preferred Stock and such parity Preferred Stock (a "Preferred Stock Director") pursuant to such special voting rights shall immediately terminate and the number of directors constituting the entire Board of Directors shall be reduced by the number of Preferred Stock Directors. Any Preferred Stock Director may be removed by, and shall not be removed otherwise than by, a majority of the votes to which the holders of the outstanding shares of Series E Preferred Stock and all other such series of Preferred Stock ranking on a parity with the Series E Preferred Stock with respect to dividends who were entitled to participate in such Preferred Stock Directors election, voting as a single class, are entitled. If the office of any Preferred Stock Director becomes vacant by reason of death, resignation, retirement, disqualification, removal from office, or otherwise, the remaining Preferred Stock Director may choose a successor who shall hold office for the unexpired term in respect of which such vacancy occurred.

- (e) In connection with any right to vote, each holder of Series E Preferred Stock shall be entitled to one vote for each share held (the holders of shares of any other series of Preferred Stock being entitled to such number of votes, if any, for each share of stock held as may be granted to them).
- 7. RANKING. The Common Stock shall rank junior to the Series E Preferred Stock as to dividends and upon liquidation, dissolution or winding up, as described in Sections 2 and 3. The Series A Preferred Stock and Series C Preferred Stock shall rank senior to the Series E Preferred Stock, and the Series D Preferred Stock shall rank on a parity with the Series E Preferred Stock, as to dividends and upon liquidation, dissolution or winding up, in each case as described in Section 2 or 3, respectively, provided that the Series E Preferred Stock shall so rank on a parity with the Series C Preferred Stock at such times as there shall be no shares of Series A Preferred Stock outstanding. Any other class or series of stock of the Corporation shall be deemed to rank:
- (a) prior to the Series E Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in Section 3, respectively, if the holders of such class shall be entitled to the receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in preference or priority to the holders of the Series E Preferred Stock;
- (b) on a parity with the Series E Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in section 3, respectively, whether or not the dividend rates, dividend payment dates or redemption or liquidation prices per share thereof be different from those of the Series E Preferred Stock, if the holders of such class of stock and the Series E Preferred Stock shall be entitled to the receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in proportion to their respective amounts of accrued and unpaid dividends per share or liquidation prices, without preference or priority one over the other: and

- (c) junior to the Series E Preferred Stock, as to dividends or upon liquidation, dissolution or winding up as described in section 3, respectively, if the holders of Series E Preferred Stock shall be entitled to receipt of dividends or of amounts distributable upon such a liquidation, dissolution or winding up, as the case may be, in preference or priority to the holders of shares of such stock.
- 8. DEFINITIONS. For purposes of this Certificate of Designations, Preferences and Rights of Series E Preferred Stock, the following terms shall have the meanings indicated:
 - (a) "Automatic Conversion" is defined in Section 4(a).
- (b) "Automatic Conversion Date" shall mean the third anniversary of the Initial Issuance Date.
- (c) "Base Number" shall mean the number derived from dividing \$100 by the Initial Common Stock Price.
- (d) "Business Day" shall mean any day other than a Saturday, Sunday, or a day on which banking institutions in the State of New York or The Commonwealth of Massachusetts are authorized or obligated by law or executive order to close or a day which is or is declared a national or New York or Massachusetts state holiday.
- (e) "Closing Price" with respect to any securities on any day shall mean the closing sale price regular way on such day or, in case no such sale takes place on such day, the average of the reported closing bid and asked prices, regular way, in each case on the New York Stock Exchange, or, if such security is not listed or admitted to trading on such Exchange, on the principal national securities exchange or quotation system on which such security is quoted or listed or admitted to trading, or, if not quoted or listed or admitted to trading on any national securities exchange or quotation system, the average of the closing bid and asked prices of such security on the over-the-counter market on the day in question as reported by the National Association of Securities Dealers, Inc. Automated Quotation System, or a similarly generally accepted reporting service, or if not so available, in such manner as furnished by any New York Stock Exchange member firm selected from time to time by the Board of Directors for that purpose.
- (f) "Current Market Price" shall mean the average of the daily Closing Prices per share of Common Stock for the ten consecutive Trading Days immediately prior to the date in question, provided, however, that, if any event that results in an adjustment of the Exchange Rate occurs during the period beginning on the first day of such ten-day period and ending on the applicable conversion date, the Current Market Price as determined pursuant to the foregoing shall be appropriately adjusted to reflect the occurrence of such event.
- (g) The "Exchange Rate" shall be equal to (a) if the Current Market Price on the $\,$

date of determination is equal to or greater than 120% of the Initial Common Stock Price (the "Threshold Common Stock Price"), the number of shares of Common Stock equal to 0.83333333 of the Base Number (the "Upper Exchange Rate"), (b) if the Current Market Price on the date of determination is less than the Threshold Common Stock Price but greater than the Initial Common Stock Price, the number of shares of Common Stock having a value (determined at the Current Market Price) equal to the Initial Preferred Stock Price (the "Middle Exchange Rate"), and (c) if the Current Market Price on the date of determination is equal to or less than the Initial Common Stock Price, a number of shares of Common Stock (the "Lower Exchange Rate") equal to the Base Number; provided that for all purposes relating to optional conversion by a holder pursuant to Section 4(b) the Exchange Rate shall be equal to the Upper Exchange Rate. The Exchange Rate is subject to adjustment as set forth in Section 4(d).

- (h) "Fair Market Value" on any day shall mean the average of the daily Closing Prices of a share of Common Stock of the Corporation on the five (5) consecutive Trading Days selected by the Corporation commencing not more than 20 Trading Days before, and ending not later than, the earlier of the day in question and the day before the "ex" date with respect to the issuance or distribution requiring such computation. The term "'ex' date", when used with respect to any issuance or distribution, means the first day on which the Common Stock trades regular way, without the right to receive such issuance or distribution, on the exchange or in the market, as the case may be, used to determine that day's Closing Price.
- (i) "Full Cumulative Dividends" shall mean, with respect to the Series E Preferred Stock, or any other capital stock of the Corporation, as of any date the aggregate amount of all then accumulated, accrued and unpaid dividends payable on such shares of Series E Preferred Stock, or other capital stock, as the case may be, in cash, whether or not earned or declared and whether or not there shall be funds legally available for the payment thereof.
- (j) "Initial Common Stock Price" shall mean $\$ ____ per share of Common Stock./1/
- (k) "Initial Issuance Date" shall mean the date on which shares of Series E Preferred Stock are initially issued by the Company.
 - (1) "Initial Preferred Stock Price" shall mean \$100 per share.
- (m) "Lower Exchange Rate" is defined in the definition of "Exchange Rate".

^{/1/} Average Closing Price for 5 Ttrading Days prior to Closing Date, but in no event less than \$13.4375 per share or greater than \$15.4375 per share.

- (n) "Middle Exchange Rate" is defined in the definition of "Exchange Rate".
 - (o) "Optional Conversion" is defined in Section 4(b).
 - (p) "Optional Conversion Date" is defined in Section 4(b).
- (q) "Record Date" shall mean, with respect to any dividend, distribution or other transaction or event in which the holders of Common Stock have the right to receive any cash, securities or other property or in which the Common Stock (or other applicable security) is exchanged or converted into any combination of cash, securities or other property, the date fixed for determination of stockholders entitled to receive such cash, securities of other property (whether such dated is fixed by the Board of Directors or by statute, contract or otherwise), and with respect to any subdivision or combination of the Common Stock, the effective date of such subdivision or combination.
- (r) "Rights" shall mean the rights of the Corporation which are issuable under the Rights Agreement, or rights to purchase any capital stock of the Corporation under any successor shareholder rights plan or plan adopted in replacement of the Rights Agreement.
- (s) "Rights Agreement" shall mean any agreement similar to the Corporation's previous Rights Agreement dated as of April 26, 1988 between the Corporation and State Street Bank and Trust Company, as Rights Agent, as the same may be amended from time to time.
- (t) "Series A Preferred Stock" shall mean the Corporation's New Series A Cumulative Convertible Preferred Stock.
- (u) "Series C Preferred Stock" shall mean the Corporation's 3.125 Series C Cumulative Convertible Preferred Stock.
- (v) "Series D Preferred Stock" shall mean the Corporation's Series D Cumulative Convertible Preferred Stock.
- (w) "Threshold Common Stock Price" is defined in the definition of "Exchange Rate".
- (x) "Trading Day" shall mean (x) if the applicable security is listed or admitted for trading on the New York Stock Exchange or another national securities exchange, a day on which the New York Stock Exchange or such other national securities exchange is open for business or (y) if the applicable security is quoted on the National Market System of the National Association of Securities Dealers Automated Quotation System, a day on which trades may be made on such National Market System or (z) if the applicable security is not so listed, admitted for trading or quoted, any day other than a Saturday or Sunday or a day on

which banking institutions in the State of New York are authorized or obligated by law or executive order to close.

- (y) "Transfer Agent" shall mean State Street Bank and Trust Company, or any other national or state bank or trust company having combined capital and surplus of at least \$100,000,000 and designated by the Corporation as the transfer agent and/or registrar of the Series E Preferred Stock, or if no such designation is made, the Corporation.
- $\mbox{(z)}$ "Upper Exchange Rate" is defined in the definition of "Exchange Rate".

IN WITNESS WHEREOF, The TJX Companies,	
of Designation to be signed by its Vice Presi	dent - Finance and its Secretary
this, day of, 1995.	
	THE TJX COMPANIES, INC.
	P
	By:
	Otanaa B. Hishaaa
	Steven R. Wishner
	Vice President - Finance
Attact	
Attest:	
Jan. H. Malkana	
Jay H. Meltzer	
Secretary	

B-22

STANDSTILL AND REGISTRATION RIGHTS AGREEMENT

This Standstill Agreement (the "Agreement"), dated as of ______, 1995, is between Melville Corporation, a New York corporation ("Subscriber"), and The TJX Companies, Inc., a Delaware corporation ("Issuer").

WHEREAS, simultaneously with the execution of this Agreement, Subscriber is acquiring shares of Issuer's Series D Cumulative Convertible Preferred Stock, par value \$1.00 per share (the "Series D Preferred Stock") and shares of Issuer's Series E Cumulative Convertible Preferred Stock, par value \$1.00 per share (the "Series E Preferred Stock" and together with the Series D Preferred Stock, the "Preferred Stock"), pursuant to a Preferred Stock Subscription Agreement dated as of the date hereof (the "Subscription Agreement") between Subscriber and Issuer;

WHEREAS, Subscriber and Issuer entered into the Subscription Agreement pursuant to, and in connection with the transactions contemplated by, the Stock Purchase Agreement dated as of October 14, 1995 (the "Purchase Agreement") between Subscriber and Issuer; and

WHEREAS, Issuer and Subscriber desire to establish in this Agreement certain conditions of Subscriber's relationship with Issuer;

NOW, THEREFORE, in consideration of the mutual covenants and agreements contained herein and in the Subscription Agreement and the Purchase Agreement, the parties hereto agree as follows:

ARTICLE I

DEFINITIONS; REPRESENTATIONS AND WARRANTIES

SECTION 1.01 Definitions. Except as otherwise specified herein, defined terms used in this Agreement shall have the respective meanings assigned to such terms in the Purchase Agreement. Unless otherwise specified all references to "days" shall be deemed to be references to calendar days.

SECTION 1.02 Representations and Warranties of Issuer. Issuer represents and warrants to Subscriber as follows:

- (a) The execution, delivery and performance by Issuer of this Agreement and the consummation by Issuer of the transactions contemplated by this Agreement are within its corporate powers and have been duly authorized by all necessary corporate action on its part. This Agreement constitutes a legal, valid and binding agreement of Issuer enforceable against Issuer in accordance with its terms (i) except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to or affecting creditors' rights generally, including the effect of statutory and other laws regarding fraudulent conveyances and preferential transfers, and (ii) subject to the limitations imposed by general equitable principles (regardless of whether such enforceability is considered in a proceeding at law or in equity); and
- (b) The execution, delivery and performance of this Agreement by Issuer does not and will not contravene or conflict with or constitute a default under Issuer's Charter or By-laws.
- SECTION 1.03 Representations and Warranties of Subscriber. Subscriber represents and warrants to Issuer as follows:
 - (a) The execution, delivery and performance by Subscriber of this Agreement and the consummation by Subscriber of the transactions contemplated by this Agreement are within its corporate powers and have been duly authorized by all necessary corporate action on its part. This Agreement constitutes a legal, valid and binding agreement of Subscriber enforceable against Subscriber in accordance with its terms (i) except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect relating to or affecting creditors' rights generally, including the effect of statutory and other laws regarding fraudulent conveyances and preferential transfers, and (ii) subject to the limitations imposed by general equitable principles (regardless of whether such enforceability is considered in a proceeding at law or in equity);
 - (b) The execution, delivery and performance of this Agreement by Subscriber does not and will not contravene or conflict with or constitute a default under Subscriber's Charter or By-laws; and
 - (c) Subscriber "beneficially owns" (as such term is defined in Rule 13d-3 under the Exchange Act) the shares of the Preferred Stock issued to it pursuant to the Subscription Agreement and neither Subscriber nor any "affiliate" or "associate" (such terms being used in this Agreement as such terms are defined in Rule 12b-2 under the Exchange Act), owns any other Voting Securities (as defined in Section 2.01 herein).

ARTICLE II

TERM

SECTION 2.01 Term. The term (the "Term") of this Agreement shall commence on the date hereof and shall continue until the date on which the Voting Power of the Voting Securities, on a fully diluted basis, beneficially owned by Subscriber shall represent less than three percent (3%) of the Total Voting Power. For the purposes of this Agreement (i) the term "Voting Securities" shall mean any securities entitled to vote generally in the election of directors of Issuer, or any direct or indirect rights or options to acquire any such securities or any securities (including without limitation the Preferred Stock) convertible or exercisable into or exchangeable for such securities, whether or not such securities are so convertible, exercisable or exchangeable at the time of determination, (ii) the term "Voting Power" shall mean the voting power in the general election of directors of Issuer, and (iii) the term "Total Voting Power" shall mean the total combined Voting Power of all the Voting Securities then outstanding and entitled to vote; provided, however, that for

purposes of this Agreement, the Voting Power of the Preferred Stock on any date shall mean the voting power of the shares of the Issuer's common stock, par value \$1.00 per share ("Common Stock"), into which the shares of Preferred Stock would be convertible on such date, assuming for this purpose only that the Automatic Conversion Date (as defined in the Series D Preferred Stock and the Series E Preferred Stock, respectively) were such date, at the Exchange Rate (as so defined) then in effect.

ARTICLE III

STANDSTILL AND VOTING PROVISIONS

SECTION 3.01 Restrictions of Certain Actions by Subscriber. During the Term, Subscriber will not, and will cause each of its affiliates and associates not to, singly or as part of a partnership, limited partnership, syndicate or other group (as those terms are used in Section 13(d)(3) of the Exchange Act), directly or indirectly:

- (a) acquire, offer to acquire, or agree to acquire, by purchase, gift or otherwise, any Voting Securities, except pursuant to a stock split, stock dividend, rights offering, recapitalization, reclassification or similar transaction;
- (b) make or in any way participate in any "solicitation" of "proxies" to vote (as such terms are defined in Rule 14a-1 under the Exchange Act), solicit any consent or communicate with or seek to advise or influence any person or entity with respect to the voting of any Voting Securities or become a "participant" in any "election contest" (as such terms are defined or used in Rule 14a-11 under the Exchange Act) with respect to Issuer;

- (c) form, join or encourage the formation of, any "person" within the meaning of Section 13(d)(3) of the Exchange Act with respect to any Voting Securities; provided that this Section 3.01(c) shall not prohibit any such arrangement solely among Subscriber and any of its wholly-owned Subsidiaries;
- (d) deposit any Voting Securities into a voting trust or subject any such Voting Securities to any arrangement or agreement with respect to the voting thereof; provided that this Section 3.01(d) shall not prohibit any such arrangement solely among Subscriber and any of its wholly-owned Subsidiaries;
- (e) initiate, propose or otherwise solicit stockholders of the Issuer for the approval of any stockholder proposal with respect to Issuer as described in Rule 14a-8 under the Exchange Act, or induce or attempt to induce any other person to initiate any such stockholder proposal;
- (f) seek election to or seek to place a representative on the Board of Directors of Issuer or seek the removal of any member of the Board of Directors of Issuer;
- (g) call or seek to have called any meeting of the stockholders of Issuer;
- (h) otherwise act to seek to control, direct or influence the management, policies or affairs of Issuer;
- (i) except as otherwise provided in Section 4.02 or Article V, sell or otherwise transfer in any manner any Voting Securities to any "person" (within the meaning of Section 13(d)(3) of the Exchange Act) who to the knowledge of Subscriber beneficially owns or who as a result of such sale or transfer will beneficially own at least three percent (3%) of the Total Voting Power or who, without the approval of the Board of Directors of Issuer, has proposed a business combination or similar transaction with, or a change of control of, Issuer or who has proposed a tender offer for Voting Securities or who has discussed with Subscriber the possibility of proposing a business combination or similar transaction with, or a change in control of, Issuer or a tender offer for Voting Securities; provided, however, that insofar as this clause (i) has application to a sale or other transfer by Subscriber to an institutional investor pursuant to Section 4.01, the reference to the words "three percent (3%)" shall be deemed to be deleted from this clause (i) and replaced with the words "five percent (5%)";
- (j) solicit, seek to effect, negotiate with or provide any information to any other party with respect to, or make any statement or proposal, whether written or oral, to the Board of Directors of Issuer or any director or officer of Issuer or otherwise

make any public announcement or proposal whatsoever with respect to, any form of business combination transaction involving Issuer, including, without limitation, a merger, exchange offer or liquidation of Issuer's assets, or any acquisition, disposition, restructuring, recapitalization or similar transaction with respect to Issuer; or

(k) instigate or encourage any third party to do any of the foregoing. $% \begin{center} \end{center} \begin{center} \end{center}$

If Subscriber or any of its affiliates or associates owns or acquires any Voting Securities in violation of this Agreement, such Voting Securities shall immediately be disposed of to persons who are not affiliates or associates thereof but only in compliance with the provisions of this Section 3.01; provided, however, that Issuer may also pursue any other available remedy to which it may be entitled as a result of such violation.

SECTION 3.02 Voting. (a) During the Term, whenever Subscriber (or any of its affiliates or associates) shall have the right to vote their Voting Securities, Subscriber (and any such affiliates or associates) shall (i) be present, in person or represented by proxy, at all stockholder meetings of Issuer so that all Voting Securities beneficially owned by it and its affiliates and associates shall be counted for the purpose of determining the presence of a quorum at such meetings, and (ii) subject to Section 3.02(b) below, vote or cause to be voted, or consent with respect to, all Voting Securities beneficially owned by it and its affiliates and associates in the manner recommended by Issuer's Board of Directors, except that during any period or at any time when there shall be in full force and effect a valid order or judgment a court of competent jurisdiction or a ruling, pronouncement or requirement of the New York Stock Exchange, Inc. ("NYSE") to the effect that the foregoing provisions of this Section 3.02 are invalid, void, enforceable or not in accordance with NYSE policy, then Subscriber will, if so requested by the Board of Directors of Issuer, vote or cause to be voted all of its Voting Securities beneficially owned by it and its affiliates and associates in the same proportion as the votes cast by or on behalf of the other holders of Issuer's Voting Securities.

(b) Notwithstanding anything to the contrary contained in Section 3.02(a) above, Subscriber shall have the right to vote freely, without regard to any request or recommendation of the Board of Directors of Issuer, with respect to the matters specified in Section 7 of the Certificate of Designations establishing the terms of the Series D Preferred Stock and Section 7 of the Certificate of Designations establishing the terms of the Series E Preferred Stock.

ARTICLE IV

TRANSFER RESTRICTIONS

SECTION 4.01 Right of First Offer. (a) If Subscriber desires to transfer any Voting Securities, it shall in each case comply with the provisions of Section 3.01 and give written notice ("Subscriber's Notice") to Issuer (i) stating that it desires to make such transfer, and

- (ii) setting forth the number of shares of Voting Securities proposed to be transferred (the "Offered Shares"), the cash price per share that Subscriber proposes to be paid for such Offered Shares (the "Offer Price"), and the other material terms and conditions of such transfer. Subscriber's Notice shall constitute an irrevocable offer by Subscriber to sell to Issuer the Offered Shares at the Offer Price in cash.
- (b) Within 5 Business Days after receipt of Subscriber's Notice, Issuer may elect to purchase all (but not less than all) of the Offered Shares at the Offer Price in cash by delivery of a notice ("Issuer's Notice") to Subscriber stating Issuer's irrevocable acceptance of the Offer.
- (c) If Issuer fails to elect to purchase all of the Offered Shares within the time period specified in Section 4.01(b), then Subscriber may, subject to compliance with the provisions of Section 3.01, within a period of 120 days following the expiration of the time period specified in Section 4.01(b), transfer (or enter into an agreement to transfer) all or any Offered Shares for cash; provided, that if the purchase price per share to be paid by any purchaser of the Offered Shares is less than 90% of the Offer Price (the "Reduced Transfer Price"), Subscriber shall promptly provide written notice (the "Reduced Transfer Price Notice") to Issuer of such intended transfer (including the material terms and conditions thereof) and Issuer shall have the right, exercisable by delivery of a written election notice to Subscriber within five Business Days of receipt of such notice, to purchase such Offered Shares at the Reduced Transfer Price.
- (d) If Issuer fails to elect to purchase the Offered Shares at the Offer Price (or, if applicable, the Reduced Transfer Price) within the relevant time period specified in Section 4.01(b) (or, if applicable, Section 4.01(c)) and Subscriber shall not have transferred or entered into an agreement to transfer the Offered Shares prior to the expiration of the 120-day period specified in Section 4.01(c), the right of the first offer under this Section 4.01 shall again apply in connection with any subsequent transfer of such Offered Shares.
- (e) Any purchase of Voting Securities by Issuer pursuant to this Section 4.01 shall be on a mutually determined closing date which shall be not less than 30 days nor more than 45 days after the last notice is given with respect to such purchase. The closing shall be held at 10:00 A.M., local time, at the principal office of Issuer, or at such other time or place as the parties mutually agree.
- (f) On the closing date, Subscriber shall deliver (i) certificates representing the shares of Voting Securities being sold, free and clear of any Lien, and (ii) such other documents, including evidence of ownership and authority, as Issuer may reasonably request. The purchase price shall be paid by wire transfer of immediately available funds no later than 2:00 P.M. on the closing date.

- (g) This Section 4.01 shall not apply to Subscriber's sale of Voting Securities in an underwritten public offering registered under the Securities Act pursuant to Article V.
- SECTION 4.02 Rights Pursuant to a Tender Offer. Subscriber shall have the right to sell or exchange all its Voting Securities pursuant to a tender or exchange offer for at least a majority of the Voting Securities (an "Offer"). However, prior to such sale or exchange, Subscriber shall give Issuer the opportunity to purchase such Voting Securities in the following manner:
 - (i) Subscriber shall give notice (the "Tender Notice") to Issuer in writing of its intention to sell or exchange Voting Securities in response to an Offer no later than three calendar days prior to the latest time (including any extensions) by which Voting Securities must be tendered in order to be accepted pursuant to such Offer, specifying the amount of Voting Securities proposed to be tendered by Subscriber (the "Tendered Shares") and the purchase price per share specified in the Offer at the time of the Tender Notice.
 - (ii) Issuer shall have the right to purchase all, but not part, of the Tendered Shares exercisable by giving written notice (an "Exercise Notice") to Subscriber at least two calendar days prior to the latest time after delivery of the Tender Notice by which Voting Securities must be tendered in order to be accepted pursuant to the Offer (including any extensions thereof) and depositing in escrow (or similar arrangement) a sum in cash sufficient to purchase all Tendered Shares at the price then being offered in the Offer, without regard to any provision thereof with respect to proration or conditions to the offeror's obligation to purchase. The delivery by Issuer of an Exercise Notice and deposit of funds as provided above in response to a Tender Notice will, except as provided below, constitute an irrevocable agreement by Issuer to purchase, and Subscriber to sell, the Tendered Shares in accordance with the terms of this Section 4.02, whether or not the Offer or any other tender or exchange offer (a "Competing Tender Offer") for Voting Securities that was outstanding during the Offer is consummated.
 - (iii) The purchase price to be paid by Issuer for any Voting Securities purchased by it pursuant to this Section 4.02 shall be the highest price offered or paid in the Offer or in any Competing Tender Offer. For purposes hereof, the price offered or paid in a tender or exchange offer for Voting Securities shall be deemed to be the price offered or paid pursuant thereto, without regard to any provisions thereof with respect to proration or conditions to the offeror's obligation to purchase. If the purchase price per share specified in the Offer includes any property other than cash (the "Offer Noncash Property"), the purchase price per share at which Issuer shall be entitled to purchase all, but not part, of the Tendered Shares shall be (y) the amount of cash per share, if any, specified in such Offer (the "Cash Portion"), plus (z) an amount

of cash per share equal to the value of the Offer Noncash Property per share (the "Cash Value of Offer Noncash Property"), as determined in good faith by the mutual agreement of the parties hereto, or if the parties cannot agree, by a nationally recognized investment banking firm selected by mutual agreement of the parties. If Issuer exercises its right of first refusal by giving an Exercise Notice, the closing of the purchase of the Voting Securities with respect to such right (the "Closing") shall take place at 3:00 p.m., local time (or, if earlier, two hours before the latest time by which Voting Securities must be tendered in order to be accepted pursuant to the Offer), on the last day on which Voting Securities must be tendered in order to be accepted pursuant to the Offer (including any extensions thereof) (the "Latest Tender Date"), and Issuer shall pay the purchase price for the Voting Securities specified above. Subscriber shall be entitled to rescind its Tender Notice at any time prior to the Latest Tender Date by Notice in writing to Issuer; provided, however, that if on or before the Latest Tender Date, Issuer publicly announces that Issuer has approved, proposed or entered into an agreement with respect to (either individually or together with any other persons) a recapitalization, reorganization or business combination with respect to Issuer or all or substantially all of its assets, Subscriber shall be entitled to rescind its Tender Notice by notice in writing to Issuer at any time prior to the Closing on the Latest Tender Date. If Subscriber rescinds its Tender Notice pursuant to the immediately preceding sentence, Issuer's Exercise Notice with respect to such Offer shall be deemed to be immediately rescinded and Subscriber's disposition of its Voting Securities in response to the Offer with respect to which the Tender Notice is rescinded or any other Offer shall again be subject to all of the provisions of this Section 4.02.

(iv) If Issuer does not exercise its right of first refusal set forth in this Section 4.02 within the time specified for such exercise by giving an Exercise Notice, then Subscriber shall be free to accept, for all its Voting Securities, the Offer with respect to which the Tender Notice was given (including any increases and extensions thereof).

SECTION 4.03 Assignment of Rights. Issuer may assign any of its rights of first refusal under this Article IV to any Subsidiary or Affiliate of Issuer without the consent of Subscriber, provided, however, that no such assignment shall relieve Issuer of any of its obligations pursuant to this Article IV. In the event that Issuer elects to exercise a right of first refusal under this Article IV, Issuer may specify in its Exercise Notice (or thereafter prior to purchase) another such Person as its designee to purchase the Voting Securities to which such notice relates.

ARTICLE V

REGISTRATION RIGHTS

SECTION 5.01 Registration Upon Request. At any time commencing on the

date hereof and continuing thereafter, Subscriber shall have the right to make written demand upon Issuer, on not more than two separate occasions (subject to the provisions of this Section 5.01), to register under the Securities Act, shares of Series E Preferred Stock or shares of Common Stock received by Subscriber upon conversion or redemption of shares of Preferred Stock (such shares of Series E Preferred Stock and Common Stock being referred to as the "Subject Stock"), and Issuer shall use its best efforts to cause such shares to be registered under the Securities Act as soon as reasonably practicable so as to permit the sale thereof promptly; provided, however, that each such demand shall cover at least \$40 million liquidation preference of Series E Preferred Stock (or any balance thereof exceeding \$15 million) or 2 million shares of Common Stock (subject to adjustment for stock splits, reverse stock splits, stock dividends and similar events after the date hereof). In connection therewith, Issuer shall prepare, and within 120 days of the receipt of the request, file, on Form S-3 if permitted or otherwise on the appropriate form, a registration statement under the Securities Act to effect such registration. Subscriber agrees to provide all such information and materials and to take all such action as may be reasonably required in order to permit Issuer to comply with all applicable requirements of the Securities Act, the rules and regulations thereunder and the Securities and Exchange Commission (the "SEC") and to obtain any desired acceleration of the effective date of such registration statement. If the offering to be registered is to be underwritten, the managing underwriter shall be selected by Subscriber and shall be reasonably satisfactory to Issuer and Subscriber shall enter into an underwriting agreement containing customary terms and conditions. Notwithstanding the foregoing, Issuer (i) shall not be obligated to prepare or file more than one registration statement other than for purposes of a stock option or other employee benefit or similar plan during any twelve-month period and (ii) shall be entitled to postpone for a reasonable period of time, the filing of any registration statement otherwise required to be prepared and filed by Issuer if (A) Issuer is, at such time, conducting or about to conduct an underwritten public offering of securities and is advised by its managing underwriter or underwriters in writing (with a copy to Subscriber), that such offering would, in its or their opinion, be materially adversely affected by the registration so requested, or (B) Issuer determines in its reasonable judgment and in good faith that the registration and distribution of the shares of Subject Stock would interfere with any announced or imminent material financing, acquisition, disposition, corporate reorganization or other material transaction of a similar type involving Issuer. In the event of such postponement, Subscriber shall have the right to withdraw the request for registration by giving written notice to Issuer within 20 days after receipt of the notice of postponement (and, in the event of such withdrawal, such request shall not be counted for purposes of determining the number of registrations to which Subscriber is entitled pursuant to this Section 5.01). Issuer shall not grant to any other holder of its securities, whether currently outstanding or issued in the future (other than as provided in the Share Purchase Agreement dated as of

April 15, 1992 among Issuer and the other parties thereto and the Exchange Agreement dated as of August 20, 1992 among the same parties, as presently in effect, relating to Issuer's former Series A Cumulative Convertible Preferred Stock and its New Series A Cumulative Convertible Preferred Stock (collectively, the "Series A Agreements")), any incidental or piggyback registration rights with respect to any registration statement filed pursuant to a demand registration under this Section 5.01. Without the prior consent of Subscriber (other than as provided in the Series A Agreements), Issuer will not permit any other holder of its securities to participate in any offering made pursuant to a demand registration under this Section 5.01.

In the event that Issuer does not redeem all of the then outstanding shares of Series D Preferred Stock pursuant to Section 4(b) of the Certificate of Designation of the Series D Preferred Stock (unless Subscriber shall have elected to convert any such shares following receipt of notice of redemption pursuant to Section 4(a) of such Certificate), (i) Subscriber shall be entitled to an additional demand right under the first sentence of this Section 5.01, subject to the minimum offering amounts requirement referred to above and (ii) Issuer shall, from time to time, at Subscriber's reasonable request, provide an opportunity for senior officers of Subscriber to meet with senior officers of Issuer to discuss the business and affairs of Issuer.

SECTION 5.02 Incidental Registration Rights. If Issuer proposes to register any of its Common Stock under the Securities Act (other than (i) pursuant to Section 5.01 hereof, (ii) securities to be issued pursuant to a stock option or other employee benefit or similar plan, and (iii) securities proposed to be issued in exchange for other securities or assets (other than cash) or in connection with a merger or consolidation with another corporation), Issuer shall, as promptly as practicable, give written notice to Subscriber of Issuer's intention to effect such registration. If, within 15 days after receipt of such notice, Subscriber submits a written request to Issuer specifying not less than one million shares of Common Stock constituting Subject Stock that are then beneficially owned by Subscriber and that Subscriber proposes to sell or otherwise dispose of in accordance with this Section 5.02, Issuer shall use its best efforts to include the shares specified in Subscriber's request in such registration. Subscriber may exercise its rights under this Section 5.02 on no more than three separate occasions; provided that if the number of securities that Subscriber had initially requested be included in a registration under this Section 5.02 is reduced pursuant to clause (C) below and Subscriber withdraws from such registration, then Subscriber's request shall not be counted as one of such three requests. If the offering pursuant to such registration statement is to be made by or through underwriters, the Subscriber and such underwriter shall execute an underwriting agreement in customary form. If the managing underwriter reasonably determines in good faith and advises Subscriber that the inclusion in the registration statement of all the Common Stock proposed to be included by all holders of Common Stock entitled to participate (other than on a demand basis) would interfere with the successful marketing of the securities proposed to be registered, then Issuer will include in such registration that number of such shares of Common Stock which does not exceed the number which such managing underwriter

reasonably determines in good faith can be sold without interfering with the successful marketing of the securities proposed to be registered based upon the following order of priority: (A) first, the securities Issuer proposes to sell, (B) second, the securities any participant exercising demand registration rights proposes to sell and (C) third, the securities of each other Person who is entitled to participate (other than on a demand basis) in such registration (including Subscriber) on a pro rata basis based on the number of shares of Common Stock owned by each such Person; provided that if the number of securities that Subscriber had initially requested be included in a registration under this Section 5.02 is reduced pursuant to clause (C), Subscriber may withdraw all securities from such registration. No registration effected under this Section 5.02 shall relieve Issuer of its obligation to effect any registration upon request under Section 5.01. If Subscriber has been permitted to participate in a proposed offering pursuant to this Section 5.02, Issuer thereafter may determine either not to file a registration statement relating thereto, or to withdraw such registration statement, or otherwise not to consummate such offering, without any liability hereunder. Any underwriters participating in a distribution of Subject Stock pursuant to Sections 5.01 and 5.02 hereof shall use all reasonable efforts to effect as wide a distribution as is reasonably practicable, and in no event shall any sale (other than a sale to underwriters making such a distribution) of shares of Subject Stock be made knowingly to any person (including its affiliates or associates and any group in which that person or its affiliates or associates shall be a member if Subscriber or underwriters know of the existence of such a group or affiliate or associate) that, after giving effect to such sale, would beneficially own at least three percent (3%) of the Total Voting Power. Subscriber shall use all reasonable efforts to secure the agreement of the underwriters, in connection with any underwritten offering of its Subject Stock, to comply with the

SECTION 5.03 Registration Mechanics. In connection with any offering of shares of Subject Stock registered pursuant to Section 5.01 and 5.02 herein, Issuer shall (i) furnish to Subscriber such number of copies of any prospectus (including preliminary and summary prospectuses) and conformed copies of the registration statement (including amendments or supplements thereto and, in each case, all exhibits) and such other documents as it may reasonably request, but only while Issuer shall be required under the provisions hereof to cause the registration statement to remain current; (ii)(A) use its best efforts to register or qualify the Subject Stock covered by such registration statement under such blue sky or other state securities laws for offer and sale as Subscriber shall reasonably request and (B) keep such registration or qualification in effect for so long as the registration statement remains in effect; provided, however, that Issuer shall not be obligated to qualify to do business as a foreign corporation under the laws of any jurisdiction in which it shall not then be qualified or to file any general consent to service of process in any jurisdiction in which such a consent has not been previously filed or subject itself to taxation in any jurisdiction wherein it would not otherwise be subject to tax but for the requirements of this Section 5.03; (iii) use its best efforts to cause all shares of Subject Stock covered by such registration statement to be registered with or approved by such other federal or state government agencies or authorities as may be necessary in the opinion of counsel to Issuer to enable Subscriber to consummate

the disposition of such shares of Subject Stock; (iv) at any time when a prospectus relating thereto is required to be delivered under the Securities Act notify Subscriber upon discovery that, or upon the happening of any event as a result of which, the prospectus included in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein not misleading, in the light of the circumstances under which they were made, and (subject to the good faith determination of Issuer's Board of Directors as to whether to permit sales under such registration statement), at the request of Subscriber promptly prepare and furnish to it a reasonable number of copies of a supplement to or an amendment of such prospectus as may be necessary so that, as thereafter delivered to the purchasers of such securities, such prospectus shall not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading, in the light of the circumstances under which they were made; (v) otherwise use its best efforts to comply with all applicable rules and regulations of the SEC; (vi) use its best efforts to list, if required by the rules of the applicable securities exchange or, if securities of the same class are then so listed, the Subject Stock covered by such registration statement on the New York Stock Exchange or on any other securities exchange on which Subject Stock is then listed; and (vii) before filing any registration statement or any amendment or supplement thereto, and as far in advance as is reasonably practicable, furnish to Subscriber and its counsel copies of such In connection with the closing of any offering of Subject Stock registered pursuant to Section 5.01 or 5.02, Issuer shall (x) furnish to the underwriter, if any, unlegended certificates representing ownership of the Subject Stock being sold in such denominations as requested and (y) instruct any transfer agent and registrar of the Subject Stock to release any stop transfer orders with respect to such Subject Stock. Upon any registration becoming effective pursuant to Section 5.01 or 5.02, Issuer shall use its best efforts to keep such registration statement effective for a period of 60 days (or 90 days, if Issuer is eligible to use a Form S-3, or successor form) or such shorter period as shall be necessary to effect the distribution of the Subject Stock.

Subscriber agrees that upon receipt of any notice from Issuer of the happening of any event of the kind described in subdivision (iv) of this Section 5.03, it will forthwith discontinue its disposition of Subject Stock pursuant to the registration statement relating to such Subject Stock until its receipt of the copies of the supplemented or amended prospectus contemplated by subdivision (iv) of this Section 5.03 and, if so directed by Issuer, will deliver to Issuer all copies then in its possession of the prospectus relating to such Subject Stock current at the time of receipt of such notice. If Subscriber's disposition of Subject Stock is discontinued pursuant to the foregoing sentence, unless Issuer thereafter extends the effectiveness of the registration statement to permit dispositions of Subject Stock by Subscriber for an aggregate of 60 days (or 90 days, if Issuer is eligible to use a Form S-3, or successor form), whether or not consecutive, the registration statement shall not be counted for purposes of determining the number of registrations to which Subscriber is entitled pursuant to Section 5.01.

SECTION 5.04 Expenses. Subscriber shall pay all agent fees and commissions and underwriting discounts and commissions related to shares of Subject Stock being sold by Subscriber and the fees and disbursements of its counsel and accountants and Issuer shall pay all fees and disbursements of its counsel and accountants in connection with any registration pursuant to this Article V. All other fees and expenses in connection with any registration statement (including, without limitation, all registration and filing fees, all printing costs, all fees and expenses of complying with securities or blue sky laws) shall (i) in the case of a registration pursuant to Section 5.01, be borne by Issuer and (ii) in the case of a registration pursuant to Section 5.02, be shared pro rata based upon the respective market values of the securities to be sold by Issuer, Subscriber and any other holders participating in such offering provided, that Subscriber shall not pay any expenses relating to work that would otherwise be incurred by Issuer including, but not limited to, the preparation and filing of periodic reports with the SEC.

SECTION 5.05 Indemnification and Contribution. In the case of any offering registered pursuant to this Article V, Issuer agrees to indemnify and hold Subscriber, each underwriter, if any, of the Subject Stock under such registration and each person who controls any of the foregoing within the meaning of Section 15 of the Securities Act, and any officer, employee or partner of the foregoing, harmless against any and all losses, claims, damages or liabilities (including reasonable legal fees and other reasonable expenses incurred in the investigation and defense thereof) to which they or any of them may become subject under the Securities Act or otherwise (collectively "Losses"), insofar as any such Losses shall arise out of or shall be based upon (i) any untrue statement or alleged untrue statement of a material fact contained in the registration statement relating to the sale of such Subject Stock (as amended if Issuer shall have filed with the SEC any amendment thereof), or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading or (ii) any untrue statement or alleged untrue statement of a material fact contained in the prospectus relating to the sale of such Subject Stock (as amended or supplemented if Issuer shall have filed with the SEC any amendment thereof or supplement thereto), or the omission or alleged omission to state therein a material fact necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, provided, however, that the indemnification contained in this Section 5.05 shall not apply to such Losses which shall arise out of or shall be based upon any such untrue statement or alleged untrue statement, or any such omission or alleged omission, which shall have been made in reliance upon and in conformity with information furnished in writing to Issuer by Subscriber or any such underwriter, as the case may be, specifically for use in connection with the preparation of the registration statement or prospectus contained in the registration statement or any such amendment thereof or supplement therein.

In the case of each offering registered pursuant to this Article V, Subscriber agrees and each underwriter, if any, participating therein shall severally agree, substantially in the same manner and to the same extent as set forth in the preceding paragraph, to indemnify and hold harmless Issuer and each person, if any, who controls Issuer within the meaning of Section 15

of the Securities Act, and the directors and officers of Issuer, with respect to any statement in or omission from such registration statement or prospectus contained in such registration statement (as amended or as supplemented, if amended or supplemented as aforesaid) if such statement or omission shall have been made in reliance upon and in conformity with information furnished in writing to Issuer by Subscriber or such underwriter, as the case may be, specifically for use in connection with the preparation of such registration statement or prospectus contained in such registration statement or any such amendment thereof or supplement thereto.

Notwithstanding the provisions of this Section 5.05, no underwriter shall be required to contribute any amount in excess of the amount by which the total price at which the securities underwritten by it and distributed to the public were offered to the public exceeds the amount of any damages which such underwriter has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission, and the Subscriber registering shares pursuant to Section 5.01 or Section 5.02 shall not be required to contribute any amount in excess of the amount by which the total price at which the securities of the Subscriber were offered to the public (less underwriters' discounts and commissions) exceeds the amount of any damages which the Subscriber has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission.

Each party indemnified under this Section 5.05 shall, promptly after receipt of notice of the commencement of any claim against such indemnified party in respect of which indemnity may be sought hereunder, notify the indemnifying party in writing of the commencement thereof. The failure of any indemnified party to so notify an indemnifying party shall not relieve the indemnifying party from any liability in respect of such action which it may have to such indemnified party on account of the indemnity contained in this Section 5.05, unless (and only to the extent) the indemnifying party was prejudiced by such failure, and in no event shall such failure relieve the indemnifying party from any other liability which it may have to such indemnified party. In case any action in respect of which indemnification may be sought hereunder shall be brought against any indemnified party and it shall notify an indemnifying party of the commencement thereof, the indemnifying party shall be entitled to participate therein and, to the extent that it may desire, jointly with any other indemnifying party similarly notified, to assume the defense thereof through counsel reasonably satisfactory to the indemnified party, and after notice from the indemnifying party to such indemnified party of its election so to assume the defense thereof, the indemnifying party shall not be liable to such indemnified party under this Section 5.05 for any legal or other expenses subsequently incurred by such indemnified party in connection with the defense thereof, other than reasonable costs of investigation (unless such indemnified party reasonably objects to such assumption on the grounds that there may be defenses available to it which are different from or in addition to those available to such indemnifying party in which event such indemnified party, and any other indemnified party to which any different or additional defenses apply, shall be reimbursed by the indemnifying party for the reasonable expenses incurred in connection with retaining one separate legal counsel for all such indemnified

-14-

parties). No indemnifying party shall, without the prior written consent of the indemnified party (which consent shall not be unreasonably withheld), effect any settlement of any claim or pending or threatened proceeding in respect of which the indemnified party is or could have been a party and indemnity could have been sought hereunder by such indemnified party, unless such settlement includes an unconditional release of such indemnified party from all liability arising out of such claim or proceeding.

If the indemnification provided for in this Section 5.05 is unavailable to an indemnified party or is insufficient to hold such indemnified party harmless from any Losses in respect of which this Section 5.05 would otherwise apply by its terms (other than by reason of exceptions provided herein), then each applicable indemnifying party, in lieu of indemnifying such an indemnified party, shall have a joint and several obligation to contribute to the amount paid or payable by such indemnified party as a result of such Losses, in such proportion as is appropriate to reflect the relative benefits received by and fault of the indemnifying party, on the one hand, and such indemnified party, on the other hand, in connection with the offering to which such contribution relates as well as any other relevant equitable considerations. The relative benefit shall be determined by reference to, among other things, the amount of proceeds received by each party from the offering to which such contribution The relative fault shall be determined by reference to, among other things, each party's relative knowledge and access to information concerning the matter with respect to which the claim was asserted, and the opportunity to correct and prevent any statement or omission. The amount paid or payable by a party as a result of any Losses shall be deemed to include any legal or other fees or expenses incurred by such party is connection with any investigation or proceeding, to the extent such party would have been indemnified for such expenses if the indemnification provided for in this Section 5.05 was available to such party.

The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 5.05 were determined by pro rata allocation or by any other method of allocation that does not take account of the equitable considerations referred to in the immediately preceding paragraph. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the 1993 Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

SECTION 5.06 Limitations on Registration Rights. Notwithstanding anything to the contrary in this Article V, all rights of Subscriber under this Article V shall be subject to the provisions of the Series A Agreements. To the extent that any of the provisions of this Article V conflict with any provisions of the Series A Agreements, the provisions of the Series A Agreements shall control and there shall be no breach of or default under Article V of this Agreement to the extent the performance of any term of Article V of this Agreement would cause a breach of or default under either of the Series A Agreements.

SECTION 5.07 Holdback Agreements. If and to the extent requested by the managing underwriter or underwriters, in the case of any underwritten public offering, Subscriber agrees

not to effect, except as part of such registration, any sale of shares of Common Stock or Preferred Stock during the 14 days prior to, and during the 90-day period beginning on, the effective date of such registration statement.

ARTICLE VI

MISCELLANEOUS

SECTION 6.01 Enforcement. Subscriber, on the one hand, and Issuer, on the other, acknowledge and agree that irreparable damage would occur if any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, the parties will be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically its provisions in any court having jurisdiction, this being in addition to any other remedy to which they may be entitled at law or in equity.

SECTION 6.02 Entire Agreement; Waivers. This Agreement, the other Closing Agreements, the Confidentiality Agreement and the Purchase Agreement constitute the entire agreement among the parties hereto pertaining to the subject matter hereof and thereof and supersede all prior and contemporaneous agreements, understandings, negotiations and discussions, whether oral or written, of the parties with respect to such subject matter. No waiver of any provision of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), shall constitute a continuing waiver unless otherwise expressly provided nor shall be effective unless in writing and executed (i) in the case of a waiver by Issuer, by Issuer and (ii) in the case of a waiver by Subscriber, by Subscriber.

SECTION 6.03 Amendment or Modification. The parties hereto may not amend or modify this Agreement except in such manner as may be agreed upon by a written instrument executed by Issuer and Subscriber.

SECTION 6.04 Survival. All representations, warranties, covenants and agreements made by or on behalf of any party hereto in this Agreement shall survive the execution and delivery of this Agreement and the Closing.

SECTION 6.05 Successors and Assigns. All the terms and provisions of this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective transferees, successors and assigns (each of which such transferees, successors and assigns shall be deemed to be a party hereto for all purposes hereof); provided, however, that (i) neither Issuer nor Subscriber may assign or transfer any of its rights or obligations hereunder without the prior written consent of the other (except as set forth in Section 4.03) and (iii) no transfer or assignment by any party shall relieve such party of any of its obligations hereunder.

SECTION 6.06 Severability. If any provision of this Agreement is held by a court of competent jurisdiction to be unenforceable, the remaining provisions shall remain in full force and effect. It is declared to be the intention of the parties that they would have executed the remaining provisions without including any that may be declared unenforceable.

SECTION 6.07 Headings. Descriptive headings are for convenience only and will not control or affect the meaning or construction of any provision of this Agreement.

SECTION 6.08 Counterparts. For the convenience of the parties, any number of counterparts of this Agreement may be executed by the parties, and each such executed counterpart will be an original instrument.

SECTION 6.09 Notices. Any notices or other communications required or permitted hereunder shall be sufficiently given if in writing (including telecopy or similar teletransmission), addressed as follows:

If to Seller, to it at: Melville Corporation

One Theall Road Rye, New York 10580 Telecopier: 914-925-4052

Telecopier: 914-925-4052 Attention: Chief Executive Officer,

Chief Financial

Officer and General Counsel

With a copy to: Davis Polk & Wardwell

450 Lexington Avenue New York, New York 10017 Telecopier: 212-450-5744 Attention: Dennis S. Hersch

770 Cochituate Road Framingham, MA 01701 Telecopier: 508-390-2457

Attn: President and General Counsel

With a copy to: Ropes & Gray

One International Place Boston, MA 02110 Telecopier: 617-951-7050

Attention: Arthur G. Siler, Esq.

Unless otherwise specified herein, such notices or other communications shall be deemed received (a) in the case of any notice or communication sent other than by mail, on the date actually delivered to such address (evidenced, in the case of delivery by overnight courier, by confirmation of delivery from the overnight courier service making such delivery, and in the case of a telecopy, by receipt of a transmission confirmation form or the addresse's confirmation of receipt), or (b) in the case of any notice or communication sent by mail, three Business Days after being sent, if sent by registered or certified mail, with first-class postage prepaid. Each of the parties hereto shall be entitled to specify a different address by giving notice as aforesaid to each of the other parties hereto.

SECTION 6.10 Governing Law. This Agreement shall be governed by and construed in accordance with the domestic substantive law of the State of Delaware, without giving effect to any choice or conflict of law provision or rule that would cause the application of the law of any other jurisdiction.

SECTION 6.11 Termination. This Agreement may be terminated by Issuer and Subscriber by mutual written consent at any time prior to the Closing, and this Agreement shall automatically terminate immediately upon the termination of the Purchase Agreement in accordance with its terms.

IN WITNESS WHEREOF, the parties hereto have caused this Standstill and Registration Rights Agreement to be executed as of the date first referred to above.

THE TJX COMPANIES, INC.

By:

Name:
Title:

MELVILLE CORPORATION

By:
Name:
Title:
-18-

The First National Bank of Chicago One First National Plaza Chicago, Illinois 60670

The Bank of New York One Wall Street New York, New York 10286 Bank of America Illinois 231 South LaSalle Street Chicago, Illinois 60697

Pearl Street L.P. c/o Goldman, Sachs & Co. 85 Broad Street New York, New York 10004

October 14, 1995

The TJX Companies, Inc. 770 Cochituate Road Framingham, Massachusetts 01701 Attn: Don Campbell Chief Financial Officer

Re: Commitment Letter

Gentlemen:

The TJX Companies, Inc., a Delaware corporation ("TJX" or the "Borrower") has requested of each of The First National Bank of Chicago ("First Chicago"), Bank of America Illinois ("BOA"), The Bank of New York ("BNY") and Pearl Street L.P., an affiliate of Goldman, Sachs & Co. ("Pearl") (First Chicago, BOA, BNY and Pearl each, together with its successors and assigns, being individually referred to as a "Co-Arranger" and collectively being referred to as the "Co-Arrangers") that senior unsecured credit facilities (the "Facilities") be made available to Borrower in the aggregate principal amount of \$875,000,000 (the "Aggregate Facilities") on the terms and conditions set forth in this letter and in the term sheet of even date attached hereto (the "Term Sheet"; together with this letter, the "Commitment Letter"). Capitalized terms used in this letter and not defined in this letter shall have the meanings given to them in the Term Sheet. \$500,000,000 of the Aggregate Facilities will be available on a revolving credit basis to the Borrower for its and its direct and indirect guarantor subsidiaries working capital needs, for letters of credit (in a maximum amount of \$100,000,000) and for other general corporate purposes. \$375,000,000 of the Aggregate Facilities will be available to the Borrower as a single-draw term loan to finance the "Acquisition Transactions" (defined below), including the acquisition by the Borrower of all of the outstanding capital stock of the target corporation identified by the Borrower to the Co-Arrangers ("Blue") from its sole stockholder ("XYZ Corporation") for an initial cash purchase price of \$375,000,000 and to finance transaction costs in connection therewith.

Borrower has advised the Co-Arrangers that:

(a) Borrower intends to enter into a Stock Purchase Agreement (the "Purchase Agreement") between Borrower and XYZ Corporation; and

THE BANK OF NEW YORK PEARL Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 2

(b) Borrower intends to enter into a Preferred Stock Subscription Agreement with XYZ Corporation (the "Subscription Agreement") pursuant to which Borrower will issue to XYZ Corporation shares of the Borrower's Series D Preferred Stock, par value \$1 per share in the aggregate principal amount of \$25,000,000 and Series E Preferred Stock, par value \$1 per share in the aggregate principal amount of \$150,000,000 (collectively, the "Preferred Stock") (collectively, the transactions under the Purchase Agreement and the Subscription Agreement being referred to herein as the "Acquisition Transactions").

The Co-Arrangers are pleased to confirm their interest in acting as the Co-Arrangers for the Facilities and to confirm their several commitments and to each provide its several commitment to participate as a co-lender in the Facilities in a maximum amount of \$218,750,000 allocated pro rata to the Facilities ("Commitment Amount"), for an aggregate amount of \$875,000,000 (the "Co-Arrangers' Aggregate Commitment") on the terms and conditions set forth in this Commitment Letter and subject to the conditions contained in the letter regarding fees of even date herewith (the "Fee Letter").

The Co-Arrangers intend and reserve the right to syndicate the Facilities on a pro rata or non-pro rata basis to a group of banks and financial institutions (collectively, including the Co-Arrangers, the "Lenders") selected by the Co-Arrangers with a corresponding pro rata reduction in the commitments of the Co-Arrangers. The Co-Arrangers will jointly manage all aspects of the syndication, including decisions as to the selection of institutions to be approached, the timing of all offers to potential Lenders and the acceptance of commitments, the amounts offered and finally allocated, and the distribution of compensation from the fees discussed herein. To assist the Co-Arrangers in their syndication efforts, by its acceptance hereof, Borrower agrees (i) to provide and cause its advisors to provide the Co-Arrangers upon request with all information deemed reasonably necessary by the Co-Arrangers to complete successfully the syndication of the Facilities, including, without limitation, all information and projections prepared by the Borrower or on the Borrower's behalf relating to Blue and the Acquisition Transactions; (ii) to actively participate in, and to cause the management of Borrower and the Borrower's subsidiaries, their respective advisors, and to the extent possible prior to its becoming a subsidiary of the Borrower, the management of XYZ Corporation and Blue, to actively participate in, both (a) the preparation of an information package regarding the operations and prospects of the Borrower, its present subsidiaries and Blue, (b) the presentation thereof in bank meetings, (c) other communications requested by the Co-Arrangers with prospective Lenders in connection with the syndication of the Facilities; (iii) not to make any statement publicly about the Facilities which might negatively affect the Coability to syndicate the Facilities and (iv) not to take to market any additional bank facilities or debt securities until completion of the syndication of the Facilities. Such assistance will also include

Bank of America Illinois Pearl Street L.P.

ne Bank of New York Pearl Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 3

the Borrower using its best efforts to cause the other parties participating in the contemplated transactions to assist in the syndication to other prospective Lenders.

Each of the Co-Arrangers' commitments hereunder is subject to their customary legal due diligence investigation of the assets and liabilities of Blue and its subsidiaries and the Borrower and its subsidiaries with results reasonably satisfactory to each of the Co-Arrangers. In addition, each of the Co-Arrangers' commitments hereunder is subject to confirmation that no information with respect to the Borrower and its subsidiaries, Blue and its subsidiaries or the combination thereof furnished to the Co-Arrangers in connection with the Acquisition Transactions and the Facilities contained any material misstatement of fact or omitted to state a material fact or any fact necessary to make the statements contained therein (in the context of all other information provided, taken as a whole) not misleading. Each of the Co-Arrangers has received and reviewed the draft Purchase Agreement dated October 12, 1995 (Document Id #3075631.RED) ("Draft Agreement"). Each of the Co-Arrangers' commitments hereunder is conditioned on (i) receipt and review of the definitive Purchase Agreement (including all schedules and exhibits thereto), which Purchase Agreement shall not differ in any respect reasonably deemed material by the Co-Arrangers from the Draft Agreement and which Purchase Agreement shall have completed information and schedules on terms reasonably acceptable to the Co-Arrangers; (ii) the accuracy, as of the closing date of all of the representations and warranties in the Purchase Agreement (consistent with the provisions set forth in Section 7.1.1 thereof); and (iii) satisfaction by the parties to the Purchase Agreement of the conditions precedent contained therein without amendment or waiver, other than as consented to by the Co-Arrangers (such consent not to be unreasonably withheld). The Co-Arrangers have each performed a review of certain historical and pro forma financial information of Borrower and certain of its affiliates, and of Blue and certain of its affiliates based upon materials prepared by Borrower. The terms contained in this Commitment Letter are based solely upon such review and assume the accuracy of such information. Accordingly, each of the Co-Arrangers may terminate its commitment under this Commitment Letter upon any materially adverse modifications to such information. In furtherance (but not in limitation) of the on-going confirmation of their due diligence investigation, the Co-Arrangers shall continue to be given access to the management, records, books of account, contracts and properties of the Borrower and its subsidiaries and Blue and its subsidiaries to the extent reasonably requested, and the Co-Arrangers shall have received such financial, business and other information regarding the Borrower and its subsidiaries and Blue and its subsidiaries as they shall have reasonably requested, including, without limitation, historical and pro forma financial statements, management forecasts, information as to possible contingent liabilities, tax information, liabilities for employee and retiree health and benefits claims, environmental information, obligations under ERISA, collective bargaining agreements and other arrangements with employees.

Bank of America Illinois Pearl Street L.P.

ne Bank of New York Pearl Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 4

The commitments of each of the Co-Arrangers described herein are subject to (i) the preparation, execution, and delivery of a mutually acceptable credit agreement ("Credit Agreement") and other loan documents (collectively, the "Loan Documents") incorporating, without limitation, substantially the terms and the conditions outlined herein and in the Term Sheet; (ii) any of the Co-Arranger's determination that (a) there has been no material adverse change in the business, condition (financial or otherwise), operations, performance, properties, projections or prospects of the Borrower and its material subsidiaries taken as a whole from the audited financial information supplied to the Co-Arrangers or of Blue and its subsidiaries taken as a whole from the financial and other information supplied to the Co-Arrangers; and (b) there has been no material disruption or material adverse change prior to closing in primary and secondary loan syndication markets or capital markets or public equity markets generally; and (iii) the absence of termination of any of the Co-Arrangers' commitments. In the event there shall occur a material adverse change as described in clause (ii) above or the termination of any of the Co-Arrangers' Commitments described in clause (iii), at each of the Co-Arranger's option, its commitment hereunder shall thereupon immediately terminate without liability to the Borrower or any other person or party. The terms of this Commitment Letter are intended to outline only, and not to summarize all of, terms, conditions, covenants, representations and warranties, events of default, and other provisions that will be contained in the Loan Documents, which will include, in addition to the provisions outlined herein, provisions customary for this type of lending transaction. Without limiting the foregoing, all documents and matters pertinent to the financing transactions contemplated by this Commitment Letter, such as the Credit Agreement, the other Loan Documents and opinions of Borrower's counsel to the Co-Arrangers and the Lenders and all other documentation relating to Acquisition Transactions, including, without limitation the Purchase Agreement and the Preferred Stock shall be reasonably satisfactory to each of the Co-Arrangers and the Co-Arrangers' counsel, Sidley &

Borrower agrees to (i) reimburse each of the Co-Arrangers for all reasonable out-of-pocket expenses (including, without limitation, the fees and expenses of one outside counsel firm (and any necessary local counsel), and travel and related expenses) incurred in connection with the preparation, negotiation, and execution, administration, and syndication of any document (including, without limitation, this Commitment Letter and the Fee Letter) relating to this transaction and its role hereunder and the Co-Arrangers' ongoing due diligence in connection therewith; (ii) reimburse each of the Co-Arrangers (and the other Lenders) for all reasonable out-of-pocket costs, fees and expenses incurred in the enforcement of any document (including, without limitation, this Commitment Letter and the Fee Letter) relating to this transaction, including, the reasonable fees and expenses of outside counsel and time charges for inside counsel; (iii) indemnify and hold harmless the Lenders (including each of the Co-Arrangers) and their affiliates and each of their respective officers, employees, partners, agents and directors (collectively, including the Lenders, the "Indemnified Persons") against any and

Bank of America Illinois Pearl Street L.P.

ne Bank of New York Pearl Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 5

all losses, claims, damages, or liabilities of every kind whatsoever to which any Indemnified Person may become subject in connection in any way with the transaction which is the subject of this Commitment Letter, including without limitation, reasonable expenses incurred in connection with investigating or defending against any liability or action whether or not a party thereto, except to the extent any of the foregoing is found in a final judgment by a court of competent jurisdiction to have arisen solely from such Indemnified Person's gross negligence or wilful misconduct; and (iv) to assert no claim against any Co-Arrangers, the Lenders or any other Indemnified Persons, on any theory of liability, for special, indirect, consequential, exemplary or punitive damages on any theory of liability in connection in any way with the transaction which is the subject of this Commitment Letter. The obligations described in this paragraph are independent of all other obligations of Borrower hereunder and under the Loan Documents, shall survive the expiration, revocation or termination of this Commitment Letter by, and shall be payable whether or not the financing transactions contemplated by this letter shall close. Each of the Co-Arranger's obligations under this commitment are enforceable solely by the party signing this letter and may not be relied upon by any other person, including without limitation, XYZ Corporation or Blue. For purposes of enforcing this indemnity, Borrower irrevocably submits to the non-exclusive jurisdiction of any court in which a claim arising out of or relating to the services provided under this Commitment Letter is properly brought against the Co-. Arrangers or the Lenders and irrevocably waives any objection as to venue or inconvenient forum. IF THIS COMMITMENT LETTER, THE TERM SHEET, THE FEE LETTER, OR ANY ACT, OMISSION OR EVENT DESCRIBED IN THIS PARAGRAPH BECOMES THE SUBJECT OF A DISPUTE, THE BORROWER AND THE CO-ARRANGERS EACH HEREBY WAIVE TRIAL BY JURY. Borrower agrees not to settle any claim, litigation or proceeding relating to this transaction (whether or not any of the Co-Arrangers is a party thereto) unless such settlement releases all Indemnified Persons from any and all liability in respect of such transaction.

Borrower acknowledges and agrees that each of the Co-Arrangers may share any and all information obtained by it with respect to the Borrower and its subsidiaries or Blue and its subsidiaries with each other Co-Arranger and with their respective affiliates and may employ the services of their respective affiliates in providing certain services hereunder, and the Borrower acknowledges and agrees that such affiliates shall be entitled to the benefits afforded the Co-Arrangers under this Commitment Letter.

By accepting delivery of the Commitment Letter and the Fee Letter, Borrower agrees that, prior to executing this Commitment Letter, it will not disclose either expressly or impliedly, without the prior written consent of each of the Co-Arrangers, which consent may not be unreasonably withheld, to any person any of the terms of this Commitment Letter or the Fee Letter, or the fact that this Commitment Letter or the Fee

Bank of America Illinois Pearl Street L.P.

The Bank of New York Pearl Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 6

Letter or the financing proposal represented thereby exists except that each may disclose any of the foregoing to any employee, financial advisor (but not to any commercial lender or other financial institution) or attorney of Borrower and, other than with respect to the Fee Letter, XYZ Corporation to whom, in each case, it is necessary to disclose such information so long as any such employee, advisor or attorney is made aware of, is directed to observe this confidentiality obligation and agrees to the terms of this confidentiality agreement before the disclosure. Upon the Borrower's execution of this Commitment Letter and the commitment letters of the other Co-Arrangers for the full amount of the Aggregate Facilities, the Borrower may make public disclosure of the existence and the amount of the Facilities; and the Borrower may file a copy of this Commitment Letter (but not the Fee Letter) or make such other disclosures if such disclosure is, in the opinion of the Borrower's counsel, required by law. If the Borrower does not accept this commitment, the Borrower shall immediately return this Commitment Letter, the Fee Letter and all copies and drafts of any the foregoing to the Co-Arrangers.

Each of the Co-Arrangers and the Borrower acknowledges that the Co-Arrangers may, from time to time, effect transactions for their own accounts or the accounts of customers, and hold positions in loans or options on loans of the Borrower, Blue, XYZ Corporation and other companies that may be the subject of this arrangement. In addition, certain affiliates of one or more of the Co-Arrangers are or may be securities firms and as such may effect, from time to time, transactions for their own accounts or for the accounts of customers and hold positions in securities or options on securities of the Borrower, Blue, XYZ Corporation and other companies that may be the subject of this arrangement.

This Commitment Letter (including the Term Sheet attached hereto) and the Fee Letter supersedes and replaces all prior letters and communications regarding this matter.

Please indicate your acceptance of this Commitment Letter in the space indicated below and return a copy of this Commitment Letter so executed together with the executed Fee Letter to each of the Co-Arrangers. This Commitment Letter will expire at 5 p.m. (Chicago time), Tuesday, October 17, 1995, unless on or prior to such time the Co-Arrangers shall have received (i) a copy of this Commitment Letter executed by Borrower, (ii) the Fee Letter executed by Borrower and (iii) the fees required to be paid to the Co-Arrangers under the Fee Letter at the time required by the Fee Letter. Notwithstanding timely acceptance of the commitment pursuant to provisions of this paragraph, the commitment will automatically terminate unless definitive Loan Documents are executed on or before January 31, 1996. By its acceptance hereof, Borrower agrees to pay the Co-Arrangers the fees described in the Fee Letter of even date herewith. This Commitment Letter and the Fee Letter shall be governed by and construed in accordance with the internal laws of the State of Illinois.

Bank of America Illinois Pearl Street L.P.

he Bank of New York Pearl Street L.P.

The TJX Companies, Inc. October 14, 1995 Page 7

None of the Co-Arranger's commitments shall be assignable by the Borrower without the prior written consent of the Co-Arrangers and the granting of such consent in any given instance shall not constitute a waiver of this requirement as to any subsequent assignment. Each of the Co-Arranger's may assign any or all of its rights and obligations under this Commitment Letter or the Fee Letter to one or more of its affiliates or may perform some or all of its functions in connection with the arrangement and syndication of the Facilities through one or more of its affiliates. In the event of such an assignment or use of an affiliate, all references to the Co-Arrangers shall be deemed to be and include such affiliate. The terms of this Commitment Letter may not be waived or modified unless such waiver or modification is expressly stated as such and specifically agreed to by the parties in writing, and shall be enforceable by the Co-Arrangers and their respective successors and assigns.

This Commitment Letter may be executed by one or more of the parties hereto on any number of separate counterparts and all of said counterparts taken together shall be deemed to constitute one and the same instrument.

Γhe	First	Nation	al Bar	ık of	Chicago
Γhe	Rank (of New	Vork		

Bank of America Illinois Pearl Street L.P.

he Bank of New York Pearl Street L.P.

The TJX Companies, Inc.

The TJX Companies, Inc. October 14, 1995 Page 8

We appreciate the opportunity to structure a financing commitment that meets your requirements and we look forward to working with you.

THE FIRST NATIONAL BANK OF CHICAGO BANK OF AMERICA ILLINOIS

By:	Ву:	
Title:	Title:	
THE BANK OF NEW YORK	PEARL STREET L.P.	
By:	Ву:	
Title:	Title:	
Accepted and agreed: THE TJX COMPANIES, INC.		
By:		
Title:		
Data		

Bank of America Illinois The Bank of New York

Pearl Street L.P.

TERM SHEET THE TJX COMPANIES, INC.

October 14, 1995

THE FACILITIES

The TJX Companies, Inc. (the "Borrower"). Obligor:

Co-Arrangers and Co-

Syndication Agents: The First National Bank of Chicago, Bank of America Illinois, The Bank of New York and Pearl Street L.P. (an affiliate of Goldman, Sachs & Co.) (or, in each

case, a designated affiliate) (the "Co-Arrangers").

Administrative Agent:

The First National Bank of Chicago.

Syndication Agent:

Bank of America Illinois (or a designated affiliate)

Documentation Agent:

The Bank of New York

Lenders:

A syndicate of Lenders, including the Co-Arrangers selected by the Co-Arrangers in consultation with the

Borrower.

Guarantees:

Blue will and each of the Borrower's material subsidiaries will, jointly and severally, provide guaranties of all of the Borrower's indebtedness under the Facilities on terms and conditions acceptable to the Co-Arrangers. The Borrower and the guarantor subsidiaries will be required to enter into an intercorporate contribution and indemnity agreement on terms and conditions reasonably satisfactory to the Co-

Facility A: Revolving Credit Facility

Amount:

\$500 million.

Swingline Subfacility:

Up to \$25 million of the revolving credit facility will be available for loans ("Swingline Loans") to be made

by one or more of the Co-Arrangers for the account of all of the Lenders to the Borrower. Each of the Lenders shall automatically acquire a pro rata unconditional risk participation in each Swingline Loan.

Page 1

The	First	National	Bank of	Chicago	The	TJX Companies,	Inc.
- 1							

Bank of America Illinois The Bank of New York

Pearl Street L.P.

Letter of Credit Subfacility:

A letter of credit facility, as a subfacility of the revolving credit facility, for the issuance of commercial and standby letters of credit will be established in an aggregate amount not to exceed \$100,000,000. Letters of credit will be issued by the Lenders designated in the Credit Agreement as issuing Lenders for the account of all of the Lenders. Upon issuance of any letter of credit, each of the Lenders shall automatically acquire a pro rata unconditional risk participation in such letter of credit.

Purpose:

For working capital and other general corporate purposes.

Clean Down:

The outstanding balance under the revolving credit facility shall be reduced to not greater than \$100 million for 30 consecutive days during each five month period from November 1 through March 31. The Co-Arrangers will consider netting cash and cash equivalents accumulated and maintained by the Borrower and its subsidiaries for purposes of meeting this requirement on terms satisfactory to the Co-Arrangers.

Maturity:

Third anniversary of closing with bullet maturity.

Facility B: Term Loan Facility

Amount:

\$375 million.

Purpose:

To fund the acquisition of all of the outstanding capital stock of Blue from XYZ Corporation, pursuant to the Purchase Agreement, the terms and conditions of which shall be acceptable to the Co-Arrangers as outlined in the cover letter attached hereto.

Amortization:

Levels to be determined on a basis satisfactory to the

Borrower and each of the Co-Arrangers.

Maturity:

Fifth anniversary of closing.

INTEREST RATES

Borrowing Options:

For loans other than the Swingline Loans:

. Adjusted LIBOR plus the applicable margins as set

forth below

. Alternate Base Rate

Page 2

Γhe	First	National	Bank	of	Chicago

The TJX Companies, Inc.

Bank of America Illinois The Bank of New York Pearl Street L.P.

For the Swingline Loans:

. Money Market Rate

"Alternate Base Rate" means the greater of (i) the corporate base rate of interest announced from time to time by the Administrative Agent changing when and as said rate changes or (ii) the federal funds rate plus $\,$ 1/2% per annum.

"LIBOR" means the rate at which deposits in U.S. Dollars are offered by the reference banks (to be selected) in the London interbank market two business days prior to the borrowing date, adjusted for reserves. LIBOR loans will be available for interest periods of one, two, three, or six months.

"Money Market Rate" shall be a transaction loan rate to be negotiated between the Swingline Lenders and the Borrower.

Swingline Loans will be available at the Money Market Rate and will have maturities on a basis to be mutually agreed upon between the Borrower and the Co-Arrangers.

Interest on LIBOR loans will be calculated on a 360-day basis. Interest on all Alternate Base Rate loans and Money Market Rate loans will be calculated on a 365/366 day basis.

Increased Costs:

The Credit Agreement will contain customary provisions regarding availability, increased costs (including capital cost increases imposed by regulatory authorities), illegality, and early payment.

Post-Default Interest: After the occurrence and during the continuance of a default, upon notice given to the Borrower the applicable interest rate for all loans will be equal to the Alternate Base Rate plus 2% per annum.

FEES

In addition to the fees agreed to be paid by the Borrower to the Co-Arrangers pursuant to fee agreements, the Borrower will pay the following fees to the Lenders:

Facility Fee:

A per annum facility fee ("Facility Fee") calculated on a 360-day basis payable on each Lender's Facility A commitment,

The First National Bank of Chicago	The TJX Companies, Inc.
Bank of America Illinois	
The Bank of New York	
Pearl Street L.P.	

irrespective of usage, payable quarterly in arrears, on the date of any reduction of the commitments with respect to the amount of such reduction and on termination of Facility A. (See pricing grid below.)

Letter of Credit Fees: Fronting fees for the account of the issuing ----- Lenders and letter of credit fees for the pro rata account of the Lenders in amounts to be agreed upon among the Borrower and the Co-Arrangers.

APPLICABLE MARGINS AND APPLICABLE FEES

Pricing:

The applicable margins on the loans and the applicable fees on the commitments will be at the rates, expressed in basis points per annum, set out in the pricing grid below and varying according to the pricing level commensurate with credit quality. Level III pricing shall apply until 60 days following the closing date; provided,

if during such 60-day period the Borrower has its senior unsecured bond ratings affirmed, the applicable Level under the pricing grid below shall be utilized from the date such ratings are affirmed. If the Borrower's senior unsecured bond ratings have not been affirmed within such 60 days, then Level IV pricing shall apply, commencing with the 60th day following the closing and continuing with the 60th day following the closing and continuing until the date on which the Borrower has its senior unsecured bond ratings affirmed. In the event that the Borrower receives an A- or A3 rating, but there is more than one ratings level difference with the other ratings agency, the Level II pricing shall be applicable.

Page 4

The First National Bank of	Chicago	The TJX Compan	ies, Inc.		
Bank of America Illinois					
The Bank of New York					
Pearl Street L.P.					
		Pricing Gri	d		
	Level I	Level II	Level III	Level IV	Level V
Credit Quality	If the Borrower's senior unsecured long-term public debt ratings are equal to or better than A- from S&P or A3 from Moody's.	If the Borrower's senior unsecured long-term public debt ratings are equal to or better than BBB+ from S&P and Baa1 from Moody's.	If the Borrower's senior unsecured long-term public debt ratings are equal to or better than BBB from S&P and Baa2 from Moody's.	If the Borrower's senior unsecured long-term public debt ratings are equal to or better than BBB- from S&P and Baa3 from Moody's.	If the Borrower's senior unsecured long-term public debt ratings are lower than BBB- from S&P or Baa3 from Moody's, or unrated.
Facility Fee on the Revolving Loan Facility	15 b.p.	20 b.p.	25 b.p.	30 b.p.	37.5 b.p.
Applicable ABR Margin for Revolving Loans and Term Loans	0 b.p.	0 b.p.	0 b.p.	0 b.p.	0 b.p.
Applicable LIBOR Margin for Revolving Loans	35 b.p.	42.5 b.p.	50 b.p.	70 b.p.	87.5 b.p.
Applicable LIBOR Margin for Term Loans	50 b.p.	62.5 b.p.	75 b.p.	100 b.p.	125 b.p.
"Unused" Cost per annum	15 b.p.	20 b.p.	25 b.p.	30 b.p.	37.5 b.p.
"Used" Cost (Facility Fee plus spread over LIBOR)	50 b.p.	62.5 b.p.	75 b.p.	100 b.p.	125 b.p.
	GENERAL FACILITY TERI	MS			

Drawdowns:

Minimum amounts of \$25 million with additional increments of \$5 million on LIBOR and ABR Loans. Minimum drawdowns on Swingline Loans of \$1 million. Drawdowns are at the Borrower's option with same-day notice for Swingline and Alternate Base Rate Loans, and three business days for Adjusted LIBOR Loans.

Adjusted Libon Louis.

Page 5

The First National Bank of Chicago	The TJX Companies, Inc.
Bank of America Illinois	
The Bank of New York	
Pearl Street L.P.	

Optional Prepayments:

Alternate Base Rate Loans may be prepaid at any time on one business day's notice. Adjusted LIBOR loans may be prepaid before the end of an Interest Period, subject to five business days prior notice and funding loss indemnity. All optional prepayments under Facility B shall be applied to the remaining scheduled amortization payments on a pro-rata basis.

Mandatory Prepayments:

The Borrower shall prepay loans under Facility B from the proceeds of certain asset sales and from the proceeds of newly issued securities according to terms to be negotiated, provided however, that the Borrower may use proceeds of certain asset sales on terms to be negotiated in connection with the mandatory redemption of the Series D Preferred Stock and may use the proceeds of equity securities to replace the Preferred Stock issued to XYZ Corporation as part of the acquisition of Blue on terms and subject to conditions to be negotiated with the Lenders (but if such replacement equity securities consist of common stock of the Borrower or convertible securities which must be converted into common stock of the Borrower with terms substantially similar to the terms of the Series E Preferred Stock no such approval of the Lenders shall be required). All mandatory prepayments under Facility B shall be applied to the remaining scheduled amortization payments on a pro-rata basis.

Termination or Reduction of Commitments:

The Borrower may terminate the unused commitments under Facility A in amounts of at least \$25 million at any time on three business days' notice. Accrued and unpaid Facility Fees with respect to the commitments so reduced shall be payable on the commitment reduction date.

Facility Size:

In the event that the Borrower, prior to closing, determines that less than the total of \$875,000,000 will be necessary in connection with the consummation of the transaction, the allocation of any reduction of the facilities between the revolving credit facility and the term loan facility will be on a basis satisfactory to the Co-Arrangers and the Borrower.

The First National Bank of Chicago	The TJX Companies, Inc.
Bank of America Illinois	
The Bank of New York	
Pearl Street L.P.	

COVENANTS

Covenants:

The Credit Agreement will have customary covenants for similar investment grade credits including, but not limited to, limitations on mergers, liens and encumbrances, limitations on subsidiary indebtedness and subsidiary guarantees, compliance with laws, sale of assets (with special conditions to be established in regard to the sale of certain assets), ERISA, acquisitions, investments, loans and advances, change of control, environmental, and the following:

Reporting

Requirements:

Annual certified audited consolidated and unaudited consolidating financial statements of the Borrower and its consolidated subsidiaries due within 105 days after each fiscal year. Quarterly certified unaudited consolidated financial statements of the Borrower and its consolidated subsidiaries due within 60 days after each of the first three fiscal quarters. The Borrower will also provide a quarterly no-default and covenant compliance certificate signed by the Chief Financial Officer. Reporting requirements will include other usual and customary information required to be supplied in similar credits or as reasonably required by the Co-Arrangers.

Financial Covenants:

The Credit Agreement will contain consolidated financial covenants containing limitations to be negotiated including, without limitation, covenants pertaining to:

- . Debt to Capital Ratio . Fixed Charge Coverage Ratio . Minimum Tangible Net Worth
- Conditions of Borrowing:

The obligations of the Lenders to make the loans under the Facilities are subject to customary conditions precedent (accuracy of representations and warranties, no default, etc.).

Conditions of Initial Borrowing:

Prior to the initial funding the Co-Arrangers shall be satisfied in regard to the following conditions:

. The purchase by the Borrower of the capital stock of Blue ("the Purchase") shall have been consummated in accordance with the Purchase Agreement not later than

Page 7

Γhe	First	National	Bank	of	Chicago

The TJX Companies, Inc.

Pearl Street L.P.

January 31, 1996.

- . Evidence that the Borrower's and Blue's respective directors shall have approved the Purchase; all material regulatory and legal approvals for the Purchase shall have been obtained
- . Absence of injunction, temporary restraining order or other similar proceedings which present, in the determination of the Co-Arrangers, a material challenge to the consummation of the Purchase or which could reasonably be expected to involve claims against the Co-Arrangers and absence of other litigation deemed by the Co-Arrangers to be material.
- . Completion of and satisfaction with customary legal due diligence investigation by counsel to the Co-Arrangers with respect to the Borrower and its subsidiaries, and Blue and its subsidiaries. Legal due diligence will involve, among others, review of contingencies, outstanding debt instruments, ERISA, litigation, contractual obligations, Schedules and exhibits to the Purchase Agreement, etc.
- . In payment of a portion of the purchase price for the Acquisition Transactions, the Borrower shall have issued at least \$175 million of Preferred Stock. The terms of the Preferred Stock shall be on substantially the terms set forth in the 10/12/95 draft of the "Series D Preferred Indicative Term Sheet" and the 10/12/95 draft of the "Series E Preferred Indicative Term Sheet" in each case as provided by the Borrower to the Co-Arrangers and otherwise reasonably acceptable to the Co-Arrangers.
- . Confirmation that Blue and its subsidiaries do not have any outstanding bank loan facilities.
- Receipt of other customary closing documentation, including, without limitation, subsidiary guarantees, legal opinions of the Borrower and its subsidiaries' counsel.
- . Absence of material adverse change in the business, condition (financial or otherwise), operations, performance, properties or prospects of the Borrower and its material subsidiaries taken as a whole, or Blue and its subsidiaries taken as a whole.

Γhe	First	National	Bank	of	Chicago

The TJX Companies, Inc.

Bank of America Illinois

The Bank of New York
----Pearl Street L.P.

- -------

. Cancellation by the Borrower of and repayment of all outstanding loans under all outstanding domestic committed credit facilities with each of the Co-Arrangers and other Lenders party to the Credit Agreement. Such cancellation and repayments shall occur prior to or simultaneously with the closing in the case of the Co-Arrangers or any other Lenders party to the Credit Agreement as of the closing date, and prior to or simultaneously with completion of syndication with respect to any Lenders to whom the Facilities are to be syndicated after closing.

- . No information with respect to the Borrower and its subsidiaries, Blue and its subsidiaries or the combination thereof furnished to the Co-Arrangers in connection with the Acquisition Transactions and the Facilities contained any material misstatement of fact or omitted to state a material fact or any fact necessary to make the statements contained therein (in the context of all other information provided, taken as a whole) not misleading.
- . All outstanding third-party indebtedness for money borrowed of the Borrower's subsidiary, Chadwick's of Boston, shall have been repaid in its entirety and the note agreements and guaranties issued in connection therewith cancelled.

Representations and Warranties:

The Borrower will make usual representations and warranties including, with respect to its financial statements, litigation, ERISA, compliance with all material requirements of law and contracts and compliance with Regulations G, T, U and X.

Defaults:

Usual and customary defaults, including without limitation, defaults for nonpayment of principal when due, nonpayment of interest and fees within five days, material misrepresentations, default in the performance of any financial covenant or negative covenant, default in performance of any other term or covenant for 30 days, bankruptcy or insolvency, ERISA, unstayed judgment in excess of \$10 million, and cross-default to any indebtedness equal to or in excess of \$10 million in the aggregate for the Borrower and its subsidiaries, which default would permit the holders of such indebtedness to cause such indebtedness to become due prior to its stated maturity.

Dama 0

The First National Ban	k of Chicago	The TJX Companies, Inc.
Bank of America Illino	is	
The Bank of New York		
Pearl Street L.P.		
Assignments/ Participations:	Following the completion by the Comprimary and general syndication, assignments with the consent of the Administrative Agent (such consent unreasonably withheld) or particity or commitments under the Facilities anything herein to the contrary, the assignments to any financial instance of the contraver of the co	the Lenders may sell ne Borrower and the sonot to be pations in their notes es. Notwithstanding the Borrower consents to itution with which the
Majority Lenders:	"Majority Lenders" means Lenders he the sum of the commitments and toommitments have been terminated, loans.	the term loans or, if the
Fees and Expenses:	The Borrower agrees to reimburse of for all reasonable costs, fees, the preparation, syndication, and Facilities, including, the reasonation of one outside counsel firm (and accounsel), whether or not the Acquithe Facilities are consummated, and fees and expenses incurred by any Co-Arrangers) in the enforcement of documents, including, the reasonation outside counsel and time charges in	and expenses incurred in execution of the able fees and expenses any necessary local isition Transaction or not the reasonable costs, Lender (including the of the final loan onle fees and expenses of

The internal laws of the State of Illinois.

Governing Law:

Page 10

NEWS RELEASE

CONTACTS: Steven Wishner Vice President Treasurer

Sherry Lang Assistant Vice President Investor Relations

FOR IMMEDIATE RELEASE (Monday, October 16, 1995)

THE TJX COMPANIES, INC. ANNOUNCES AGREEMENT TO ACQUIRE

MARSHALLS

Framingham, MA -- The TJX Companies, Inc. (NYSE:TJX), parent company of the leading off-price retailer T.J. Maxx, today announced that it has entered into a definitive agreement with Melville Corporation to acquire Marshalls, the off-price family apparel division of Melville, for \$375 million in cash and \$175 million in convertible junior preferred stock. The transaction is anticipated to be consummated during TJX's fourth quarter and is subject to certain conditions, including receipt of regulatory approvals. TJX has obtained acquisition financing commitments from a group of banks to fund the cash portion of the purchase price as well as TJX's anticipated future working capital needs.

Bernard Cammarata, President and Chief Executive Officer of The TJX Companies, Inc. commented, "We are excited to acquire Marshalls, as this acquisition creates an important business combination which enables TJX to capitalize on the significant synergies that exist between T.J. Maxx and Marshalls. To maximize profits and our customer base, we intend to continue operating stores under both the T.J. Maxx and Marshalls banners. Both businesses will realize substantial increases in buying power, economies of scale and efficiencies of operation which will dramatically reduce our expense ratios. We will invest these cost savings into lower prices for our customers, enhancing our already great values. In addition, we strongly believe that the economies of scale achieved through this acquisition will have a direct beneficial impact on TJX's EPS growth, which will translate into increased shareholder value.

"We will capitalize on the best that both T.J. Maxx and Marshalls have to offer. If a particular method of operating in one division is more efficient than in the other, we will have the opportunity to move to the more efficient method. We have the management expertise needed to tackle the task of bringing these two businesses together -- without distracting management's focus at TJX or at T.J. Maxx. In addition, we believe we are very capable of bringing about a meaningful turnaround in Marshalls' business."

-MORE-

THE TJX COMPANIES, INC. ANNOUNCES AGREEMENT TO ACQUIRE

MARSHALLS
- ----October 16, 1995
Page 2

In connection with the acquisition, TJX intends to reduce its annual common dividend from \$.56 to \$.28 per share effective with the next dividend to be declared. The previously declared common dividend, payable November 30, 1995, is unaffected. The Company will also eliminate its share repurchase program. "While the Company continues to believe that excess cash should be returned to shareholders in the form of common dividends, we are absolutely committed to maintaining a strong balance sheet and financial flexibility," noted Cammarata. "We expect the combination of T.J. Maxx and Marshalls to produce future earnings growth for our shareholders and firmly believe that applying \$20 million in annual dividend savings toward financing this acquisition will produce superior returns for our investors."

Cammarata continued, "The retail landscape is changing. In part, through consolidation, acquisition and divestiture, we have seen the emergence of stronger mass merchandisers and department stores. We are convinced that this acquisition positions TJX to compete more effectively against department stores, off-price apparel retailers and other retailing formats."

The TJX Companies, Inc. is the largest off-price specialty apparel retailer, with 571 T.J. Maxx stores, the nation's leading women's fashion off-price catalog Chadwick's of Boston, 47 Winners Apparel Ltd. off-price family apparel stores in Canada, and 23 HomeGoods off-price home fashions stores. TJX is also developing T.K. Maxx, an off-price apparel concept in the United Kingdom. As of September 30, 1995, Marshalls operates 495 stores.